The Isolation of gp41 Specific Monoclonal Antibodies from the Cervical IgA Repertoire of Highly Exposed Persistently Seronegative (HEPS) Commercial Sex Workers from Nairobi, Kenya using Mammalian Cell Display

By

Ryan G. Gaudet

A thesis submitted to the Faculty of Graduate Studies of

The University of Manitoba

In partial fulfillment of the requirements of the degree of

# **MASTER OF SCIENCE**

Department of Medical Microbiology

University of Manitoba

Winnipeg

Copyright © 2010 by Ryan Gaudet

# DEDICATION

To Courtney, whose love and support were invaluable in both my relocation to Winnipeg and the completion of this degree.

# ACKNOWLEDGEMENTS

I would like to thank my supervisor Dr. Jody Berry for his consistent help, support and guidance during the undertaking of this Master's Degree. Members of my advisory committee: Dr. Keith Fowke, Dr. Abdellilah Soussi Gounni, and Dr. Cindi Corbett for their assistance in the completion of this work. I would especially like to thank Dr. Corbett for her support in allowing me to continue to work in her lab for the duration of my degree. Also the laboratory members of the Monoclonal Antibody Lab Unit at the National Microbiology Laboratory, whose advice and expertise were greatly appreciated. Fellow graduate students Rob Vendramelli, Darren Boese, and Kevin Hay who conducted much of the preliminary work on this project and also developed some of the protocols described in this work. I would also like to thank both the laboratory members and study participants at the Pumwani clinic in Nairobi, Kenya for their participation in this study.

# **TABLE OF CONTENTS**

1 INTRODUCTION	1
1.1 GLOBAL IMPACT OF HIV-1	1
1.2 HIV-1 VIROLOGY	2
1.3 DIVERSITY OF HIV-1	3
1.4 HIV-1 LIFECYCLE AND IMMUNOPATHOGENESIS	4
1.4.1 EARLY DISCOVERIES	4
1.4.2 TYPICAL COURSE OF HIV INFECTION	5
1.4.3 HIV INFECTION OF OF THE FEMALE GENITAL TRACT	6
<b>1.5 STRUCTURE AND FUNCTION OF THE HIV-1 ENVELOPE</b>	7
1.5.1 NATIVE HIV-1 ENVELOPE SPIKE	7
1.5.2 STRUCTURE OF GP120	9
1.5.3 STRUCTURE OF GP41	9
<b>1.6 PROGRESS TOWARD AN HIV-1 ENVELOPE BASED VACCINE</b>	13
1.6.1 Cell Mediated or Humoral Immunity	13
1.6.2 ENVELOPE VACCINES	14
1.7 HOST RESISTANCE TO HIV-1	17
<b>1.8 ROLE OF IGA IN MUCOSAL IMMUNITY OF THE GENITAL TRACT</b>	19
1.8.1 IGA IN THE GENITAL TRACT	19
<b>1.8.2 POSSIBLE MECHANISMS OF IGA MEDIATED PROTECTION FROM HIV-1</b>	20
1.8.3 IGA IN HEPS SUBJECTS	22
<b>1.9 MONOCLONAL ANTIBODIES</b>	25
1.9.1 ANTIBODY STRUCTURE	25
1.9.2 ANTIGEN SELECTION AND AFFINITY MATURATION	27
1.9.3 MAB PRODUCTION METHODS	28
1.9.4 MAB PRODUCTION VIA DISPLAY ON MAMMALIAN CELLS	30
1.10 NEUTRALIZING ANTIBODIES AND HIV-1	31
1.10.1 ROLE OF THERAPEUTIC ANTIBODIES	31
1.10.2 MABS VERSUS GP120	33
1.10.3 MABS VERSUS GP41	34
1.11 SUMMARY/STATEMENT OF UBJECTIVES	36
2 MATERIALS AND METHODS	38
2.1 ANTIBODY VARIABLE (V) REGION GENE RECOVERY FROM PBMC AND CMC SAMPLES	38
2.1.1 STUDY POPULATION	38
2.1.2 LYMPHOCYTE COLLECTION	38
2.1.3 RNA ISOLATION	39
2.1.4 cDNA Synthesis	39
2.1.5 PCR STRATEGY AND PRIMER DESIGN	39
2.1.6 PCR Amplification of IGA and IGG VH REGIONS	40
2.1.7 PCR AMPLIFICATION OF KAPPA AND LAMBDA VL REGIONS	41
2.1.8 ANALYSIS OF PCR	41
2.1.9 PCR FRAGMENT ISOLATION	42
2.1.10 EXTENSION PCR OF VH- $\gamma$ , $\alpha$ and VL- $\kappa$ , $\lambda$	42
2.2 EXPRESSION VECTOR MANIPULATION	44
2.2.1 P-dbIgG Mutagenesis	44

2.2.2 P-dbLambda Generation	45
2.3 RESTRICTION ENDONUCLEASE DIGEST	46
2.3.1 RESTRICTION ENDONUCLEASE DIGEST OF VH- $\gamma_{1}\alpha$ and P-dbIgG1	46
2.3.2 RESTRICTION ENDONUCLEASE DIGEST OF VL- $\kappa$ , $\lambda$ and P-dbKappa/Lambda	47
2.4 CLONING B12 FAB FROM PCOMB3X INTO P-DBIGG1 AND P-DBKAPPA	49
2.5 ANTIBODY SECRETION SYSTEM: GENERATION OF ML 256 PBMC IGG LIBRARY	49
2.5.1 LIGATION OF VH-γ INTO P-DBIGG1	49
2.5.2 LIGATION OF VL-κ,λ INTO P-DBKAPPA/LAMBDA	49
2.5.3 Plasmid Library Amplification	50
2.5.4 DETERMINING THE DIVERSITY OF THE ML256 PBMC VH LIBRARY	50
2.5.5 GROWTH AND MAINTENANCE OF HEK293F CELLS	51
2.5.6 TRANSFECTION OF 293F CELLS WITH THE ML256 PBMC VH AND VL LIBRARY	51
2.5.7 RE-SUSPENSION OF THE TRANSFECTANTS INTO SEMI-SOLID MEDIA	52
$2.5.8\ \text{Screening Clones for Expression and reactivity against gp41 and gp120 using ELISA}$	52
2.5.9 V GENE RECOVERY FROM POSITIVE MAMMALIAN CELLS	53
2.5.10 Screening via Chain Shuffling	54
<b>2.6 CELLULAR DISPLAY SYSTEM: GENERATION OF A HIV-1 RESISTANT CERVICAL IGA LIBRARY</b>	55
2.6.1 CREATION OF P-DBIGG1-TM	55
2.6.2 CREATION OF P-DBIGG1-B12-TM FOR POSITIVE CONTROL	58
2.6.3 LIGATION OF RESISTANT CERVICAL (CMC-R) VH- $\alpha$ Genes into P-dbIgG1-TM	58
2.6.4 Ligation of CMC-R VL-κ, $\lambda$ Genes into P-dbKappa/Lambda	58
2.6.5 DETERMINING THE DIVERSITY OF THE CMC-R LIBRARY	59
2.7 Assessement of IgG1/k, $\lambda$ Surface Expression on 293F cells	59
2.7.1 TRANSFECTION	59
2.7.2 DIRECT IMMUNOFLUORESCENT CELL STAINING	59
2.7.3 FLOW CYTOMETRY ANALYSIS	59
2.8 FLUORESENCE-ACTIVATED CELL SORTING (FACS)	60
2.8.1 TRANSFECTION	60
2.8.2 INDIRECT MULTI-COLOUR IMMUNOFLUORESCENT CELL STAINING	61
2.0.5 FAC5 ISOLATION OF GP41/IGU DUUBLE POSITIVE CELLS 2.0 V CENE DECOVERY SELECTION AND AMDILIEICATION OF CD41 DOCITIVE CLONES	61 62
2.9 V GENE RECOVERY, SELECTION AND AMPLIFICATION OF GP41 POSITIVE CLONES	62
2.7.1 KINA KEUVERI FROM JU CELL JUD-LIDRARI 2.0.2. CDNA Synthesis Lising SensiSodidt	62
2.9.2 CDNA STRIFTESIS USING SENSISCHIFT 2.9.3 VH and VI. Amelification from 50 Cell Sub-Lirdady	63
2.9.5 VITAND VELAMI LIFICATION FROM 50 CEEL SOB-LIBRART 2.9.4 CLONING THE GP41 SPECIFIC VH AND VL SEQUENCES INTO EXPRESSION VECTORS	63
2.9.5 Pelosmid Ine of This Eente VITARD VESEQUENCES INTO EXPRESSION VECTORS	64
2.9.6 CHAIN SHUFFLING TRANSFECTION WITH V GENES FROM THE GP41 POSITIVE SUB-LIBRARY	65
2.9.7 SCREENING THE CHAIN SHUFFLING TRANSFECTION VIA ELISA	65
2.9.8 SELECTION AND AMPLIFICATION OF THE GP41 SPECIFIC ANTIBODIES	66
2.10 LARGE SCALE ANTIBODY EXPRESSION	66
2.10.1 LARGE SCALE TRANSFECTION OF GP41 MABS	66
2.10.2 CONCENTRATION AND PURIFICATION OF GP41 SPECIFIC MABS	66
2.11 CHARACTERIZATION OF THE GP41 MABS	67
2.11.1 MEASUREMENT OF ANTIBODY SPECIFICITY FOR GP41	67
2.11.2 MEASUREMENT OF ANTIBODY REACTIVITY WITH DENATURED/REDUCED GP41	67
2.11.3 ASSESSEMENT OF ANTIBODY BINDING TO HIV-1 ENVELOPE SPIKE (GP160) USING	
FLOW CYTOMETRY	68
2.11.4 ASSESSEMENT OF ANTIBODY BINDING TO HIV-1 ENVELOPE SPIKE (GP160) USING	
Fluoremetric Microvolume Assay Technology (FMAT)	69

2.11.5 SEQUENCE ANA	LYSIS OF	GP41	MABS
---------------------	----------	------	------

<u>3 RESULTS</u>	72
3.1 GENERATION OF AN ANTIBODY SECRETORY LIBRARY FROM HIV-1 POSITIVE PBMCs	72
3.1.1 V GENE AMPLIFICATION AND LIBRARY GENERATION	72
3.1.2 THE SCREENING AND SELECTION OF CLONES FROM THE ML 256 PBMC LIBRARY	76
3.1.3 V GENE RECOVERY AND EXPRESSION	76
3.2 GENERATION OF A MAMMALIAN CELL DISPLAY SYSTEM: HIV-1 RESISTANT CERVICAL	
IGA LIBRARY	79
3.2.1 Amplification of V gnees and Library Generation	79
3.2.2 ANTIBODY EXPRESSION ON THE CELL SURFACE IN TRANSIENT TRANSFECTION	81
3.2.3 FACS ANALYSIS OF TRANSIENTLY TRANSFECTED CMC LIBRARY	84
3.2.4 VH AND VL RECOVERY AND SUB-CLONING	85
3.3 CHARACTERIZATION OF GP41 SPECIFIC MABS	90
3.3.1 EVALUATION OF MAB SPECIFICITY FOR GP41	90
3.3.2 DETERMINATION OF THE NATURE OF THE MAB EPITOPES	90
3.3.3 BINDING OF MABS TO HIV-1 ENVELOPE SPIKE	93
3.3.4 MOLECULAR CHARACTERIZATION OF THE GP41 SPECIFIC MABS	96
4 DISCUSSION	100
4.1 INTRODUCTION	100
4.2 ADVANTAGES OF THE CELL DISPLAY SYSTEM	101
4.3 NATURE OF THE GP41 EPITOPES RECOGNIZED BY THE FOUR MABS	105
4.4 MOLECULAR ANALYSIS OF THE GP41 SPECIFIC MABS	108
4.5 PATH FORWARD	111
4.6 CONCLUSION	113
5 REFERENCES	114
APPENDIX I - PRIMER TABLE	133
APPENDIX II - PLASMID TABLE	134

# LIST OF FIGURES AND TABLES

<u>3 RESULTS</u>	<u>67</u>
FIGURE 1. THE HIV-1 VIRION AND THE ENVELOPE GLYCOPROTEIN TRIMER COMPLEX	8
FIGURE 2. STRUCTURE OF THE HIV-1 ENVELOPE PROTEIN GP41	12
FIGURE 3. STRATEGY FOR THE ISOLATION OF GP41 SPECIFIC MABS USING A MAMMALIAN CELL	
DISPLAY SYSTEM	56
FIGURE 4. CLONING THE IGG/K,A PBMC ANTIBODY REPERTOIRE FROM ML 256	
(HIV-1 POSITIVE)	73
FIGURE 5. SECRETORY ANTIBODY EXPRESSION SYSTEM	74
FIGURE 6. DIVERSITY OF THE CONSTRUCTED ML 256 PBMC LIBRARY	75
FIGURE 7. INITIAL SCREENING OF 1500 293F CLONES TRANSFECTED WITH THE ML256	
<b>PBMC</b> LIBRARY	77
FIGURE 8. V GENE RECOVERY AND CHAIN SHUFFLING FROM CLONES 2G7 AND 5B5	79
FIGURE 9. CREATION OF THE VH – $lpha$ Antibody Library from the CMCs of	
ML 1356 AND ML 1515	80
FIGURE 10. THE DIVERSITY AND V GENE COMPOSITION OF THE CMC-R VH - $lpha$ Library in	
P-DBIGG1-TM	82
FIGURE 11. ASSESSMENT OF ANTIBODY EXPRESSION ON THE CELL SURFACE	83
FIGURE 12. AFFINITY OF CELL SURFACE ANTIBODY FOR GP120 AND GP41	85
FIGURE 13. ISOLATION OF GP41 SPECIFIC ANTIBODIES BY FACS	86
FIGURE 14. ANALYSIS OF RECOVERED VH AND VL GENES FROM SORTED GP41 POSITIVE CELLS	88
FIGURE 15. IDENTIFICATION OF PRODUCTIVE VH/VL COMBINATIONS VIA CHAIN SHUFFLING	89
FIGURE 16. SEQUENCES OF RECOVERED GP41 SPECIFIC VH (A) AND VL (B) CLONES	91
FIGURE 17. ELISA TITRATIONS OF THE GP41 MABS AGAINST GP41, GP120, AND BSA	92
FIGURE 18. ELISA TITRATIONS OF THE GP41 MABS AGAINST REDUCED/DENATURED GP41	93
FIGURE 19. BINDING OF GP41 MABS TO HIV-1 ENVELOPE EXPRESSED ON THE SURFACE OF	
293F CELLS	95
FIGURE 20. TITRATION OF MABS VS. HIV-1 ENVELOPE SPIKE (CLADE B AND C) USING	
FMAT TECHNOLOGY	97
TABLE 1. GENE CHARACTERISTICS OF GP41 SPECIFIC VH (A) AND VL (B) SEQUENCES	99

# LIST OF ABBREVIATIONS USED

ADCC	ANTIBODY-DEPENDENT CELLULAR CYTOTOXICITY
AIDS	ACQUIRED IMMUNE DEFICIENCY SYNDROME
ART	ANTI-RETROVIRAL TREATMENT
AFC	ANTIBODY FORMING CELL RESPONSE
BHK	BABY HAMSTER KIDNEY CELLS
вNт мАв	BROADLY NEUTRALIZING MONOCLONAL ANTIBODY
CHR	C-TERMINAL HEPTAD REPEAT
СМС	Cervical Mononuclear Cell
CMC-R	<b>CERVICAL MONONUCLEAR CELLS FROM A HIV-1 RESISTANT INDIVIDUAL</b>
CDR	COMPLEMENTARY DETERMINING REGION
СТ	Cytoplasmic Tail
CTL	<b>C</b> ΥΤΟΤΟΧΙC <b>T L</b> ΥΜΡΗΟCΥΤΕ
DTT	DITHIOTHREITOL
ENV	Envelope
ELISA	ENZYME-LINKED IMMUNOSORBENT ASSAY
EU	Exposed yet Uninfected
FDA	FEDERAL DRUG ADMINISTRATION
FACS	FLUORESCENCE-ACTIVATED CELL SORTING
Fab	FRAGMENT ANTIGEN-BINDING REGION
FC	FRAGMENT CRYSTALLIZABLE REGION
FR	Framework Region
FP	FUSION PEPTIDE
GAL CER	GLYCOSPHINGOLIPID GALACTOSYL CERAMIDE RECEPTOR
СН	HEAVY CHAIN CONSTANT REGION
DH	HEAVY CHAIN DIVERSITY REGION
JH	HEAVY CHAIN JOINING REGION
VH	HEAVY CHAIN VARIABLE DOMAIN
HEPS	HIGHLY EXPOSED YET PERSISTENTLY SERONEGATIVE
HEK	HUMAN EMBRYONIC KIDNEY CELLS
HIV-1	HUMAN IMMUNODEFICIENCY VIRUS TYPE-1
IGA	Immunoglobulin A
IGKC	IMMUNOGLOBULIN KAPPA CONSTANT REGION
VL	Light Chain Variable Domain
MPER	MEMBRANE PROXIMAL EXTERNAL REGION
мАв	MONOCLONAL ANTIBODY
MALT	MUCOSA - ASSOCIATED LYMPHOID TISSUE
NHR	N-TERMINAL HEPTAD REPEAT
PBMC	PERIPHERAL BLOOD MONONUCLEAR CELL
PDGFR	PLATELET DERIVED GROWTH FACTOR RECEPTOR
PR	POLAR REGION
R	Replacement
RT	<b>Reverse Transcriptase</b>
S-IGA	Secretory IgA
STI	Sexually Transmitted Infection
S	SILENT
SHIV	Simian – Human Immune Deficiency Virus
ТМ	TRANSMEMBRANE DOMAIN

#### Abstract

The mucosal antibody repertoire of the cervical mucosa in commercial sex workers from Nairobi, Kenya, who are highly sexually exposed to human immune deficiency virus type-1 (HIV-1) but remain persistently IgG seronegative (HEPS), may represent a novel source of broadly neutralizing monoclonal antibodies (mAbs) against HIV-1. Mucosal IgA specific for HIV-1 envelope (Env) subunit gp41 has been suggested as a correlate of protection in HEPS individuals. The in depth studies at both the gene and function level required to confirm their role in HIV-1 resistance are possible only using recombinant monoclonal IgAs. Human mAbs have traditionally been selected from libraries displayed on the surface of microorganisms (phage, yeast). However, due to inherent limitations, such techniques may not be optimal for isolating such rare mAbs from a pool of cervical B cells. We have developed an antibody selection system based on surface display on mammalian cells and used this technology to isolate four novel monoclonal antibodies, against linear epitopes on gp41, from the IgA repertoire of the cervical mucosa in Kenyan HEPS. Furthermore, three of the four mAbs were shown to bind with surface expressed consensus clade B and clade C Env on mammalian cells. Characterization of the variable region cDNA of the two strongest binding mAbs reveals extensive somatic mutations with a bias of replacement mutations clustering in the complementary determining regions (CDR) indicating antigendriven affinity maturation had occurred. Affinity matured monoclonal IgAs, such as these, may play a role in the identification of new, vulnerable epitopes on HIV-1, or act as a component in a topical microbicide.

ix

#### **1** Introduction

#### **<u>1.1 Global Impact of HIV-1</u>**

Despite over 25 years of research, human immunodeficiency virus - type 1 (HIV-1) still represents a major health burden worldwide. During those 25 years, over 25 million people have died from the disease caused by HIV-1, acquired immunodeficiency syndrome (AIDS), and over 33 million people are currently infected [1]. Sub-Saharan Africa has been hardest hit by the pandemic, accounting for nearly two thirds of the individuals living with, and three quarters of all deaths resulting from AIDS [1]. Deplorably, over half of the individuals who acquire AIDS, are infected with HIV-1 before the age of 25, and die within the first 10 years [1]. Due to the great strides made in HIV-1 research and prevention in the past quarter century, the rate of new HIV-1 infections has dropped off in most countries; however, these trends are offset by new infections primarily occurring in regions of Sub-Saharan Africa. Globally there were 2.7 million new HIV-1 infections in 2007; 1.9 million of these cases were reported in Sub-Saharan Africa [1]. Because the majority of new HIV-1 infections arise in such low-income nations, the preponderance of individuals who are becoming infected with the virus do not have access to antiretroviral therapy. Clearly HIV-1 is still a leading global health problem and new innovative research is required to both improve treatment and develop a safe and effective HIV-1 vaccine.

#### 1.2 HIV-1 Virology

AIDS was first described in homosexual men living in the United States in 1981 [2]. French and American scientists later confirmed HIV-1 as the causative agent for AIDS in 1983 and 1984 respectively [3,4]. HIV-1 belongs to the *Lentivirinae* subgroup of *Retroviridae* and has spherically shaped virion with a diameter of approximately 100 nm [5]. The genome consists of two ribonucleic acid (RNA) molecules in a complex with nucleoproteins and surrounded by a lipid bilayer envelope covered with approximately 72 glycoprotein spikes [5].

The HIV-1 genome consists of three major genes: *gag*; *pol*; and *env*. The initial Gag polypeptide is a 55 kD precursor protein that, during maturation, is cleaved into several smaller proteins including: matrix; capsid; p2; p6; and nucleocapsid [5]. The *pol* gene encodes for three enzymes: protease, reverse transcriptase (RT) and integrase [5]. The env gene encodes the glycoprotein spike that decorates the viral envelope. The initial *env* transcript is translated into a precursor protein (gp160) that is glycosylated in the endoplasmic reticulum and has the apparent molecular weight of 160 Kd. Gp160 is then cleaved by host cellular enzymes into two mature Env proteins: gp41 and gp120. Both proteins assemble into trimers coming together at the plasma membrane forming the viral spike [5]. A hydrophobic transmembrane domain allows gp41 to anchor to the viral membrane and interact non-covalently with the extracellular gp120 [5]. HIV-1 also has numerous accessory proteins that play critical roles in the viral lifecycle including, Tat, Rev, Nef, Vpr, Vif, and Vpu.

#### **<u>1.3 Diversity of HIV-1</u>**

The extraordinary genetic diversity of HIV-1 has presented an abundance of problems to researchers attempting to generate a universal vaccine. Part of this variability can be attributed to the fast replication cycle of the virus, which can result in the production of up to  $10^{10}$  virions per day [6]. As a retrovirus, HIV-1 uses the enzyme RT to convert its own RNA genome into DNA once inside the host by taking advantage of the host cellular machinery for DNA replication, transcription, and translation. Replicated DNA is converted back into RNA for packaging into progeny virions once translation has occurred. While mutations can arise at any point, they most often occur during reverse transcription due to the high error rate of the viral RT (3 x 10<sup>-5</sup> /nucleotide base per cycle) attributable to the lack of proof reading activity [6,7]. Variability is further compounded by the recombination of different types of HIV that infect the same cell [6]. As a consequence of these different mechanisms, many different variants of HIV-1 can be isolated from one individual in just a single day resulting in 1% genetic variation per year. The evolution of quasispecies results in a high rate of antigenic variation and correspondingly, poses an enormous challenge for host immunity.

HIV-1 can be divided into three main types: M, N, and O. Type N, being the least common, is only found in Cameroon, type O is responsible for relatively minor epidemics, while type M is responsible for the current pandemic (6). Based on mutation clock calculations, it is believed that HIV-1 type M crossed from chimpanzee into man in the early 1930's [6]. Following the rapid expansion of the type M population, the virus diversified into 10 subtypes, or clades: A, B, C, D, F<sub>1</sub>, F<sub>2</sub>,

G, H, J, and K [6,8]. Four of these clades, A, B, C, and D make up the vast majority of new HIV-1 infections. Clade B is the predominant clade found in North America and Europe; clade C, responsible for approximately 50% of infections world wide, is the dominant clade in Southern Africa, India and Nepal; clade A is common in Western Africa; while clade D can be found in Eastern and Central Africa [8]. Different clades of HIV-1 are often found in the same area; for instance, clades A and D are primarily responsible for the HIV-1 epidemic in Nairobi, Kenya [7].

#### **<u>1.4 HIV-1 Life Cycle and Immunopathogenesis</u>**

#### **1.4.1 Early Discoveries**

Not long after the discovery of AIDS in 1981, Klatzmann et al, (1984) showed that *in vitro*, HIV-1 was able to replicate and cause a cytopathic effect in CD4+ T cells, but not in CD8 T cells [9]. This was supported by the observation that AIDS was associated with the decline of CD4+ T cells in the blood [2]. Additional work by the Klatzmann group demonstrated that it was the CD4 antigen itself that acted as the receptor allowing for viral invasion of these CD4 positive T helper cells [10]. Several years later, using transfection studies of CD4 cDNA, Maddon et al (1986) showed that additional receptors other than CD4 were required on the cell surface for HIV binding [11]. It wasn't until some 10 years later that it CXCR4 was identified as the co-receptor necessary on T cell lines, and CCR5 was identified as the coreceptor necessary on peripheral blood mononuclear cells (PBMC) and macrophages [12,13].

The association of HIV with dendritic cells, initially met with controversy, was first suggested by Knight and Macatonia (1988) [14,15]. The identification of two types of dendritic cells, myeloid, and plasmacytoid, allowed Patterson et al (2001) to demonstrate that myeloid dendritic cells express CD4 and CCR5 supporting HIV binding and replication, while plasmacytoid dendritic cells allow HIV to bind to their DC-SIGN receptor [16]. Binding to dendritic cells allows the virus to be escorted to local lymph nodes where it will encounter the susceptible CD4+ T cells required for viral proliferation.

#### 1.4.2 Typical Course of HIV Infection

HIV infection via the mucosal or parenteral route initially results in high viremia [15]. CD4+ T cells, macrophages and dendritic cells at the site of infection migrate to local lymph nodes delivering virus to CD4+ T cells. The most active site of HIV replication is the CD4 cell rich, mucosa - associated lymphoid tissue (MALT) of the gut [17]. After the initial infection, the viral load declines to a level that can be indicative of the rate at which the individual will progress to AIDS [18]. This decline in viremia has been attributed to both strong cell-mediated immunity and the circulation of HIV-1 neutralizing antibodies [15]. In addition to the neutralizing antibody response, non-neutralizing gp120 and gp41 antibodies may play a role in the decline of the primary infection by either inducing the host complement response, or antibody dependent cellular cytotoxicity [19,20].

Following seroconversion and the partial clearance of the primary infection, there is a long clinically asymptomatic period with fairly stable CD4+ T cell counts

[15]. During this period, CD4 cell death is balanced by a remarkable increase in CD4 cell restoration [21]. Eventually, immune regeneration can no longer offset the cell death caused by the virus, and CD4+ T cell numbers drop below a threshold level of 200 cells/ $\mu$ l [15]. At this point, the immune system of the patient has been depleted to the point where opportunistic infections will likely result in death.

# 1.4.3 HIV Infection of the Female Genital Tract

Approximately 30-40 % of new HIV-1 infections in women are acquired through vaginal intercourse [22]. Free virions are able to establish mucosal infection of the either the vagina, or the ecto and endo cervix [22]. Initially, the thick mucus layer of the female genital tract is able to trap free and cell-associated virions prolonging the contact time of the virus with the cells of the mucosa [22]. It is at this time that the virus is most susceptible to antiviral substances such as secretory IgA [22].

Infection with HIV at the mucosal surface is very different from parenteral transfer. HIV is able to invade the mucosal epithelium using several different pathways; therebye ensuring that even if one pathway is blocked, the virus is still able to initiate a productive infection. One such pathway is the capture and internalization of HIV virions by Langerhans cell in the epithelium, which then transport the virus to the stroma [22]. Once in the stroma, HIV can infect stromal dendritic cells which then circulate to CD4 cell rich regional lymph nodes where massive viral replication can occur [22]. HIV may also infect intra-epithelial CD4+ T cells directly in the vaginal mucosa, or simply migrate through physical abrasions in the epithelium caused by intercourse [22]. The virus is subsequently is taken up by

lymphatic or venous microvessels and transported to either regional lymph nodes or into peripheral blood [22]. The virus also has the ability to either transcytose through epithelial cells in the basal layer of the vaginal squamous epithelium, productively infect these epithelial cells, be endocytosed into cytosolic compartments, or penetrate the spaces between the epithelial cells [22].

#### **<u>1.5 Structure and Function of the HIV-1 Envelope</u></u>**

#### 1.5.1 Native HIV-1 Envelope Spike

The HIV virion is surrounded by a lipid bilayer envelope covered with gp120gp41 spike complexes (Figure 1). These complexes initially come together through weak noncovalent interactions at the surface of infected cells and through the budding process, are incorporated into the virus envelope [23]. Experimental evidence, largely based on scanning transmission electron microscopy images of fixed HIV glycoproteins, has suggested that the functional envelope spike (Env) is a heterodimeric trimer complex of gp120 and gp41 (Figure 1) [24]. However, there is also evidence suggesting there are other forms of envelope present on the viral surface. Kuznetsov et al (2003) was unable to demonstrate the presence of uniform trimeric envelope species on HIV virions using atomic force microscopy, while Herrera et al (2005) showed that antibodies unable to neutralize virus in solution, are able to capture virus onto ELISA plates [25, 26]. The appearance of these nonfunctional envelope species is likely the result of inefficient trimerization of the subunits in the Golgi, or the dissociation of gp120 from gp41, leaving uncapped



Figure 1. The HIV-1 virion and the envelope glycoprotein trimer complex. A) Simplified depiction of a particle of HIV-1. The Env trimers are the sole target of neutralizing antibodies, while cellular responses may target the core components [29]. B) A single, functional HIV-1 envelope spike composed of gp120 and g41 trimers. The spike interacts with CD4 receptor along with a co-receptor on host cells and initiates virus entry into host cells [29].

gp41 trimers decorating the viral envelope [23, 27]. The HIV spike is extraordinarily diverse differing by up to 35% between viral clades, and up to 20% within each clade [8]. This incredible antigenic diversity allows the virus to evade the host immune response, and has hampered vaccine attempts for decades.

# 1.5.2 Structure of gp120

The exposed region of Env, gp120, is composed of five relatively conserved regions (C1-C5), and five highly variable loops (V1-V5) [29]. In addition to their variability, the variable loops are highly glycosylated and able to shield the more conserved underlying structures from host antibodies [29]. One relatively conserved region of gp120, the CD4 binding pocket, is located deep within the inner domain [29]. For cell entry, the CD4 molecule on the target cell must bind deep within the CD4 binding pocket resulting in a conformational shift in gp120 that reveals a second conserved site: the coreceptor binding site [29]. The co-receptor binding site associates with either CCR5 or CXCR4 on the target cell initiating a second conformational shift, this time in the gp41 subunit [29].

#### 1.5.3 Structure of gp41

The Env protein gp41, is a 345 amino acid (aa) glycoprotein with a mass of 41 kDa [30]. The structure gp41 is depicted in figure 2. Likely for conservation of function, gp41 is more conserved than gp120 and is not typically subdivided into distinct V and C regions. It can, however, be divided into three major domains: the extracellular domain (aa 512 to 683); the transmembrane domain (TM) (aa 684 to

705); and the cytoplasmic tail (aa 704 to 856) [30] (Figure 2). Numbering of Env aa residues is based on numbering in HXB2 [31]. The extracellular domain, being the only domain accessible to the target cell, can be subdivided into regions that are functionally important in the fusion of the viral and host cell membranes. These include: (i) the fusion peptide (FP), the N terminal hydrophobic region; (ii) the polar region (PR); (iii) two alpha helical structures termed the N-terminal heptad repeat (NHR) and the C-terminal heptad repeat (CHR); (iv) the membrane spanning domain (TM);(v) and a tryptophan rich region next to TM termed the membrane - proximal external region (MPER) (aa 660 to 683) (Figure 2) [30].

The FP corresponds to the first 15 aa of gp41 and through mutational studies, has been shown to be involved in triggering fusion between the viral and target cell membranes [30, 32, 33, 34]. Due to its oblique nature, insertion of FP into the cell membrane causes destabilization of the lipid bilayer permitting membrane fusion [35]. Many enveloped viruses use FPs to mediate membrane fusion following a trigger such as a change in pH or coreceptor binding.

The NHR and the CHR both play an important role in initiating the conformational change in gp41 following gp120 binding to the co-receptor. During this conformational change, three NHRs form a bundle anti-parallel to three helical CHRs, together forming a six-helix bundle [30]. The creation of the six-helix bundle is essential for the formation the pore that permits the entry of the HIV nucleocapsid into the target cell [36].

The TM region of gp41 serves to secure the entire Env spike to the viral membrane. The structure of TM has yet to be elucidated; however, a computer

model proposed by Gallaher et al (1989), has suggested it adopts an  $\alpha$ -helical structure [37]. This region is among the most highly conserved between different HIV clades and there is controversy surrounding whether or not it participates directly in the virus-cell fusion process [30].

The MPER, located adjacent to the TM region, is another region of gp41 that is highly conserved among clades of HIV [30]. The structure of MPER was originally predicted as an  $\alpha$ -helix by Gallaher et al (1989); however, unlike the TM, additional information has since come out suggesting an alternative structure [37, 30]. Structural analysis of MPER in complex with the monoclonal antibody 2F5 (reviewed later), suggests MPER exists in an extended conformation with a  $\beta$ -turn at its core [38,39]. This discrepancy may be attributable to a conformational shift that occurs in the MPER during virus-cell fusion [40]. The idea that MPER plays a vital role in HIV infection is supported by the following observations: it is highly conserved among isolates [30,41]; epitopes of several of the most potent HIV-1 broadly neutralizing monoclonal antibodies (bNt mAbs) can be found in the MPER [30,41]; mutational studies with MPER have a deleterious effect on virus-cell fusion [30, 42, 43]; and MPER binds with the galactosyl ceramide receptor (Gal Cer) permitting transcytosis of the virus across the mucosal epithelium [4, 30, 44].

Gp41 itself can be found in either three different conformations depending on the stage of the fusion process. Little is known about the pre-fusion structure, however, it is believed that the majority of gp41 domains (the NHR and the CHR) fold together and are largely obscured by gp120 in the native spike [45]. The structure of the fusion intermediate state is also obscure. Based on the crystal



Figure 2. Structure of HIV-1 envelope protein gp41. A, Schematic diagram of linear gp41 with important structural regions indicated; the MPER region is highlighted in yellow, disulfide bonds are indicated with horizontal lines [83]. B, estimated structure of gp41 depicting the orientation of the structural regions indicated in A in native gp41 [83]. The reported immunodominant region is outlined in black between the NHR and CHR and contains the di-sulfide bond.

structure of gp41 peptides with different mAbs, it is has been postulated that gp41 adopts an extended bundle of three  $\alpha$ -helixes, partially stabilized by its association with gp120, that is capable of partially inserting FP into the lipid bilayer of the target cell [24].

In contrast to the pre and intermediate fusion states, the structure of the post-fusion state has been well established [24]. As stated above, the conformational shift in gp41 initiated by gp120 binding to the coreceptor allows the CHR and NHR to form the stable six-helix bundle [30]. This model suggests that the action of the unstable extended three-helix rod transitioning to the highly stable six-helix bundle draws the target cell membrane closer to the viral membrane, allowing for complete insertion of FP into the cell [24]. The destabilization of the cell membrane is followed by the formation of a fusion pore by MPER, CT, and possibly portions of TM allowing for the insertion of the nucleocapsid into the target cell [24]. While there are multiple, well-known domains on gp41, the limited surface accessibility of most of these limit the gp41 neutralizing antibody response mainly to the few regions that are surface exposed on the native Env spike.

### 1.6 Progress Toward an HIV Envelope Based Vaccine

#### **1.6.1 Cell-Mediated or Humoral Immunity**

Antibodies are the only host defense capable of providing sterile immunity and providing complete protection from infection. The failure to develop a successful HIV-1 vaccine has to be considered the biggest disappointment in the

field of HIV research during the past 25 years. Many different strategies have been attempted from DNA vector-based vaccines, to vaccines designed to elicit cellular immunity, to Env subunit vaccines designed to elicit humoral immunity. None of these strategies has shown considerable effectiveness in field trials [46,47]. An efficacious HIV vaccine will no doubt require both cell-mediated and humoral immunity against HIV-1, as the majority of successful viral vaccines work by preventing infection using neutralizing antibodies, and stimulating cell-mediated immunity to eliminate any remaining infected cells [92]. Complete or sterilizing immunity from HIV-1 infection will depend on the ability of the vaccine to elicit broadly neutralizing antibodies against HIV-1, in particular at the mucosal surface. The fact that only the presence of neutralizing antibodies has proven capable of preventing HIV/SHIV (simian-human immunodeficiency virus) infection in animal models validates the role of the humoral immune response in providing protection against HIV-1 [48-52]. Furthermore, the most commonly cited goal of CTL or T cell based vaccines is not to induce complete immunity, but to delay the onset of AIDS, for once the virus infects the CD4+ T cell population there is no hope to eradicate the virus from all reservoirs despite antiretroviral treatment [9].

# **1.6.2 Envelope Vaccines**

Researchers have long been aware that ability to elicit humoral immunity is an essential component in the development of any HIV-1 vaccine. Indeed, Env based vaccines of either gp160, or the gp120 subunit were among the first attempts in developing an HIV vaccine in the late 1980's [53]. Initial results from these trials

seemed promising, as antibodies were produced against gp120 in all recipients, with the majority also showing neutralizing potential [54-59]. However, it was quickly realized that these antibodies demonstrated little activity against naturally occurring HIV-1 strains, and did not protect against HIV infection [29]. Over 20 years later, clinical trials with recombinant gp120 have yet to produce a positive result that validate this approach [29].

Explanations for the failure of the recombinant gp120 vaccines are numerous. The most obvious being that the immunogens used did not faithfully reproduce the native structure of the viral envelope [29]. Other explanations are based on the nature of the viral envelope itself and include: the closed nature of Env hides conserved neutralizing epitopes; the heavily glycosylated nature of gp120 prevents antibody mediated neutralization; the outer V loops of gp120 can tolerate multiple mutations; the inherent genetic variability of *env* among different viral clades; epitopes crucial for virus-cell fusion (i.e CD4 binding site) are inaccessible until the conformational shift mediated by target cell binding; and the fact that gp120 is shed from the virus inducing antibodies to epitopes irrelevant in neutralization [29].

Even with the assortment of mechanisms by which HIV is able to avoid antibody - mediated neutralization, there is still hope that broadly neutralizing antibodies can be elicited by vaccination and provide protection from HIV-1. The isolation of several human monoclonal antibodies (mAbs) that are capable of neutralizing a broad spectrum of HIV-1 isolates has confirmed that there are broadly conserved neutralizing epitopes present on the HIV envelope [29]. MAbs 2F5, 4E10, and Z13 are well studied antibodies that bind to the MPER of gp41 [41,51,61,62], mAb b12 is

able to bind the CD4 binding site of gp120 [63] and mAb 2G12 binds to a conserved carbohydrate residue in the glycan shield of gp120 [88]. Indeed, Hessel et al (2009) has shown that passive transfer of gp41 mAbs 2F5 and 4E10 can provide complete protection against mucosal SHIV challenge in rhesus macaques [144]. Recent studies have also shown that passive transfer of mAbs 2G12 or b12 to macaques provided protection against repeated low-dose mucosal SHIV challenge even at low neutralizing titers [89,90]. Neutralizing titers of these mAbs in macaques were at or below those commonly observed in the sera of HIV-infected individuals against various clades of HIV-1 [90,91]. This suggests that the antibody titers achieved in these studies can be generated in humans provided an appropriate immunogen is used [90].

Several strategies aimed at inducing these neutralizing antibodies by vaccination are being investigated. Fouts et al (2000, 2002) has attempted to generate altered forms of gp120 that expose cryptic neutralizing epitopes by immunizing with a CD4 like molecule bound to gp120 [64,65]. While neutralizing antibodies are generated, it is not clear how relevant these antibodies are to naturally occurring viral isolates [64,65]. Others have attempted to use fusion intermediate complexes to generate antibodies to conserved epitopes only presented during the fusion process [66]. After some initially positive results, this strategy has since fallen out of favor due to poor reproducibility [29]. Pantophlet et al (2004) has hyperglycosylated regions of gp120 in order to mask undesired epitopes generating a gp120 mutant that preferentially binds to mAb b12 [67]. Immunization of this mutant would theoretically concentrate the antibody response

to the b12 epitope (CD4 binding site) [67]. Frey et al (2008), generated a trimeric Env with gp41in the extended fusion intermediate state and showed that it preferentially binds with bNt mAbs 2F5 and 4E10 [68]. Other groups are trying to develop soluble Env constructs that accurately represent the native viral spike [69,70]. The clear correlation between virus neutralization and antibody binding to native Env lends credence to this approach [71, 72].

The obvious potential of using rationally designed immunogens to elicit neutralizing antibodies requires additional investigation; however, the discovery of new bNt mAbs is required for the discovery of novel conserved, neutralizing epitopes on Env to improve the breadth and potency of the antibody response elicited by vaccination. In this regard, the study of the cervical mucosal immune response represents a novel source of B cell diversity.

#### **<u>1.7 Host Resistance to HIV-1</u>**

A little over a decade following the emergence of HIV-1 on the global scene, it became apparent that not all individuals exposed to the virus become persistently infected [73]. Definitively establishing the existence of individuals who are protected against HIV-1 proved to be a difficult task, as the only option was to observe individuals who were continuously exposed to the virus yet remained uninfected [74]. One of the first studies to demonstrate the phenomena of HIV resistance was done on a cohort of 424 initially HIV-1-seronegative commercial sex workers in Nairobi, Kenya [73]. Fowke et al (1996) followed these women from

1985 to 1994 and collected data on sexual behaviors, such as frequency of condom usage and the number of different partners, while monitoring their HIV-1 status using standard ELISA and immunoblot techniques [73]. Data analysis revealed that these women had an estimated 64 unprotected sexual exposures to HIV-1 per year [73]. Consequently, an extremely high rate of seroconversion to HIV-1 was observed during the first 2 years of the study [73]. Upon further investigation, they noticed that instead of continuing to rise, HIV-1 prevalence in the cohort reached a plateau, with new incidences of infection virtually non-existent in a small subset of these women (13% of cohort) [73,74]. Despite the fact that the number of annual exposures to HIV-1 has increased over time for these women, they have remained HIV-1 seronegative for more than 20 years [73].

The establishment of additional cohorts of highly exposed persistently seronegative (HEPS) individuals has only added to the drive to understand the mechanisms responsible for providing HIV-1 protection [75]. Since the correlates of immune protection from HIV-1 have yet to be elucidated, understanding how HEPS individuals are protected from HIV-1 is imperative.

Numerous studies on the Nairobi cohort have ruled out the following factors in resistance to HIV-1: a natural resistance to other STIs; mutations or altered gene expression of the CCR2, CCR5, or CXCR4 chemokine receptor; natural killer or lymphocyte activated killer cell activity; innate cellular resistance mechanisms; neutralizing IgG or other factors in the plasma; and the presence of mitogen activating β-chemokines in the plasma [74]. Other studies have focused on the role played by cell-mediated immune response in providing protection against HIV-1 in

the Nairobi cohort [74]. Rowland Jones et al (1998) demonstrated that CTLs with cross-clade specificity to HIV-1 exist in HEPS women [76], while Fowke et al (2000) showed that the CD4+ T cells from HEPS women, when exposed to Env, show increased production of IL-2 when compared with CD4 cells from HIV infected and non-infected individuals [77]. Other groups have postulated a role for HIV specific mucosal IgA in resistance to infection, and it is this theory that will be explored in further detail.

### 1.8 Role of IgA in Mucosal Immunity in the Genital Tract

### **1.8.1 IgA in the Genital Tract**

An important part of mucosal defence against sexually transmitted pathogens is the presence of secretory IgA (S-IgA) antibodies in the external secretions of the genital tract. S-IgA exists as immunoglobulin dimmers that are protease resistant and have been shown to provide protection from many pathogens including viruses [78]. While not understood in terms of homing or molecular models, the compartmentalization of the immune system of the genital tract can causese antibodies produced locally in cervicovaginal secretions to be absent from the serum antibody response [78]. The induction mechanisms of the immune response in the genital tract have yet to be fully elucidated [78]. However, it is thought that intraepithelial Langerhan's cells present antigens to both B and T lymphocytes at local lymph nodes, or directly to local lymphocytes present in the genital tract mucosa [79]. The observation that vaginal immunization induces a

stronger local antibody response when compared with other mucosal sites, suggests that the activated lymphocytes express yet unidentified adhesion receptors that mediate homing of lymphocytes back to the genital tract from local lymph nodes [79].

#### 1.8.2 Possible Mechanisms of IgA Mediated Protection from HIV-1

While S-IgA antibody mediated immune protection mechanisms have been shown to play an important role the defence against many pathogens, their role in protecting against HIV-1 infection in the genital tract is less clear. One possible mechanism of mucosal antibody mediated protection from HIV-1 infection is by cross-linking virions with non-neutralizing Env specific antibodies; a process referred to as immune exclusion [78]. Cross-linking traps microbes in the thick mucus layer of the genital tract, preventing contact with the mucosal surface and viral promoting clearance [78]. This mechanism has been correlated with protection from several other mucosal pathogens. For example, Burns et al (1996) showed that non-neutralizing IgA mAbs were able to provide protection from rotavirus infection in mice by preventing mucosal attachment [80]. Whether immune exclusion is relevant to HIV infection is unknown [81].

A second mechanism of IgA mediated protection from HIV-1 may be antibody mediated inhibition of viral transcytosis across the epithelial cell layer. Inhibiting transcytosis would prevent the virus from reaching the T cell abundant lamina propria and inhibit viral spread. Alfsen et al (2001), has shown that mucosal S-IgA and not IgG from HIV-1 infected individuals, is able to block HIV-1 transcytosis by

preventing gp41 binding to the glycosphingolipid galactosyl ceramide (Gal Cer) and disrupt the formation of the lipid raft microdomains that mediate transcytosis and endocytosis across the phosphilipid rich cell membrane [82]. Additional immunological studies established that the S-IgA targeted the MPER of gp41, in particular, the conserved ELDKWA hexapeptide that is also the target of the broadly neutralizing gp41 mAb 2F5 as shown by Muster et al (1993) [82,83,41].

A third mechanism of protection may be IgA mediated neutralization of the virus during transcytosis through epithelial cells [84]. Huang et al (2005) used pairs of IgA and IgG gp120 and gp41 neutralizing mAbs to demonstrate that mAbs of the IgA isotype ware able to act intracellularly to inhibit virion production by epithelial cells [84]. They proposed that vesicles containing internalized Env specific IgA intersect in the secretory pathway with post-Golgi vesicles carrying new Env proteins and neutralize virus production [84].

A fourth mechanism whereby IgA may provide some protection from HIV-1 infection is through antibody-dependent cellular cytotoxicity (ADCC). While nonneutralizing antibodies with ADCC activity have been shown to provide protection in other viral models, the role played by IgA induced ADCC in the immune response against HIV-1 is supported by considerably less evidence. Colostral and serum IgA from HIV-1 infected individuals has been shown to lyse gp120 coated cells in the presence of PBMCs *in vitro* [78]. Connick et al (1996), suggested that ADCC plays a role in controlling HIV-1 replication early in acutely-infected individuals, while Baum et al (1996) correlated high levels of ADCC inducing antibodies with slower progression to AIDS [85,86]. Perhaps the strongest evidence comes from Duval et al

(2008), who showed that an anti-gp41/anti-CD89 (IgA Fc receptor) bispecific mAb can direct the destruction of HIV-1 by CD89 positive neutrophils which are highly prevealent in the cervix [87].

#### **1.8.3 IgA in HEPS Subjects**

While there is evidence supporting a possible role for mucosal S-IgA in the immune response against HIV-1, the role played by mucosal IgA in mediating resistance against the virus in HEPS individuals has been the subject of much debate. Kaul et al (1999) was among the first to correlate levels of cervical HIV-1 specific IgA to host resistance to infection in the Nairobi Cohort [93]. In that study, HIV specific IgA was observed in 16 of 21 HEPS sex workers, 5 of 19 infected sex workers, and 3 of 28 low risk HIV-1 negative women [93]. HIV-1 specific IgA has also been isolated from cervicovaginal lavage fluids in 50% of female HEPS sex workers in northern Thailand [94], and has been correlated with protection in partners of HIV-1 infected individuals who remain uninfected despite repeated exposure (EU) in Cambodia [95].

It was later shown that IgA purified from the genital tract, saliva, and plasma of HEPS sex workers in Nairobi was able to neutralize a clade B HIV-1 virus isolate [98]. A second study by Devito et al (2002) showed that the IgA in mucosal and systemic compartments of HEPS sex workers in Nairobi showed significantly more cross clade HIV-1 neutralizing ability than the IgA purified from EUs with HIV-1 clade B infected partners [97]. Mucosal and plasma IgA purified from 3 of 6 HEPS

sex workers in Nairobi could also inhibit the epithelial cell transcytosis of a HIV-1 clade B isolate [98].

In addition to differences in total antibody levels, differences in the epitope specificity of serum IgA in HEPS individuals compared with HIV-1 infected individuals has also been observed. In a study of 6 EU individuals, 6 HIV-1 infected women, and 6 healthy controls, Clerici et al (2002) demonstrated that IgA from EU subjects exclusively recognized gp41, while IgA from HIV infected individuals recognized both gp120 and gp41 [99]. Furthermore, the IgA from the infected subjects bound with the immunodominant regions in gp41 (aa 589 – 618 and 642-673), while IgA from EU subjects recognized a restricted epitope (aa 582 – 589) (QARILAV) located in the  $\alpha$  - helical region of gp41 that is accessible to antibody only after the binding of gp120 to CD4 on the target cell [99]. Higher levels of IgA specific to gp41 have also been observed in EU subjects when compared with HIV-1 negative individuals in Cambodia, Italy, and in a cohort of intravenous drug users in Vietnam [100].

There is also evidence suggesting IgA does not play a major role in the HIV-1 resistance observed in HEPS individuals. HIV specific IgA could not be detected in the cervicovaginal secretions of 24 EU subjects in California [96]. Ghys et al (2000) could only detect anti-HIV antibodies in the cervix of 7.3 and 29.8% of HEPS sex workers in the Abidjan, Cote d'Ivoire [101], and Lucy et al (2000) detected no significant levels of IgA or IgG in the vaginal secretions of HEPS sex workers in the Gambia [102]. Considerable differences in the exposure levels to HIV-1 may account for the lower levels of HIV-1 specific cervical IgA, as it is estimated that sex workers

in the Gambia have only one unprotected sexual encounter with an HIV-1 infected man per month [102]. Due to these conflicting results, Horten et al (2009) carried out a comprehensive analysis of HIV-1 specific IgA and IgG levels in the cervical lavages of HEPS and HIV-1 infected sex workers in the Nairobi cohort [103]. While HIV-1 specific IgA was found in a number of the resistant women, it was not found at a higher frequency than the HIV-1 infected or uninfected women, suggesting that IgA does not play a significant role in long term protection against the virus [103]. However, because antibody levels were detected using a conventional ELISA against a HIV-1 virus lysate, the study offers no insight into differences in the gp41/g120 binding profile of the different antibody populations, differences in the epitopes recognized, differences in antibody affinity, nor any functional differences between the IgA from HEPS individuals and the IgA from HIV infected individuals as has been shown previously [99].

The conflicting results addressed above are likely due to the difficulties in detecting IgA in the notoriously variable environment of the cervix, differences in antibody detection methods, differences in the antigens used, and variations in the characteristics of the different HEPS populations studied (level of exposure to virus, opportunistic infections, frequency of protected/unprotected intercourse, mode of virus exposure, time of menstral cycle, hormone level, etc.) [104]. Due to these variations, studies at the molecular level using mAbs are required for further characterization cervical IgA [104].

Using a technique called phage display (discussed below) Berry et al (2003) was the first to isolate a HIV-1 neutralizing IgA mAb against gp120 from the cervix

of HEPS sex workers in Nairobi, Kenya [105]. In a recent study, Tudor et al (2009) also used phage display to isolate gp41 specific neutralizing IgA mAbs from the cervix of HEPS sex workers in Cambodia that are able to inhibit the in vitro transcytosis of HIV-1 clade B and C virus isolates [104]. Interestingly, IgG class switched variants of the same mAbs were not as efficient at blocking transcytosis of the virus, indicating a potential role for the IgA constant region [104]. Molecular analysis of the antibody genes revealed extensive somatic mutations and a high replacement to silent (R:S) mutation ratio in the CDR regions suggesting that these antibodies have been affinity matured from repeated exposure to gp41 over many years [104]. Clearly, access to the individual genes of the gp41 specific IgA antibodies provides much stronger evidence of their natural occurrence in the cervix, in contrast to previous studies which demonstrate only the presence of polyclonal IgA in vaginal secretions [104]. The isolation of IgA mAbs from the cervix of HEPS sex workers from other cohorts (i.e Nairobi) is required to confirm their role in vivo.

# **1.9 Monoclonal Antibodies**

#### **1.9.1 Antibody Structure**

Antibodies are composed of two distinct regions: a variable (V) region and a constant (C) region. The V region contributes to the antigen specificity of the antibody, while the C region, represented by the 5 different isotypes, IgM, IgG, IgA, IgE and IgD, dictates the effector functions of the antibody. The antibody variable

domain is composed of the V regions of the heavy (H) and light (L) chains which come together to form the three dimensional antigen binding site or the paratope [104,106]. Each V region contains three hypervariable loops termed complementary-determining regions CDR1, CDR2, and CDR3 that are largely responsible for antigen binding. The CDR regions are separated by the more conserved framework regions (FR1, FR2, FR3, and FR4). The mature heavy chain transcript is formed by somatic recombination and random assembly of three germline gene segments: variable (VH); diversity (DH); and joining (JH) [104,107,108]. The VH gene can be separated into seven families (VH1-7), each sharing over 80% nucleotide homologies in the framework regions [104,106]. The VH region encodes the CDR1 and CDR2, while the DH and JH both encode for the CDR3. Consequently, there is extraordinarily high diversity in both the composition and length of the H-chain CDR3 (as much as 10<sup>23</sup> possible sequences) [109]. There are two types of human light chains, Kappa ( $\kappa$ ) responsible for approximately 60% of the antibody repertoire, and Lambda ( $\lambda$ ), which is responsible for the remaining 40% [104]. Each light chain is composed of rearrangements of the germline VL and JL gene segments forming the CDR1, CDR2, and CDR3. The combination of the rearrangement of the VH, JH, and DH, along with the VL and JL segments, different combinations of heavy and light chains, and somatic mutations in the CDR and FR regions all contribute to the diversity of the human antibody repertoire [104].

#### **1.9.2 Antigen Selection and Affinity Maturation**

Affinity maturation refers to the process by which repeated antigenic challenge of B cell clones results in the production of an antibody with progressively higher antigenic affinity [139]. The recombination of un-mutated V,D and J genes determines the initial antigen specificity of the surface antibody. Antigen binding is then optimized via somatic mutations in the FRs and CDRs in an antigen driven process where higher affinity clones are positively selected and amplified [140, 176]. Mutations in the CDRs are more likely to alter the antigen-binding properties of the antibody due to their plasticity and structural location, whereas mutations in the FRs tend to be more structurally destructive [140]. Consequently, antibodies that have been subject to antigen-driven affinity maturation tend to have higher numbers of replacement (R) than silent (S) mutations in the CDRs, and a lower R:S mutation ratio in the FRs [141].

Chang and Casali (1994) were the first to develop a model to estimate the probability that a given VH or VL sequence had been antigen selected. This binomial model compared the R:S ratios of a given VH or VL sequence to the expected R:S value, in the absence of antigen selection, of the most homologous germline gene given random mutation [141]. Lossos et al (2000) later altered this approach by developing a multinomial model that can be applied to any VH or VL gene to estimate the probability that the gene has been subject to antigen-driven affinity maturation [142].
## **1.9.3 MAb Production Methods**

Early methods of producing antibody relied on collecting serum from animals that had been immunized with the antigen of interest [110]. Problems with this method, including the limited supply and polyclonal nature of the antibodies obtained (an undefined mixture of different Abs) were the driving force behind the development of hybridoma technology [110, 173]. This technology, developed in the late 1970's, allows for the immortalization of individual B cells from an immunized lab animal [173]. These cells can be grown *in vitro* to produce large quantities of individual monoclonal antibodies [173].

As useful as hybridoma technology has been, it has several inherent problems that prevent it from fulfilling the vast research and therapeutic potential of monoclonal antibodies. Perhaps the most glaring problem is that hybridoma technology results in the formation of non-human antibodies that, due to their inherent immunogenicity, cannot be administered directly to humans as a therapeutic. A second problem is that because antigen selection initially occurs in the serum of the lab animal there is little to control over the antibody response. Consequently, selection for sensitive molecules, or specific regions of antigens is very difficult [110]. Finally, because the DNA encoding the mAb is not provided, painstaking procedures must be undertaken to convert the antibody to a recombinant form to analyze the VH and VL genes, or to mutate the protein [110].

The development of recombinant antibody technology has largely solved the problems of hybridoma technology. There have been several methods developed for the isolation of fully human recombinant monoclonal antibodies that rely on steps

that mimic the *in vivo* selection process [111]. First, the genetic diversity of VH and VL regions is cloned by PCR from a population of immune or naïve human B cells. Second, the genotype of the antibody (DNA) is coupled to its phenotype (antigen specificity) using a system capable of expressing the library of re-arranged VH and VL genes as functional antibodies. Third, antibodies of interest are selected for their binding properties to a specific antigen. Finally, the amplification and large scale production of the desired antibodies [111, 112].

The predominant method for the expression and selection of recombinant antibodies is called phage display [113]. To a lesser extent, ribosome/mRNA display [114] and microbial display [115] are also used to produce mAbs. Using phage display, Barbas et al (1991) was the first to isolate mAbs against HIV-1 from human B cells by expressing antibody fragments on the surface of filamentous bacteriophages [113]. In that study, Barbas et al (1991) used random combinations of heavy and light chains to produce an antibody library, from wich the HIV neutralizing gp120 specific mAb b12 was isolated, that originated from bone marrow derived B cells of a HIV-1 infected individual [113]. It can be argued that using this approach may yield few antibodies of interest due to the infrequent pairing of the original heavy and light chain combinations. However, as Burton et al (1991) suggests, because the mRNA and not DNA is used to generate the library, the random combinatorial approach represents the current antibody response of the individual donor and original H and L pairings can be observed quite frequently [126]. Phage display is a robust and versatile method that has been used successfully for almost 20 years to produce the majority of human mAbs used today.

Despite the success of the phage, ribosome/mRNA, and microbial cell display systems, improvements in the antibody expression and selection processes can still be made. All of the recombinant techniques outlined above involve the expression of antibodies in a prokaryotic environment rather than the natural environment: the secretory pathway of mammalian cells [112]. This causes a bias in the antibodies produced, as antibody selection occurs not only for antigen specificity, but also for biophysical properties that promote superior expression in the foreign prokaryotic system [112]. Indeed Bowly et al (2007), has shown that a library dipslayed via a prokaryotic system (phage) yields a completely different set of mAbs when the same library is displayed in a eukaryotic system (yest) [174]. Furthermore, severe translational problems and difficulties with protein folding can occur when transferring antibodies selected via phage or microbial cell display to mammalian cell culture systems to express full antibodies [116].

# 1.9.4 MAb Production via Display on Mammalian Cells

In recent years, antibody selecting systems have been developed that exploit the secretory pathway of mammalian cells. Expressing and selecting antibodies displayed on the surface of mammalian cells ensures that the antibodies produced are less biased by characteristics other than antigen specificity [112]. Another advantage of using a mammalian cell display system is that, provided fresh PBMC samples can be obtained, fluorescence activated cell sorting (FACS) can be used to separate B cells for the presence of surface markers or antigen specificity prior to the cloning of V region diversity. This results in an antibody library that is cloned

from the repertoire of a small subset of desired B cells [117]. Researchers have used this method to clone antibodies from individually sorted B cells from whole blood against influenza antigens, vaccinating antigen, and self proteins [117, 118, 119]. Pre-sorting B cells circumvents the major problem of using mammalian cell display: the amount of cells that can be handled at one limits the size and complexity of the antibody library.

FACS has also been used in the selection of antibodies displayed on the surface of BHK and HEK 293T cells [112, 116]. Ho et al (2006) transiently transfected an expression vector with the human platelet derived growth factor receptor transmembrane domain fused to the C terminal portion of the VH region to promote antibody expression on the surface of 293T cells [116], while Beerli et al (2008) facilitated single chain V region BHK cell surface expression using a Sinbis virus expression system [112]. Both systems utilized FACS to separate a small percentage of antibody decorated cells for antigen affinity and isolated high affinity mAbs from the memory B cells of an immunized individual [112], or from a mutant antibody library generated from a single mAb [116]. Mammalian display systems have yet to be used to isolate antibodies from non-sorted cervical B cells.

# 1.10 Neutralizing Antibodies and HIV-1

## 1.10.1 Role of Therapeutic Antibodies

As discussed above, the study of broadly neutralizing monoclonal antibodies (bNt mAbs) against gp120 and gp41 has resulted in the discovery of several

conserved neutralization epitopes present on HIV-1 Env. In addition to providing invaluable structural data on HIV-1, bNt mAbs may also have potential as a therapeutic agent against both HIV-1 infection and AIDS progression. Antibodies have proven their usefulness *in vivo* against viral infections as they have been used as treatments and preventative therapies in humans against hepatitis B virus, cytomegalovirus, varicella zoster virus, rabies virus, and syncytial virus [120-123]. Indeed, over 20 mAbs currently have FDA approval for use in humans as therapeutic agents [120]. Antibody therapy has numerous advantages over traditional smallmolecule drugs. These include: a high specificity and affinity for their target; little to no metabolic side effects due to their catabolism to amino acids; a reduced toxicity due to their inability to diffuse across the cell membrane; little to no antigenicity of fully human antibodies; and their long serum half lives allow for infrequent dosing [120]. While therapeutic antibodies currently need to be injected, their infrequent dose regimens could be a viable alternative to the daily pill regimens of many therapies including anti-retroviral therapy (ART) [120].

Whether bNt mAbs can be used successfully against HIV-1 has been the subject of much debate. While passive transfer experiments in animal models has confirmed that bNt mAbs can prevent infection and limit transmission *in vivo* [50, 89, 90, 124], clinical testing in humans has so far been limited and only marginally successful [120]. In order for the development of antibody therapeutics against HIV-1 to move forward, novel mAbs against HIV-1 Env need to be isolated. The discovery of novel Env specific mAbs is required for the identification of: new susceptible target epitopes; novel epitopes conserved numerous HIV-1 subtypes; the effector

functions that provide protection from HIV-1 *in vivo*; and novel combinations of different antibodies capable of provide maximum protection from HIV-1 [120].

## 1.10.2 MAbs versus gp120

Until recently, the only gp120 specific mAb to show broad neutralizing ability of diverse primary isolates of HIV-1 has been mAb b12. This mAb was selected from an antibody library derived from the bone marrow of an HIV-1 subytype B infected individual that had been asymptomatic for 6 years [125]. This random combinatorial library was displayed on the surface of an M13 bacteriophage and selected for its affinity to gp120 IIIB [125]. Burton et al (1994) converted this fab into a full IgG1 molecule and demonstrated a neutralizing ability of primary virus isolates of more than 75% [126]. Zhou et al (2007) later resolved the crystal structure of b12 with gp120 and confirmed that b12 bound to the conformational dependent CD4 binding site wedged between the non-immunogenic glycan shield and the flexible inner domain [127]. Phage display has yielded numerous other mAbs vs gp120 from subtype B infected individuals that have neutralizing ability [128-131], none however, demonstrate the broad neutralizing ability of b12.

Walker et al (2009) has recently developed the first broadly neutralizing HIV-1 mAbs directed against gp120 since b12 [145]. In this study, memory B cells from an African donor with potent serum neutralizing activity were used to generate two mAbs with remarkable neutralization breadth and potency [145]. The two somatic variants, PG9 and PG16, neutralized 127 and 119 out of 162 pseudoviruses respectively in a large multiclade panel [145]. Both mAbs

demonstrated binding to an epitope along the V2 and V3 domain of gp120 only when gp120 was expressed as part of the functional Env spike [145].

#### 1.10.3 MAbs versus gp41

The relatively conserved nature of gp41 suggests it could be a target for bNt mAbs. The major problem with targeting this protein is that in the native spike, gp41 is largely hidden from antibodies by the gp120 trimer [61]. In a comprehensive study of 6 HIV infected individuals, Scheid et al (2009) cloned a total of 433 antibodies from individual B cells sorted for affinity to gp140 [132]. Using ELISA experiments, they determined that 70% of the antibodies bound gp120, and only 30% bound gp41 [132]. Of the 130 gp41 mAbs, zero bound to the MPER containing the 2F5 and 4E10 epitopes, and only 9 bound to the previously reported immunodominant region of gp41 cluster 1 (aa 579-604) [132, 133]. The lack of reactivity against gp41 peptides suggests that the majority of gp41 antibodies from HIV-1 infected individuals recognize conformational dependent epitopes present in the native protein [132].

Binley et al (1996) constructed phage display combinatorial libraries from six HIV-1 positive donors and isolated 23 different mAbs against conformational epitopes on gp41 [134]. All of the antibodies could be mapped to one of three regions on gp41: cluster 1, cluster II (aa 649-668), and cluster III (aa 619-648) [134]. The fact that none of these antibodies showed neutralizing ability against the virus led the authors to conclude that these epitopes are obscured by gp120 on the native spike [134]. These antibodies were likely generated against precursor

envelope proteins, viral debris, or spikes in which the gp120 cap had been stripped away [134].

Many random combinatorial phage display libraries from HIV positive donors have been selected for binding to chimeric gp41 constructs that mimic the various stages of gp41 during the fusion process. This strategy; however, has yielded only a few mAbs with significant cross-clade neutralizing ability. Gustchina et al (2007) screened a combinatorial library for binding to a gp41 construct that displays the N-HR as an  $\alpha$ -helix extending from the six-helix bundle [135]. From that library, one Fab was selected (3674) that bound a shallow groove located between two C-HR helices in the six-helix bundle and neutralized primary isolates of clades A, B and C [135]. Two other mAbs showing moderate neutralizing activity, D5 and m46, have been selected for binding to the six-helix bundle, N-HR and C-HR peptides [136, 137].

By far the most intensely studied mAbs against gp41 are 2F5, 4E10, and to a lesser extent, Z13. MAb 2F5 was produced using traditional fusion technology by fusing the PBMCs of a HIV-1 asymptomatic seropositive donor with heteromyeloma cells and screening against gp41 [41]. This antibody and has been shown to recognize a conserved linear epitope (72% of variable isolates) in the MPER region (aa 662-668, ELDKWAS) of gp41 and neutralize a variety of clinical and lab adapted strains of HIV [41]. The broadly neutralizing antibody 4E10 was produced in much the same way as mAb 2F5 and binds an adjacent linear epitope in the MPER (aa 671-681) NWFNITNWLW) [61]. mAb Z13 was selected from a random combinatorial phage display library from an asymptomatic seropositive donor and binds to a

linear epitope in between the 4E10 and 2F5 epitopes on the MPER (aa 669-679, SLWNWFDITN) [61, 138]. To determine how these three bNt mAbs mediate virus neutralization, Song et al (2009) used electron paramagnetic resonance (EPR) and nuclear magnetic resonance (NMR) to examine the conformational changes in the MPER following antibody binding [138]. It was discovered that all three mAbs inhibit MPER related functions in the virus-cell fusion process via different mechanisms [138]. 2F5 binding to gp41 caused a drastic conformational shift in the N terminal hinge region of the MPER exposing the amino acid residues 669 and 670 [138]. 4E10 binding also mediated a conformational shift in the MPER exposing amino acids in the center of the hinge region, 672 and 673 [138]. MAb Z13 does not initiate a conformational shift, but rather rigidifies the MPER preventing the structural rearrangement needed during the fusion process [138].

# **<u>1.11 Summary/ Statement of Objectives</u>**

With the exception of the mAb recently generated by Tudor et al (2009), all of the bNt mAbs against HIV-1have been isolated from peripheral blood or bone marrow derived B cells from HIV-1 seropositive individuals, predominantly infected with clade B virus. To develop antibodies with potency against non-clade B viruses, donors from different geographical regions must be used. The development of the African derived gp120 bNT mAbS PG9 and PG16 validates this approach. To ultimately protect against HIV-1 infection, the virus must be neutralized at the site of initial contact; however, the antibodies that play the most important role at this

site in women, cervicovaginal IgA, has been largely ignored. Furthermore, the majority of the mAbs herein discussed have been developed using comparatively old technology with inherent limitations.

The role played by IgA in resistance to HIV-1 infection is highly controversial. Traditional techniques to measure HIV-1 specific cervical IgA in HEPS individuals have been wildly inconsistent. The isolation of gp41-specific neutralizing monoclonal IgAs from a HEPS cohort in Cambodia has confirmed the presence of HIV-1 neutralizing IgA in the cervix of women from this cohort. In order to elucidate the role played by cervical IgA in the Nairobi cohort, in-depth studies of HIV-1 specific monoclonal IgAs must be undertaken.

This project was initiated to develop novel HIV-1- specific IgA mAbs from the cervix of Kenyan HEPS individuals using methods that avoid the biases and limitations inherent in other antibody production methods. We hypothesize that using a recombinant approach, coupled with a mammalian cell tissue culture system, we will be able to retrieve gp41 specific antibodies from the IgA repertoire of cervical mononuclear cells (CMCs) obtained from Kenyan HEPS commercial sex workers. This work describes the development a mammalian cell display antibody expression and screening system and the role it played in the isolation and characterization of 4 novel gp41 specific monoclonal antibodies derived from the cervical B cell IgA repertoire of two HEPS commercial sex workers in Nairobi, Kenya.

# 2 Materials and Methods

## 2.1 Antibody Variable (V) Region Gene Recovery From PBMC and CMC Samples

# 2.1.1 Study Population

A cohort of commercial sex workers was established through a clinic in the Pumwani area of Nairobi Kenya [73]. Even though the women are encouraged to used condoms and provided with behavioral counseling, it is estimated they have at least 64 unprotected sexual exposures to the HIV-1 virus every year [73]. Initially seronegative women who remain HIV-1 seronegative and PCR-negative for at least 3 years while continuing sex work are classified as HIV-1 resistant [73].

# 2.1.2 Lymphocyte Collection

For the construction of the cervical mononuclear cell (CMC) antibody library, CMCs from two female commercial sex workers in Nairobi, Kenya were collected. ML 1515 and ML 1356 were selected as lymphocyte donors based on their status as being HIV-1 seronegative despite being highly exposed to the HIV-1 virus through commercial sex. ML 1356 and ML 1515 have continued to test HIV-1 negative since testing began in 1991, and 1992 respectively. CMCs were harvested from the cervix in January 2007 as part of a routine survey using a clinical cytobrush. Samples consisted of on average 1000 to 100 000 lymphocytes (median 10 000).

As proof of principle, a population of B cells known to contain clones specific for HIV-1 was harvested. For construction of this separate antibody library,

peripheral blood mononuclear cells (PBMCs) were harvested from an HIV-1 infected sex worker by Ficoll-Hypaque gradient centrifugation. ML 256, HIV-1 positive since 1985 was selected as a lymphocyte donor based on the high serum titer to both gp120 and gp41.

All samples were washed and re-suspended in buffer RLT (Qiagen) containing 2-mercaptoethanol and stored in liquid Nitrogen. Samples were shipped to Winnipeg using dry ice and subsequently stored at -80°C until ready for use.

# 2.1.3 RNA Isolation

Total RNA was recovered from each CMC and PBMC cell population using the RNeasy Mini Kit (Qiagen) according to manufacturer's instructions. RNA was quantified using a ND-1000 Spectrophotometer (NanoDrop®). Typical yields of RNA were 500 to 700 ng for CMC samples and 1500 to 2000 ng for PBMC samples.

# 2.1.4 cDNA synthesis

cDNA was synthesized from each RNA sample using the Thermoscript RT-PCR system (Invitrogen) with an oligo dT primer according to manufacturer's instructions.

# 2.1.5 PCR Strategy and Primer Design

Due to the prevalence and importance of IgA in the cervical mucosa, immunoglobulin variable heavy (VH) of the IgA isotype ( $\alpha$ ) were amplified from the CMC cDNA, while VH regions of the IgG isotype ( $\gamma$ ) were amplified from the PBMC

cell population. Immunoglobulin variable light chain genes (VL) Kappa ( $\kappa$ ) and Lambda ( $\lambda$ ) were amplified from both the CMC and PBMC samples. All PCR reactions were carried out using Expand High Fidelity PCR System (Roche) using 2  $\mu$ l of cDNA as a template. Immunoglobulin light and heavy chain variable regions encoding each V-domain pool were amplified using panels of sense oligonucleotide primers that hybridize with cDNA in the framework 1 region (Fr-1), (corresponding to the Nterminal amino acid region), of the different antibody families. These were paired with isotype specific anti-sense primers that hybridize with the N terminal amino acid region of constant region 1 (CH1) of either: IgG1-3 (in the case of PBMC cDNA); IgA1,2 ( in the case of CMC cDNA as template); IgKappa, or IgLambda. All primers were synthesized by the DNA core Facility (NML). Primers were designed based largely upon sequences available to the public and as in Berry et al (2003) [105].

# 2.1.6 PCR amplification of IgA and IgG VH regions

Immunoglobulin VH regions 1-6 were independently amplified for both IgA subtypes 1 and 2 from the cDNA CMC template, as well as the three IgG subtypes 1,2a, and 3 from the PBMC cDNA template. Framework 1 gene specific sense primers Fr-1 VH1,3, Fr-1 VH2, Fr-1 VH4, Fr-1 VH4b, Fr-1 VH5, and Fr-1 VH6 were paired with isotype specific antisense primers IgG1 Reverse, IgG2 Reverse, IgG3 Reverse, IgA1,2 Reverse, and IgA1 Reverse (see appendix 1 for all primer sequences).

All PCR reactions were performed in thin walled PCR tubes (FisherBrand) in a total volume of 50  $\mu$ l containing 600 nM sense and anti-sense primers, 400  $\mu$ M

each dNTP (Roche), 37μl nuclease free water (Promega), 5 μl 10X Expand High Fidelity Buffer with 15 mM MgCl<sub>2</sub> (Roche), and 3.5 units of Expand High Fidelity Enzyme Mix (Roche). Thermal cycling was performed on a GeneAmp PCR system 9700 Thermocycler (Applied Biosystems) consisting of 1 cycle at 94°C for 5 min, followed by 10 cycles of 94°C for 30 sec, 55°C for 30 sec and 72°C for 45 sec; then by 25 cycles of 94°C for 30 sec, 55°C for 30 sec, and 72°C for 45 sec\* (15 sec increase in incubation time with every cycle), and 1 cycle of 72°C for 7 min.

# 2.1.7 PCR Amplification of Kappa and Lambda VL regions

Amplification of Kappa VL domains 1-5 was done individually using Fr-1 specific sense primers IgK-V1, IgK-V2,4, IgK-V3, and IgK-V5, paired with the corresponding antisense primer specific for the C terminal amino acid region of Fr-4, VJKrev1, VJKrev2,4, VJKrev3, or VJKrev5.

Similarly, amplification of Lambda VL domains 1-10 was done using individual Fr-1 specific sense primers IgLam-V1a, IgLam-V1b, IgLam-V2, IgLam-V3, IgLam-V4, IgLam-V6, IgLam-V7,8, IgLam-V9, and IgLam-V10. The Fr-1 specific sense primers were each paired with a pool of 4 Lambda Fr-4 specific anti-sense primers (200 nM each): VJLamRev1236, VJLamRev4, and VJLamRev5,7. PCR was amplification was done as per section 2.1.6.

# 2.1.8 Analysis of PCR

A commercial 6X blue/orange loading dye (Promega) was diluted 1:6 in 20  $\mu$ l of PCR sample and each dispensed into a 100 ml 1% agarose gel in 1X TAE buffer

(Bio-Rad) containing 40  $\mu$ g ethidium bromide (Invitrogen). 2  $\mu$ l of the blue/orange loading dye was added to 10  $\mu$ l of Low Mass DNA Ladder (Invitrogen), and subsequently dispensed into the first well. Samples were run from negative to positive for 1 hour at 100 V using a Bio-Rad PowerPac HC. The gel was visualized under UV light using a Gel Doc EQ (Bio-Rad) and analyzed using Quantity One 1D analysis software 4.6 (Bio-Rad).

# 2.1.9 PCR fragment Isolation

Immunoglobulin VH and VL DNA of approximately 400 to 700 base pairs were excised from the agarose gel using a sterile scalpel (Feather) and purified using a Qiaquick Gel Extraction Kit (Qiagen) according to manufacturer's instructions.

# 2.1.10 Extension PCR of VH- $\gamma$ , $\alpha$ and VL- $\kappa$ , $\lambda$ .

A second round of PCR was carried out to add restriction endonuclease sites to the 5' and 3' ends of each VH and VL family to facilitate cloning into expression vectors using primers complementary to the termini of the cDNA containing embedded restriction sites. An EcoRV site was added to the 5' end of both the VH and VL regions, while an Apa1 site was added to the 3' end of the VH regions and an NcoI site was added to 3' ends of the VL regions.

As template, 10 to 20 ng of each VH region family was amplified using one of the following sense VH extension primers of appropriate familial specificity: VH1236EXT; VH4bEXT; or VH5ext. Each VH family specific sense primer was paired

with either IgA1,2bpEXT, or IgGbpEXT depending on the source and isotype of the V region being amplified. An EcoRV recognition sequence (GATATC) was encoded approximately 15 bases from the 5' end of each sense oligonucleotide extension primer, while an ApaI recognition sequence (GGGCCC) was encoded approximately 15 bases from the 3' end of each anti-sense oligonucleotide extension primer.

Ten to twenty micrograms of Kappa V regions families 1-5 were amplified using KappaFext sense oligonucleotide primer with an EcoRV site approximately 15 bases from the 5' end, paired with one of the following anti-sense oligonucleotide primers of appropriate familial specificity: VJKRev-1X; VJkRev-2,4X; VJKRev-3X; or VJKRev-5X. Each antisense primer had an NcoI recognition sequence (CCATGG) approximately 10 bases from the 5' end. Lambda V region families 1-10 were amplified using a pool of two sense oligonucleotide primers (LambdaFext1, and LambdaFext2, 300 nM each) with an EcoRV recognition sequence 15 bases from the 5' end. The pool of Lambda sense extension primers was paired with a pool of 3 Lambda anti-sense extension primers (VJLRev1236X, VJLRev4X and VJLRev5,7X, 200 nM each) with an NcoI recognition sequence 10 bases from the 5' end of each primer.

All PCR reactions were performed in thin walled PCR tubes (FisherBrand) in a total volume of 50 µl per well containing 600 nM sense and anti-sense primers, 400 µM each dNTP (Roche), 37µl nuclease free water (Promega), 5 µl 10X Expand High Fidelity Buffer with 15 mM MgCl<sub>2</sub> (Roche), and 3.5 units of Expand High Fidelity Enzyme Mix (Roche). Thermal cycling was performed on a GeneAmp PCR system 9700 Thermocycler (Applied Biosystems) consisting of 1 cycle at 94°C for 5

min, 20 cycles of 94°C for 30 sec, 55°C for 1 min and 72°C for 45 sec, and 1 cycle at 72°C for 7 min. PCR fragments were visualized, extracted, purified and quantified as per sections 2.1.8 – 2.1.9.

# 2.2 Expression Vector Manipulation

## 2.2.1 P-dbIgG1 Mutagenesis

To facilitate site specific ApaI mediated cloning into the P-dbIgG1 expression vector (Figure 5, Appendix II), a second ApaI endonuclease recognition sequence (GGGCCC) was removed from the zeocin resistance gene of the vector using a QuickChange Site-Directed Mutagenesis Kit (Stratagene) according to manufacturer's instructions. Due to the location of the recognition sequence, three different mutagenesis reactions were performed using the following primer sets, each resulting in three different single base mutations: pIgGmut1F and pIgGmutR (resulting in a GCGCCC); pIgGmut2F, and pIgGmut2R (resulting in GGGCGC); and pIgGmut3F and pIgGmut3R (resulting in GGGCCG).

Next, 2  $\mu$ l of each P-dbIgG1 mutation were added to chemically competent *E. coli* OneShot ® Top10 cells (Invitrogen) and incubated for 20 min on ice. Cells were heat shocked for 30 sec at 42°C, and incubated for 1 hr at 37°C in 250  $\mu$ l S.O.C. media (Invitrogen) with shaking at 220 rpm. 40  $\mu$ l of each transformation was plated on an LB agar plate with 25  $\mu$ g/ml zeocin (Invitrogen). The mutation with the least effect on the zeocin resistance gene of the plasmid corresponded with the plate showing the highest level of growth. Next, 10 colonies from mutation number 2

were transferred to 3 ml of LB-Lennox media containing 25 μg/ml of zeocin and cultured overnight at 37°C with shaking at 250 rpm. Plasmids were purified using a Qiagen Plasmid Miniprep Kit according to manufacturer's instructions.

# 2.2.2 P-dbLambda Generation

To ensure optimal expression of VL-Lambda genes, the cDNA encoding the human CH1 Lambda region was used to replace the CH1 Kappa region in the expression vector P-dbKappa (Figure 5). To obtain the Lambda CH1 region, PCR was done using the Expand High Fidelity PCR system as per section 2.1.6 with 2 µl of cDNA from the PBMC sample of ML 256 as the template along with LamConF, and LamConR as the sense and anti-sense primers respectively. A PCR fragment of approximately 350 bp was excised from a 1% agarose gel and purified as per sections 2.1.8-2.1.9.

To permit TA cloning, adenine nucleotides were added to the 5' and 3' ends of the PCR fragment. 1  $\mu$ l of 100 mM dATP (Invitrogen) was added to 100 ng of Lambda-CH1 in a 14  $\mu$ l reaction volume containing 1 U of *Taq* DNA polymerase (Invitrogen), and 2  $\mu$ l of 10 X PCR Buffer (Invitrogen). The reaction was incubated for 30 min at 72°C.

For insertion into PCR@2.1 cloning vector (Invitrogen), 4 µl of the A – tailed PCR product was added to 1 µl of PCR@2.1 vector and 1 µl of salt solution (1.2 M NaCl, 0.6M MgCl<sub>2</sub>) (Invitrogen), and incubated for 5 min at room temperature. 2 µl was then transformed into chemically competent OneShot @ TOP10 *E.coli* cells (Invitrogen) as per section 2.2.1. Cells were plated on LB-agar plates containing 100

μg/ml ampicillin, previously coated with 40 μl of 40 mg/ml X-gal in dimethylformamide (Invitrogen), and incubated overnight at 37°C. 10 white colonies were selected, expanded in 3 ml of LB broth containing 100 μg/ml ampicillin overnight, and processed in a similar fashion as 2.2.1. The ND-1000 Spectrophotometer was used to quantify the DNA and 10 μl of each sample (150 ng/μl) was sent to the Genomics Core Facility within the NML for sequencing along with M13 Reverse (5'CAGGAAACAGCTATGC 3') and T7

(5'TAATACGACTCACTATAGGG 3') oligonucleotide primers (Invitrogen). Data was analyzed using DNAStar software.

Extension PCR was done using a single PCR®2.1 vector clone containing the lambda CH1 insert as a template with the Expand High Fidelity PCR System (Roche) as per section 2.1.10. Oligonucleotide primers LamConFext, and LamConRext were used to add an NcoI recognition sequence on the 5' end and an NheI recognition sequence on the 3' end of Lambda CH1. The human Lambda CH1 extension PCR product was inserted in the place of the human Kappa CH1 in P-dbKappa following a similar protocol as section and 2.4.

# 2.3 Restriction Endonuclease Digests

# 2.3.1 Restriction Endonuclease Digest of VH- α,γ and P-dbIgG1

All VH extension PCR products purified per section 2.1.10 were double digested with EcoRV, and ApaI (New England BioLabs) for cloning into P-dbIgG1 expression vector (Figure 5). Each digest was done in a total volume of 40 µl

containing approximately 500 ng of each VH family, 4 μl of 10X Buffer 4 (New England BioLabs), 4 μl of 10 mg/ml BSA (New England BioLabs), and 50 Units of ApaI restriction enzyme. Each reaction was incubated for 2 hr at 25°C, after which, 40 Units of EcoRV restriction enzyme was added and incubated for an additional 2 hours at 37°C. Next, 5 μg of P-dbIgG1 was digested in a 40 μl total reaction volume under the same conditions. Following digestion of the vectors, 1U of Calf Intestinal Alkaline Phosphatase (Invitrogen) was added to the P-dbIgG1 digestion and incubated for 20 min at 50°C to reduce re-ligation of the vector. Next, 5 μl of blue/orange loading dye was added to each digest and loaded onto a 2% agarose gel as per section 2.1.8. Samples were run for 1.5 hr at 100 V and VH bands of approximately 400 basepairs, and linearized P-dbIgG1 of approximately 4500 basepairs, were excised and purified as per sections 2.1.8-2.1.9.

## 2.3.2 Restriction Endonuclease Digest of VL-κ,λ, P-dbKappa, and P-dbLambda

All VL- $\kappa$ , $\lambda$  extension PCR products purified per section 2.1.10 were double digested with EcoRV, and NcoI (New England BioLabs) for cloning into P-dbKappa and P-dbLambda expression vectors. Each digest was done in a total volume of 40 µl containing approximately 500 ng of each VL family, 4 µl of 10X Buffer 3 (New England BioLabs), 4 µl of 10 mg/ml BSA, 40 Units of EcoRV and 50 Units of NcoI. Each reaction was incubated for 4 hr at 37°C. 5 µg of P-dbKappa and 5 µg of PdbLambda were digested in a 40 µl total reaction volume under the same conditions. Following digestion of the vectors, 1 U of Calf Intestinal Alkaline

Phosphatase was added and incubated for 20 min at 50°C. Samples were subsequently processed in a similar fashion as 2.3.2.

# 2.4 Cloning B12 Fab from pComb3X into P-dbIgG1 and P-dbKappa

The phagemid cloning vector pComb3X containing the VH and VL sequence of the gp120 specific antibody B12 was generously donated by Dr. Carlos Barbas III (Scripps Research Institute). The VH and VL regions were amplified by extension PCR as per section 2.1.10 using 10 ng pComb3X as the template with either B12VHFext and IgGbpExt (VH) or 4LFext and 4LRext (VL) as sense and anti-sense primers. Amplified VH and VL regions were purified, digested, and purified a second time as outlined in sections 2.1.7, 2.1.8, and 2.3. Next, 30 ng of digested B12 VH and VL were ligated with linearized Pdb-IgG1 and P-dbKappa in a 10 µl reaction volume containing 2 µl of 5X T4 DNA Ligase Buffer (Invitrogen), and 1 unit of T4 DNA ligase (Invitrogen). Ligations were incubated for 16 hr at 14°C.

Both P-dbIgG1B12, and P-dbKappaB12 ligations were transformed into chemically competent TOP10® *E.coli* cells (Invitrogen) as per section 2.2.1. Cells were plated on LB-amp plates (100 µg/ml) and 10 colonies were selected, expanded, and plasmids isolated following a similar protocol as section 2.2.1. The resulting clones were sequenced using flanking primers B12VHFext and IgG1reverse for P-dbIgG1B12; or 4LFext and 4LRext for P-dbKappaB12 to confirm the presence and orientation of each insert. Clones with the correct insert were expanded using a Qiagen Maxi Prep Kit according to manufacturer's instructions.

## 2.5 Antibody Secretion System: Generation of ML 256 PBMC IgG Library

# 2.5.1 Ligation of VH-γ with P-dbIgG1

VH- $\gamma$  regions digested per section 2.3.1 were pooled and ligated into PdblgG1 creating a plasmid library encoding a population of VH genes originating from the PBMCs of the HIV-1 positive sex worker ML 256. The 14 cDNA VH-γ regions that were amplified by two rounds of PCR and subsequently digested by restriction endonucleases, were pooled into seven groups, each consisting of 3.5-4 µg of total DNA. Next, 200 ng of DNA from each pool was ligated with 200 ng of linearized PdbIgG1 in a 15 µl total volume containing 3 µl 5X T4 DNA Ligase Buffer and 1 Unit of T4 DNA Ligase. Each ligation was incubated separately for 24 hr at 14°C then pooled forming one total ligation. Plasmid DNA was precipitated from the ligation by adding 2.2 volumes 96-100% ethanol (Commercial Alcohols), 0.1 volumes 3M sodium acetate pH 5.5 (Ambion), and 40 µg of glycogen (Roche). Following a 2 hr incubation at -80°C, the ligation was centrifuged for 15 min at 13000 g (4°C) and supernatant discarded. The pellet was washed twice by re-suspension in 500 µl ice cold 70% ethanol followed by gentle agitation and prompt centrifugation for 5 min at 13000 g (4°C). The pellet was air dried for 5 min and re-suspended in 10 µl of nuclease free water (Promega).

# 2.5.2 Ligation of VL-κ,λ into P-dbKappa and P-dbLambda

The cDNA amplified from four VL- $\kappa$  families from PBMC sample of ML 256 digested per section 2.3.2 were pooled into 2 groups, while the 9 VL- $\lambda$  families were pooled into 4 groups. 200 ng of DNA from each group were ligated with 200 ng of either P-dbKappa or P-dbLambda and precipitated as per section 2.5.1.

## 2.5.3 Plasmid Library Amplification

For amplification, 2 µl from the ML 256 V region ligations into Pdb-IgG1, PdbKappa, and Pdb-Lambda, were each added to 2 vials (6 vials total) of One Shot Electrocompetent TOP10® *E.coli* cells (Invitrogen). Cells were transformed according to manufacturer's instructions with an *E. coli* Gene Pulser® Cuvette (Bio-Rad). Following incubation for 1 hr at 37°C with shaking at 220 rpm in S.O.C media (Invitrogen), 1 µl from each library was plated using a cell spreader on LB plates containing 25 µg/ml of zeocin to assess the diversity of the library. The remaining 250 µl of each transformation was added to 110 ml LB-Lennox broth with 25 µg/ml zeocin in 400 ml Erlenmeyer flasks. Cells were incubated for 18 hr at 37°C with shaking at 250 rpm. Plasmids were isolated with a MaxiPrep Kit (Qiagen) according to manufacturer's instructions.

# 2.5.4 Determining the Diversity of the ML 256 VH Library

To examine diversity, 15 clones from heavy chain ligation plated in section 2.5.3 were selected and expanded as per section 2.2.1. Plasmids were sequenced using the following set of flanking oligonucleotide sense and anti-sense primers:

SeqF1 and IgG1reverse. Sequence data was analyzed using both DNAstar, and MacVector software.

A restriction digest profile was done on 15 VH clones by digesting 100  $\mu$ g of each VH region in a total volume of 40  $\mu$ l with 1 U of AluI restriction enzyme(New England BioLabs) and 4  $\mu$ l of buffer 4 (New England BioLabs) for 2 hr at 37°C. Digest patterns were visualized on a 1% agarose gel as per section 2.1.8.

# 2.5.5 Growth and Maintenance of HEK 293F Cells

HEK 293F cells (Invitrogen) were grown in sterile, polycarbonate Erlenmeyer Flasks (VWR) using FreeStyle<sup>TM</sup> 293 Expression Medium (Gibco) to a density of 2-3 x 10<sup>6</sup> cells/ml at 37°C with 8% CO<sub>2</sub> and shaking at 135 rpm. Cells were passaged every 3 to 4 days to 3 x 10<sup>5</sup> cells/ml. Early passages were frozen in FreeStyle<sup>TM</sup> 293 Expression Medium with 10% DMSO (Sigma) and stored at -150°C.

#### 2.5.6 Transfection of 293F cells with the ML256 PBMC VH and VL Library

The day of transfection, cells were diluted to 1.1 x 10<sup>6</sup> cells/ml in 28 ml of FreeStyle<sup>™</sup> 293 Expression Medium. 80 µl of 293fectin<sup>™</sup> (Invitrogen) was diluted in 1 ml of Opti-Mem® media (Invitrogen) and incubated at room temperature for 5 min. 20 µg of the VH P-dbIgG1 library, and 20 ug of VL P-dbKappa/Lambda library diluted in 1ml of Opti-Mem® media was added to the activated 293fectin<sup>™</sup> – Opti-Mem® media mixture and incubated for 25 min at room temperature. Immediately following the incubation, The DNA-293fectin<sup>™</sup> complexes were added to the previously prepared 293F cells. As a positive control for both expression and

affinity, P-dbIgG1B12 and P-dbKappaB12 were co-transfected into HEK 293F cells following the same protocol.

## 2.5.7 Re-suspension of the Transfectants into Semi-Solid Media

The day prior to re-suspension, CloneMedia-HEK (Genetix) was allowed to thaw at 4°C. The day of re-suspension, 10 ml of sterile H<sub>2</sub>O containing zeocin was added to the media creating a final zeocin concentration of 50  $\mu$ g/ml. 48 hours post transfection, 50 000 cell aliquots of cell suspension were added to 2 ml of the prepared, semi-solid, CloneMedia-HEK that had been aliquoted into 5 ml Falcon tubes. Following thorough mixing, the suspensions were plated in 6 well tissue culture plates (Genetix) and incubated in humidified air at 37°C with 8% CO<sub>2</sub>.

After incubating for 14 to 17 days, colonies could be visually identified and transferred to 96 well tissue culture plates (Corning) containing 200 µl of FreeStyle<sup>™</sup> 293 Expression Medium with 50 ug/ml of zeocin. Cells were incubated for another 14 days prior to screening. Individual colonies were considered "clonal" for purposes of experimentation.

# 2.5.8 Screening Clones for Expression and Reactivity Against gp41 and gp120 Using ELISA

Fourteen to seventeen days following colony picking, culture supernatants from each clone were tested for antibody expression against 100 ng of goat antihuman IgG F(ab')<sub>2</sub> fragment specific (Southern Biotech). Antibody specificity was tested in ELISA against 100 ng of recombinant gp120-gp41 mosaic antigen

(Virogen). This antigen consisted of the immunodominant regions of both gp120 and gp41 Env proteins. Antigen was coated overnight at  $4^{\circ}$ C in 60  $\mu$ l of Phosphate Buffered Saline (PBS) pH 7.2 onto individual wells of a MaxiSorp 96 well plate (NUNC<sup>™</sup>). Plates were blocked at 37°C for 2 hr with 0.2% Bovine Serum Albumin (BSA) (Sigma), 0.2% Skim Milk (Promega) in PBS. Plates were washed twice with MilliQ® water (resistivity  $18.2M\Omega \bullet cm$ ) using a Power Washer 384 (Tecan) automatic washing device. Next, 60 µl of culture supernatant was added and incubated at 37°C for 2 hr, then washed 8X with MilliQ® water as stated previously. Supernatant from cells transfected with B12IgG1/ $\kappa$  was included as a positive control. Next, 60 μl of a 1:2000 dilution of goat anti-human IgG Fc fragment specific polyclonal antibody conjugated to horseradish peroxidase (Southern Biotech) in PBS was added and incubated for 1 hr at 37°C. After 8 washes with water, 60 µl of 2,2'-azino-bis(3-ethylbenzthiazoline-6-sulphonic acid) (ABTS) substrate (Roche) was added and incubated for 45 min at room temperature. Colour development was measured at 405 nm by use of a SpectroMax250 Spectrometer (Molecular Devices) and analyzed using SoftMax Pro 4.6 Software (Molecular Devices).

# 2.5.9 V Gene Recovery from Positive Mammalian Cells

To identify the optimum method of V gene recovery from positive wells of transiently transfected cells, VH and VL genes were isolated using two techniques from two different clones (2G7 and 5B5) that had absorbance values against the gp41-gp120 mosaic antigen of 0.1 or greater in the ELISA. The first strategy was to isolate the RNA from the clone and use RT-PCR to amplify the V regions, while the

second strategy was to amplify the V regions directly from the plasmid DNA of the clone.

To test the first strategy, total RNA was harvested from clone 5B5 using an RNEasy Mini Kit (Qiagen) as per section 2.1.3, clonal mRNA was converted into cDNA using Thermoscript RT-PCR system (Invitrogen) as per section 2.1.4 and the VH and VL regions amplified using the Expand High Fidelity PCR system as per section 2.1.6 using with the following primer sets:

IL2ssVrecovF (sense), GcVrecovR (antisense)

IL2ssVrecovF (sense), KcVrecovR (antisense)

IL2ssVrecovF (sense), LcVrecovR (antisense)

To test the second strategy, clone 2G7 was transferred to 50  $\mu$ l H<sub>2</sub>O, incubated for 10 min at 99.9 °C in a thermocycler, vortexed vigorously, and centrifuged for 15 sec at 10 000 g. 10  $\mu$ L of supernatant was used directly as template in the PCR reaction outlined above.

Amplified V genes were re-cloned into expression vectors P-dbIgG1, PdbKappa or P-dbLambda following a similar protocol outlined previously. Plasmids were transformed into Top10 ® *E. coli* cells and 24 colonies from both 5B5 and 2G7 comprising the heavy chain (gamma), light chain (kappa) and light chain (lambda) were selected, expanded, and sequenced as per section 2.5.4.

# 2.5.10 Screening via Chain Shuffling

One productive VH immunoglobulin sequence was recovered from clone 2G7, while 2 productive VH and 2 productive VL-lambda sequences were recovered

from clone 5B5. Clearly, 5B5 was not clonal. Consequently, to deduce the binding of mAb 5B5, a total of 9 transfections were carried out pairing the 3 productive VH sequences in P-dbIgG1 with each of the 2 VL sequences in P-dbLambda. Transfections were carried out using 293fectin<sup>™</sup> in 6 well tissue culture plates (Corning) in a 6 ml total volume following a scaled down version of the protocol outlined in section 2.5.6. Supernatants were screened via ELISA against 100 ng of Goat anti-Human IgG F(ab')<sub>2</sub> fragment specific, 100 ng of recombinant, monomericgp120 (Immunodiagnostics) and 100 ng of recombinant, monomeric gp41 (Virogen) following a similar protocol outlined in section 2.5.9.

# 2.6 Cellular Display System: the Generation of a HIV-1 Resistant Cervical IgA Library

Due to several inefficiencies observed in the antibody secretion method outlined above, a cellular display system was developed to create the IgA CMC library from the HEPS sex workers ML 1356 and ML 1515 (Figure 3).

## 2.6.1 Creation of P-dblgG1-TM

To allow for antibody expression on the cell surface, the human platelet derived growth factor receptor (PDGFR) trans-membrane (TM) domain was cloned onto the 3'end of the CH3 region of P-dbIgG1 (Figure 9). The PDGFR TM domain was amplified by PCR from the expression vector pDisplay (Invitrogen) using the Expand High Fidelity PCR system as per section 2.1.10. Oligonucleotide primers PDGFR-rev and Link PCR4-g were used as the sense and anti-sense primers



Figure 3. Strategy for the Isolation of gp41 specific mAbs using a mammalian Cell Display system. VH and VL genes were amplified from the cervical B cells from two HIV resistant sex workers from Nairobi, Kenya and cloned into Heavy and Light chain expression vectors. Co-transfection into HEK 203F cells yielded a pool of cells expressing full-length IgG1/Kappa/Lambda antibodies encoding the V regions from the IgA expressing cervical B cells. FACS was done to isolate cells expressing surface IgG that had affinity for gp41. VH and VL genes were recovered from a pool of 50 positive cells and cloned into heavy and light chain expression vectors. The TM region was removed from the heavy chain vector to allow for antibody secretion. 96 clones from both the heavy and light chain ligations were sequenced, and productive heavy and light chain combinations were co-transfected into HEK293F cells. Secreted antibody was tested for expression, cross-reactivity and affinity for gp41.

respectively to add an NheI recognition sequence to the 3'of PDGFR-TM, and the reverse complement of the 3' terminus of IgG1-CH3 to the 5' end of the PDGFR-TM. This resulted in the display of VH IgA libraries on IgG1 constant regions.

The IgG1 constant region CH1-3 was amplified off of P-dbIgG1 using the Expand High Fidelity PCR system with IgGconF and Link PCR3-g as the sense and anti-sense oligonucleotide primers respectively. This resulted in the addition of a portion of the 5' end of the PDGFR-TM domain onto the 3' end of the CH3 region. Both PCR products IgG1 CH1-3, and PDGFR-TM were isolated by gel electrophoresis as per section 2.1.8, 2.1.9.

To link IgG1 CH1-3 with PDGFR-TM overlap PCR was performed using the Expand High Fidelity PCR system. Fifty micromolar of both PCR products were included in the 50 µl reaction volume as template DNA along with 600 nM of IgGconF and PDGFR-rev as the sense and anti-sense oligonucleotide primers. The reaction was cycled as follows: 94°C for 5 min, 7 cycles at 94°C for 30 sec, 50°C for 45 sec, and 72°C for 2 min, 30 cycles at 94°C for 30 sec, 55°C for 30 sec, and 72°C for 1 min\* (15 sec increase in incubation time with every cycle), and one final cycle of 72°C for 7 min. PCR fragments were analyzed by gel electrophoresis as per section 2.1.8 and a band corresponding to approximately 1200 bp (CH1-3 [1050bp] + PDGFR-TM [150 bp]) was excised and purified using a Qiagen Gel Extraction Kit.

The overlap PCR product, as well as the expression vector P-dbIgG1, were double digested using the restriction enzymes ApaI and NheI in NEB buffer 4 and purified using gel electrophoresis following a similar protocol as section 2.3.1. The digested IgG1 CH1-3 – PDGFR-TM PCR product was ligated with the linearized P-

dbIgG1 lacking regions CH1-3 using T4 DNA Ligase following a similar protocol outlined in section 2.4. The resulting ligation was transformed into chemically competent TOP10® *E. coli* cells, plated on LB plates with zeocin (25 μg/ml), 10 clones were selected, amplified, and sequenced with flanking primers PDGFR-rev, and SeqF1 as per section 2.4 to verify the presence and orientation of the insert.

## 2.6.2 Creation of P-dblgG1-B12-TM for Positive Control

The B12 VH region was cloned from P-dbIgG1B12 into P-dbIgG1-TM following the protocol outlined in section 2.4.

#### 2.6.3 Ligation of Resistant Cervical (CMC-R) VH-α Genes into P-dblgG1-TM

VH-α regions amplified by extension PCR, and digested per section 2.3.1 were pooled and ligated into P-dbIgG1-TM creating a plasmid library encoding a population of VH genes from the cervical B cell IgA repertoire of two Kenyan HIV-1 resistant sex workers: ML 1356; and ML 1515. The digested V regions were pooled into 7 groups, ligated into P-dbIgG1-TM and precipitated as per section 2.5.1. Ligations were transformed into electrocompetent *E.coli* TOP10® cells, and isolated using a Qiagen MaxiPrep Kit as per section 2.5.3. Individual Maxipreps were pooled forming one P-dbIgG1-TM library.

## 2.6.4 Ligation of CMC-R VL-κ,λ into P-dbKappa and P-dbLambda

VL- $\kappa$ , $\lambda$  genes from the CMC-R sample were amplified by two rounds of PCR and digested by restriction endonucleases per section 2.3.2 were pooled and ligated

into P-dbKappa or P-dbLambda, then precipitated as per section 2.5.2. Ligations were transformed into electrocompetent *E.coli* TOP10® cells, and isolated using a Qiagen MaxiPrep Kit as per section 2.5.3. Individual Maxipreps were pooled together forming one P-dbKappa and one P-dbLambda library.

# 2.6.5 Determining the Diversity of the CMC-R Library

Twenty five clones from the P-dbIgG1-TM encoded VH- $\alpha$  library and 10 clones each from the CMC-R P-dbKappa and P-dbLambda VL- $\kappa$ , $\lambda$  libraries were selected, expanded, sequenced and analyzed for diversity as per section 2.5.4. DNAstar software and a ClustalW alignment algorithm was used to generate a neighbor joining guide tree for the 21 productive heavy chain sequences isolated.

## <u>2.7 Assessment of IgG1/κ,λ Surface Expression on 293F cells</u>

#### 2.7.1 Transfection

Either P-dbIgG1B12-TM and P-dbKappaB12, or PdbIgG1-TM, PdbKappa and PdbLambda encoding the cervical VH- $\alpha$  / VL $\lambda$ , $\kappa$  library from ML1515 and 1356 were transfected into HEK 293F cells as per section 2.5.6. As a negative control, 40 ng of P-dbKappa-SVK18 (an irrelevant a light chain sequence) was transfected following the same procedure.

# 2.7.2 Direct Immunofluorescent Cell Staining

At three different time points following the transfection, cells were counted using a hemacytometer (Reichert) and the viability determined using trypan blue

exclusion (Gibco). Cells were then harvested from the transfections via centrifugation for 5 min at 800 g and re-suspended to a concentration of 2 x 10<sup>7</sup> cells/ml in 2% BSA, 0.1% Sodium Azide (Sigma) (FACS Buffer). Cells were incubated on ice for 10 min and 50 µl aliquots from each transfection (1 x 10<sup>6</sup> cells) were subsequently dispensed into 1.5 ml Eppendorf tubes. One of the following labeled primary reagents was added to each tube and incubated on ice for 50 min with frequent vortexing: 1 µg FITC labeled gp120 (ImmunoDiagnostics); 0.5 µg PE labeled goat anti-human IgG Fc (Southern Biotech); 0.5 µg FITC labeled goat antihuman Kappa (Southern Biotech); and 0.5 µg FITC labeled goat anti-human Lambda (Southern Biotech). Cells were washed three times with 800 µl of FACS buffer and re-suspended in 300 µl of FACS buffer for analysis.

# 2.7.3 Flow Cytometry Analysis

Cells were stored on ice in the dark for up to an hour prior to analysis. Cells were analyzed using a MoFlo XDP Cell Sorter with the help of Monroe Chan, University of Manitoba, Faculity of Medicine. The percentage of gated cells with fluorescence levels in the positive (R2) region when compared with the negative control was determined for each population.

# 2.8 Fluorescence-Activated Cell Sorting (FACS)

# 2.8.1 Transfection

The plasmids encoding the P-dbIgG1-TM VH- $\alpha$  library along with the plasmids encoding the P-dbKappa and P-dbLambda VL- $\kappa$ , $\lambda$  libraries were co-transfected into HEK 293F cells as per section 2.7.3. The transfection was incubated for 72 hr at 37°C with 8% CO<sub>2</sub> and shaken at 135 rpm.

## 2.8.2 Indirect Multi-Colour Immunofluorescent Cell Staining

Following the incubation, cells were harvested, counted, blocked, and resuspended in FACS buffer at a concentration of 2 x  $10^7$  cells/ml as per section 2.7.2. A total of 15 million cells was added in 100 µl aliquots (2 million cells) to eppendorf tubes containing 2 µg of biotin-labeled gp41 (Capital Biosciences) diluted in 5 µl of FACS buffer. Samples were incubated on ice for 1 hr with frequent vortexing and washed twice with FACS buffer. Following re-suspension of the cells in 100 µl FACS buffer, 10 µl of FACS buffer containing 1 µg of R-PE labeled Goat anti-Human IgG Fc, and 1 µg of dylight 488 labeled streptavidin (Jackson ImmunoResearch) was added to each sample and incubated on ice for 1 hr with frequent vortexing. Samples were washed 3 times with FACS buffer and re-suspended in a total of 2 ml FACS buffer at 7.5 x 10<sup>6</sup> cells/ml.

# 2.8.3 FACS Isolation of gp41/IgG Double Positive Cells

Cells were dispensed into 5 ml polystyrene round-bottom tubes (Falcon) by passing through a 0.35  $\mu$ m cell-strainer (Falcon) and analyzed using the MoFlo XDP Cell Sorter. Side scatter plots were used to exclude dead and clumped cells. The gating region was set so that approximately 0.2% of the gated cell population would

fall within the defined parameters. This region would include cells that both express high levels of IgG on their surface, and bind with high affinity to recombinant gp41. In total, 70 000 Dylight 488 (FL1) and R-PE (FL2) positive cells were collected into 2 ml of RNAprotect Cell Reagent (Qiagen). As it was not possible to isolate individual cells, pools of cells were diluted in RNAprotect Cell Reagent to 50, 1000, or 10000 cells/ml and stored at -80°C.

# 2.9 V Gene Recovery, Selection and Amplification of gp41 Positive Clones

## 2.9.1 RNA Recovery from 50 Cell Sub-Library

Total RNA was recovered from a 50 cell aliquot of the double positive cells isolated per section 2.8.3 using an RNeasy® Micro Kit (Qiagen) following proper aseptic techniques to avoid RNase contamination. Cells were re-suspended in 75 µl of buffer RLT (Qiagen) with 0.143M β-mercaptoethanol (Bio-Rad). Due to the small cell number, 20 ng of carrier RNA (Qiagen) was added to the sample. One volume of 70% ethanol was added to the sample and transferred to an RNeasy MiniElute spin column and centrifuged for 15 s at 8000 g. The sample was washed by passing 350 µl buffer RW1 (Qiagen) through the column by centrifuging as above. To avoid DNA contamination, 27 Kunitz units of DNase I diluted in 80 µl buffer RDD (Qiagen) was added to the column and incubated at room temperature for 15 min. The sample was then washed once with 350 µl buffer RW1, once with 500 µl buffer RPE (Qiagen), and once with 80% ethanol by centrifuging as above. The column was thoroughly dried by centrifuging with the spin column lid open at 13000 g for 5 min.

RNA was eluted in 14  $\mu$ l of RNase-free water (Qiagen) by centrifuging for 1 min at 13000 g.

## 2.9.2 cDNA Synthesis Using Sensi-Script®

RNA was converted to cDNA using Sensi-Script® Reverse Transcription Kit (Qiagen). The entire volume (12 μl) of RNA eluted per section 2.9.1 was added to 8 μl of master mix containing: 1X buffer RT (Qiagen), 0.5mM each dNTP (Qiagen), 1 μM Oligo-dT primer (Invitrogen), 10 Units of Superase•IN RNase Inhibitor (Ambion), and 1 Unit of Sensiscript Reverse Transcriptase (Qiagen). The reaction was incubated at 37°C for 60 min.

# 2.9.3 VH and VL Amplification from 50 Cell Sub-Library

VH and VL genes were amplified from the 50 cell sub-library of gp41 positive clones using the Expand High Fidelity PCR System. Next, 5 µl of cDNA created in section 2.9.2 was used as the template with each of primer sets outlined in section 2.5.10. Reactions were cycled as follows: 94°C for 5 min, 15 cycles at 94°C for 30 sec, 55°C for 30 sec, and 72°C for 45 sec, 30 cycles at 94°C for 30 sec, 55°C for 30 sec, and 72°C for 45 sec increase in incubation time with every cycle), and one final cycle at 72°C for 7 min. PCR products were visualized and purified as stated previously.

## 2.9.4 Cloning gp41 specific VH and VL Sequences into Expression Vectors
Recovered VH and VL sequences as well as P-dbIgG1, P-dbKappa, and P-dbLambda were digested with appropriate restriction endonucleases and purified as per section 2.3.1. 200 ng of each of the digested VH- $\alpha$ , VL- $\kappa$ , and VL- $\lambda$ , regions from the 50 cell sub-library were ligated into 500 ng of linearized P-dbIgG1, P-dbKappa or P-dbLambda following protocols outlined in section 2.5.1.

# 2.9.5 Plasmid Isolation and Sequencing

Ligations from the sub-library were each transformed into chemically competent TOP10® *E.coli* cells and plated on LB agar with zeocin as per section 2.3.1. A total of 96 clones from the P-dbIgG1 ligation, 48 clones from the P-dbKappa ligation and 48 clones from the P-dbLambda ligation were transferred from the plate into individual wells of two 96 deep well blocks containing 1 ml of LB-Lennox broth with 25 µg/ml of zeocin. Blocks were incubated overnight at 37°C with shaking at 250 rpm. Plasmids were isolated from each well using a Montage Plasmid Miniprep96 Kit (Millipore) following manufacturer's instructions.

Plasmids were sequenced as outlined in section 2.5.4 using SeqF1 as the sense primer, and either IgG1 Reverse, or PdbLiRevSEQ as the anti-sense primer. Sequences from the heavy and light chain clones were aligned using the ClustalW alignment algorithm to generate individual guide trees using DNAStar software. From the heavy chain guide tree, 11 productive immunoglobulin VH sequences were identified. From the light chain guide tree, 13 productive immunoglobulin Vκ sequences were identified.

#### 2.9.6 Chain Shuffling Transfection with V Genes from the gp41 +ve Sub-Library

To identify the VH and VL combinations resulting in functional IgG1-κ antibodies that bind to gp41, each of the 11 VH clones were co-transfected with each of the 13 VL clones into HEK293F cells. In sterile 96 well plates, 1 µg of light chain vector and 0.8 µg of heavy chain vector were diluted into 30 µl Opti-Mem®. 293fectin<sup>™</sup> was diluted 1/12 in 5.25 ml Opti-Mem®, incubated for 5 min at room temperature, and added in 33 µl increments to the DNA. DNA-293fectin<sup>™</sup> complexes were formed by incubating the reactions for 25 min at room temperature. Each reaction was immediately added to a well of a 24 well tissue culture plate (Corning) containing 1.1 x 10<sup>6</sup> cells in 1 ml of FreeStyle<sup>™</sup> 293 Expression Medium. Sterile H<sub>2</sub>O was added in between the wells to prevent desiccation. Transfections were incubated for 48 hrs at 37°C and 8% CO<sub>2</sub> with shaking at 250 rpm.

## 2.9.7 Screening the Chain Shuffling Transfection via ELISA

Antibody expression and affinity for gp41 was determined for each VH/VL combination using a standard ELISA. 200 ng of recombinant gp41, 200 ng of goat anti-human IgG F(ab')<sub>2</sub>, and 200 ng of BSA were coated onto individual wells of a MaxiSorp 96 well plate and blocked as per section 2.5.9. 60 µl of supernatant from each transfection was added to the wells and incubated for 2 hrs at 37°C. Plates were then processed as per section 2.5.9 and the absorbance at 405 nm was determined.

### 2.9.8 Selection and Amplification of gp41 Specific Antibodies

Four mAbs with the highest OD value against gp41, H3L2, H4L9, H6L9, and H9L2 were chosen from the 144 different VH/VL combinations tested. The four VH regions encoded in P-dbIgG1, and the two VL regions encoded in P-dbKappa were transformed into chemically competent One Shot TOP10® *E.coli* cells as per section 2.3.1. Glycerol stocks for each plasmid were made by diluting 200 µl bacterial culture in 800 µl of 33 % glycerol and stored at -80°C. Plasmids were expanded and isolated using the Qiagen Plasmid MaxiPrep as per manufacturer's instructions.

### 2.10 Large Scale Antibody Expression

## 2.10.1 Large Scale Transfection of gp41 mAbs

HEK 293F cells were grown to 150 ml volumes in 500 ml flasks and transfected with 300  $\mu$ g of light chain expression vector, and 150  $\mu$ g heavy chain expression vector for each of the four mAb combinations following a scaled up version of the protocol outlined in section 2.5.6. Cells were incubated for 5 days at 37°C with 8% CO<sub>2</sub> and shaking at 135 rpm.

# 2.10.2 Concentration and Purification of gp41 MAbs

Supernatants from the four transfections were concentrated 10 to 15 times using a Centriprep Centrifugal Filter Device (Millipore) with an Ultracel® 30 kDa cut-off membrane. Supernatants were then mixed 1:1 with PBS pH 7.2 and loaded onto a 1 ml Protein A column (GE Healthcare) that had been pre-equilibrated with 10 volumes of PBS. Flow through was collected and passed over the column twice more to collect any residual antibody. The column was washed with 10 ml PBS and eluted with 10 ml 0.5 M acetic acid pH 3.0 in 1ml fractions into 1.5 ml microfuge tubes containing 100  $\mu$ l of 1.0 M Tris-HCl pH 8.0. The fractions were pooled and dialyzed three times against PBS at 1500 g and concentrated to a final volume of 1.5 ml using the centriprep device. Each sample was filter sterilized by passing through a low protein binding 0.2  $\mu$ m filter (Millipore).

## 2.11 Characterization of the gp41 MAbs

# 2.11.1 Measurement of Antibody Specificity for gp41

ELISA titrations of the 4 gp41 mAbs in 0.1% BSA versus 200 ng of gp41, gp120, and BSA were performed using standard ELISA techniques as per section 2.5.9.

## 2.11.2 Measurement of Antibody Reactivity with Denatured/Reduced gp41

Recombinant gp41 was heat - denatured by incubating in H<sub>2</sub>O at 99.9°C for 20 min in a thermocycler, then diluted 1/10 in PBS for coating. Additionally, recombinant gp41 was reduced by incubating for 20 min in 0.05 M DTT (dithiothreitol) (Invitrogen) then diluted 1/10 in PBS for coating. The process was repeated for recombinant gp120. ELISA titrations in 0.1% BSA for the 4 gp41 mAbs were performed versus 200 ng of denatured/reduced gp41, while ELISA titrations

for IgG1B12 were performed versus 200 ng of denatured/reduced gp120 as per section 2.5.9.

# 2.11.3 Assessment of Antibody Binding to HIV-1 Envelope Spike (gp160) using Flow Cytometry

HEK 293F cells were grown to 30 ml volumes and transfected with 30 µg of one of the following HIV-1 envelope expression vectors as per section 2.5.6: consensus clade B Env expression vector (pConBgp160-opt); or consensus clade C Env expression vector (pConCgp160-opt). P-dbKappa (SVK18) (irrelevant light chain expression vector) was also transfected as a negative control. Both HIV-1 Env expression vectors were obtained through the AIDS Research and Reference Reagent Program, Division of AIDS, NIAID, NIH from Dr. Beatrice Hahn. Vectors contained the consensus clade B or C Env amino acid sequence ligated into the pcDNA3.1 (-) expression vector and optimized for expression in mammalian cells. Both vectors had been amplified using a Qiagen MaxiPrep kit as outlined previously. Cells were incubated for 48 hr at 37°C with 8% CO<sub>2</sub> and shaking at 135 rpm.

Following incubation, 1 x 10<sup>6</sup> cell aliquots were washed once with FACS buffer, and re-suspended in 0.1 or 10  $\mu$ g/ml of mAb H3/L2, H4/L9, H6/L9, H9/L2, or G75. Cells were incubated on ice for 1 hr with frequent vortexing and washed three times with FACS buffer. Cells were re-suspended in 75  $\mu$ l FACS buffer with 6.7  $\mu$ g/ml R-PE labeled Goat Anti-Human IgG and incubated for one hour on ice. Cells were washed 3 more times and re-suspended in 300  $\mu$ l FACS buffer. 10 000 cell

aliquots were analyzed as per section 2.7.3 and the median R-PE fluorescence was determined each for cell population.

# 2.11.4 Assessment of Antibody Binding to HIV-1 Envelope Spike (gp160) using Fluorometric Microvolume Assay Technology (FMAT)

Transiently transfected HEK293F cells expressing consensus HIV-1 envelope clade B, clade C or SVK18 were prepared as per section 2.11.6. Cells were then counted and the viability determined using trypan blue exclusion. All three cell populations were washed twice with 1% BSA and re-suspended at 5 x 10<sup>5</sup> cells/ml.

Six micron polystyrene beads (Spherotech) were passively coated with either recombinant gp41 or BSA as follows: 200  $\mu$ l of beads were added to 1 ml PBS (pH 7.2), vortexed, centrifuged at 13000 g for 1 min, and supernatant discarded. The beads were washed a second time in PBS following the same procedure. Beads were then re-suspended in 1 ml PBS containing 100  $\mu$ g of recombinant gp41 or BSA and tumbled overnight at 4°C. Beads were washed twice as stated above, re-suspended in 1 ml 1% BSA, and tumbled for 2hr at room temperature. Finally, the beads were washed two more times and re-suspended in 1 ml of 0.1% BSA.

To run the assay, either 2 x  $10^4$  of the transfected cells, or 1 x  $10^4$  antigen coated beads were added in 40 µl of dilution buffer (1% BSA in PBS) to each well of a FMAT 8100 HTS 96 well black-clear bottom plate (Applied Biosystems). Next, 50 µl of dilution buffer containing 0.4 ug/ml of FMAT Blue goat anti-human IgG (Fc) (Applied Biosystems) was then added to each well. Then, 15 µl of dilution buffer containing 1/2 serial dilutions (8 µg/ml  $\rightarrow$  0.016 ug/ml) of the following purified

primary antibodies were added to the wells: H3/L2; H4/L9; H6/L9; H9/L2; B12; or G75. Following the simultaneous addition of the beads/cells, secondary, and primary antibodies, the plates were incubated in the dark for 4 hr at room temperature and analyzed using the Applied Biosystems 8200 Cellular Detection System (8200 cds). Event counts were measured to determine positive binding. A minimum of 50 events was set as the cut-off for a positive sample, and titers were recorded as the lowest mAb concentration with an event count over 50.

# 2.11.5 Sequence Analysis of gp41 mAbs

Nucleotide and amino acid sequences of the VH and VL genes (5' end of framework 1, to the 3'end of framework 3) from the 4 novel gp41 mAbs generated herein were compared with known human germ line VH and VL genes using IMGT/V-Quest software :

(http://www.imgt.org/IMGT\_vquest/vquest?livret=0&Option=humanIg) [143]. Heavy and light chain framework (FR) and complementary determining regions (CDR) sequences were analyzed using V-Quest software and the R:S mutation ratio for each region (CDR1-2, or FR1-3) were calculated using the following equation: R:S = (r /s)

Where r = total number of replacement mutations in CDR1,2 or FR1-3 And s = total number of silent mutations in CDR 1,2 or FR1-3.

The multinomial model developed by Lossos et al (2000) was used to compare the observed R:S value of each VH and VL sequence to the expected R:S value of the closest related germline sequence given random mutation [142]. Germline FR and CDR sequences along with the number of observed R and S mutations in each region were input into the JAVA applet available at:

# http://www-stat.stanford.edu/immunoglobulin [142].

Using this model, the probability (*Pm*) could be calculated that the observed excess or scarcity of replacement mutations in the CDRs or FRs respectively occurred by random mutation and not by antigen-driven affinity maturation.

## **3 Results**

## 3.1 Generation of an Antibody Secretory Library from HIV-1 positive PBMCs

## 3.1.1 V Gene Amplification and Library Generation

To investigate the ability of a mammalian cell antibody secretion system to isolate HIV-1 Env specific antibodies, an antibody library was initially developed from the bulk, unselected V genes of a B cell population known to contain clones specific to HIV-1 Env. The PBMCs of the HIV-1 positive sex worker ML256 were chosen as the source for the B cells based on a the high serum titer to both gp120 and gp41 (Figure 4A, B). VH genes from families 1-5 of the IgG isotype, and VL genes of the Kappa and Lambda isotype were amplified from these PBMCs (Figure 4C) and cloned into expression vectors of the appropriate isotype designed for the secretion of soluble antibody (Figure 5A-C). Supernatants from the co-transfection of heavy and light chain plasmids encoding either the V gene library or B12, were screened for expression of IgG, IgKappa, IgLambda, and reactivity to recombinant gp120 and gp41. As indicated in Figure 5D, heavy chain and light chain expression was achieved for both the antibody library and B12, while the IgG1/Kappa B12 transfection maintained antigen affinity for gp120. The Diversity of the PBMC antibody library was assessed by AluI digest (Figure 6B), and by sequence alignment using the clustalW alignment algorithm (Figure 6A). A diverse selection of VH digestion patterns suggests a diverse VH library (Figure 6A), while Figure 6B demonstrates that 4 different VH families were represented in the 15 clones that



(C)



Figure 4. Cloning the IgG/ $\kappa$ , $\lambda$  PBMC antibody repertoire from ML 256 (HIV-1 positive). Antibody serum titer of ML 256 against recombinant gp120 (A) and recombinant gp41 (B) compared with HIV-1 negative serum. Amplification of antibody V regions from the PBMCs of ML 256 (c). VH domains 1-6 of the IgG1-3 isotype (orange), VL $\lambda$  domains 1-10 and VL $\kappa$  domains 1-5(red) were extracted from the gel and cloned into expression vectors.



Figure 5. Secretory antibody expression system. P-dbKappa (A), P-dbLambda (B) and P-dbIgG1 (C) expression vectors. ELISA of antibody expression and specificity of supernatant when 293F cells were transfected with B12 VH/VL cloned into P-dbIgG1/P-dbkappa, or the ML256 PBMC VH/VL library cloned into PdbIgG1/P-dbKappa/P-dbLambda (D). Supernatants were tested for IgG, IgK, and Ig $\lambda$  expression, as well as reactivity with recombinant gp120 and gp41.

(A)

	FR	-1	CDR1	FR-2	CDR2		FR	2–3		CDR3		
clone 12	ISVQLLESGGGLIEF	GGSLRLSCAVS	SGFSFGNAWMS	WVRQAPGKGLEV	WVGLIKSNTDGG	TTEYGAPVRGR	FTISRDDSKNS	SLYLQMNSLK	TEDTAVYYCTT	ARRWELG	PDFWGQGT	LVTV
clone 10	ISVQLLESGGGLVQE	GGSLRLSCVGS	SGFVFSDFAIS	WVRQAPGKGLEV	WVSAFSGNRD	ATWYADTVKGR	FSISRDNSKNA	AVFLQMNSLR	VEDTAIYYC <mark>AK</mark>	ESPYRSGSRE-	YYFDSWGQGI	LVTV
clone 9	ISVQLLESGGGLVKE	GGSLRLSCAAS	SGFTFSDYYMS	WIRQAPGKGLE	WVSYLSYSSN	TIYYADSVQGR	FTFSRDNAKNS	SLYLQMNSLR	DDDTAVYFCAR	QKEPG		LVTV
clone 8	ISVQLLESGGGLVKE	GGSLRLSCAAS	SGFTFSDHYMN	WIRQAPGKGLE	WVSYISSSGT	TIYYADSVKGR	FTISRDNAKNS	SLYLQINILR	AEDTAVYYC <mark>AK</mark>	EGVGDGYNLGY	YYFLDVWGKGT	YTVA
clone 7	ISVQLLESGGGVVQF	GGSLRLSCAAS	SGFTFRDYGIH	WVRQAPGKGLE	WVSYIRHHGR	EMYYADSVKGR	FTISRDNSKNI	LSLQMNSLR	VDDTAVYFC <mark>AK</mark>	DYDWA	LDIWGQGT	RVTV
clone 5	ISVQLLESGGGVVQE	GGSLRLSCAAS	SGFTFSSYWMT	WVRQAPGKGLEV	WVANIEQDGT	QKYYVDSVKGR	FTISRDNAKNS	SLYLQMNSLR	VEDTAVYYCAR	GLLGD	YWGQGT	LVTV
clone 3	ISVQLLESGGRLVQE	GGSLRLSCVGS	STFTFTRYAMS	WVRQVPGRGLE	WVAAVSSDGA	HTFYPDSVKGR	FTISRDNSKNI	LYLQMNSLR	AEDTAVYYC <mark>AK</mark>	-TPAKLLTHY-	YYMDVWGKGT	TVTV
clone 2	ISVQLLESGGGLVQE	GGSLRLSCAAS	SGFTFSSYWMT	WVRQAPGKGLEV	WVANIKQDGS	EKFYLDSVKGR	ITLSRDNAKNS	SLYLQMNSLR	AEDTAVYHCAR	GLLGD	YWGQGT	LVTV
	********* *	*******	* * * .	*.** **.***	* *	* *.**	. *** .**.	. **.* *.	.***.* *.		**.*	***
clone 14	ISVQLLESGPGLVKF	SQTLSLTCAIS	SGDSVSNNTSA	AWNWIRQSPSR	GLEWLGRTYYRS	KWYNDYAVSVK	SRITINPDTSK	NQFFLQLNS	VTTEDTAVYYC	ARDCSNSWKWL	D-PLWGQGTLV	/TV
clone 11	ISVQLLESGPGLVKF	SQTLSLTCAIS	SGDCVSSNS-A	AWNWIRQSPSR	GLECLGRTYYRS	KWDNDYAVSVK	SPITINPDTSK	NQFFLQLNS	VTPEDTAVYYC	ARDGSNSWYWF	DNPLWGQGTLV	TV
	******	*******	*** ** ** *	*********	*** *******	** *******	* *******	*******	** *******	*** **** *	* * ******	**
clone 15	TSVOLLESCOCLVK	SET1.SI.TOTV	SCOLWROAWYW	NWIRODDCCVG	.FPMCVTVVSAK	THVNDST.KSP	VTTSVDTSKNC	FGT.KT.SGWT		ALSCCOPERVM	IGOGTI VSV	
clone 4	ISVOLLESCECLVKE	SETLOLICIVO SETLSI.TCTV	SCCS-TSSVVW	NWIROPDG-KG	LERMOVIVVSG-	STHANDSI'K SB	VTISVDISKNO	FGI KI GGVT	ANDTAVVVCAS	CCECCUDEEVW	ICOCTLVSV	
cione i	*************	*********	*** **	*******	*** *******	**********	**********	*********	*******	** *****	*******	
clone 1	ISKKPGESLKISCKG	SGYSFGNYWIC	GWVRQMPGRGL	EWMGIIDPGDSI	ETRYSPSFQGQV	IMSVDRSSGTA	FLQWNTLGASE	DTAMYYCTRY	PRSSCGDSCSW	DVQFWGQGTLV	VIV	
clone 6	ISKKPGESLKISCKG	SGYSFGNYWIC	SWVRQMPGRGL	EWMGIIDPGDSI	ETRYSPSFQGQV	IMSVDRSSGTA	FLQWNTLGASL	DTAMYYC'I'RY	LRSSCGDSCSW	DVQFWCQGTLV	/1V	
cione 13	ISKRPGESLKISCKG	SGISFISYWI	JWVRQMPGRGL	EWMGIIDPGDSI	ETRISPSFQGQV	TMSVDRSSGTA	FLQWNTLXASE	JTAXYYC <mark>TRY</mark>	KRSSTGDSCSW	CVQFWGQGTLV	TV.	
	***********	****		********			****		*******		^ ^	

(B)



Figure 6. Diversity of the constructed ML 256 PBMC library. 15 clones were selected from the IgG vector library and analyzed by sequencing (A) and AluI digest (B). Clones were aligned based on their heavy chain variable region domain. Greatest diversity is seen in the CDR3 region (red) of the selected clones. Asterisks indicate totally conserved amino acids, while periods indicate fairly conserved amino acids.

that were sequenced and, as expected, diversity was primarily localized to the three CDR regions, in particular CDR3.

#### 3.1.2 The Screening and Selection of Clones from the ML 256 PBMC Library

In order to isolate individual mAbs from the cells expressing the ML 256 antibody library, the cells had to be separated into individual clones in semi-solid media and screened for antibody expression and binding to the Env proteins. Following co-transfection of the heavy and light chain plasmids, the cells were resuspended in semi-solid media with zeocin selection. 1500 Clones were selected, expanded and screened for antibody expression and reactivity to a recombinant gp120/gp41 mosaic antigen in an ELISA. Figure 7A summarizes the antibody expression profiles of the 1500 clones tested. The 9 clones with the highest antibody expression and binding to the gp120/41 mosaic antigen are shown in Figure 7C. The top two clones, 2G7 and 5B5 were selected for further sub-cloning.

#### 3.1.3 V Gene Recovery and Expression

Due to their limited lifespan, further characterization of the 2 clones selected required that their V genes be recovered from and cloned back into the expression vectors. Two determine the optimal method of V gene recovery, two methods were explored: V gene amplification using the plasmid DNA in the clones as a template in the PCR reaction; or V gene amplification using cDNA synthesized from the recovered RNA as a template. Figure 8 A,B indicates that only 1 VH sequence could



Figure 7. Initial Screening of 1500 293F clones transfected with the ML256 PBMC library. Results summary (A) and corrected ELISA OD values (OD value against BSA subtracted from all values) of the top 9 clones for antibody expression (Fab) and affinity to a recombinant gp120/41 mosaic antigen (B). Two clones, 2G7, and 5B5 were selected for V gene recovery based on their expression levels and OD values against gp120/41.

Clone

be recovered from clone 2G7 using plasmid DNA as the template, while 2 VH (5B5a, 5B5b) and 2 VL sequences (VL1,VL2) were recovered from clone 5B5 when the RNA isolation method was used to recover the V genes.

Recovered V genes were co-transfected into HEK-293F cells in various heavy and light combinations including B12 VL (4L) and G75 VL (anti-PA antibody) as controls (Figure 8C). Supernatants were screened for expression of Fab, and reactivity against monomeric, recombinant gp120 and gp41. The OD values of each supernatant vs. BSA were subtracted as background. One novel combination, 5B5a/VL2 had a corrected OD value vs. gp41 of approximately 0.5 (Figure 8C).

# 3.2 Generation of a Mammalian Cell Antibody Display System: HIV-1 Resistant CMC IgA Antibody Library

# 3.2.1. Amplification of V genes and Library Generation

A novel strategy was developed to isolate gp41 mAbs from cervical IgA B cells using mammalian cell display (Figure 9). VH families 1-6 were amplified from the cervical IgA repertoire of two Kenyan HEPS: ML 1356 and ML 1515 (Figure 9A,B). VH- $\alpha$  and VL-kappa/lambda bands (400 bp) were cloned into antibody expression vectors. VL-kappa/lambda regions were ligated into the kappa/lambda expression vectors used previously (Figure 5A, B). Due to the inefficiencies of the antibody secretion system (summarized in Figure 7A), a new vector was generated by inserting the human PDGFR-TM domain onto the C terminus of the IgG CH3 in P-dbIgG1 creating P- dbIgG1-TM (Figure 9C).





SISEVQLQQSGPVLVKPGASVKMSCKASGYTF TDYYVNWVKQSRGKSLEWLGLIIPSNGGTTYNQK FRGKATLTVDKSSSTAYMELNSLTSEDSAVYY CARRGLTGALFAYWGQGTLVTVSAAMASTKGP

\*Unable to recover light chain sequence of 2G7

#### (B) RNA V gene Recovery Revealed 2 VH Sequences in Clone SBS



\*Recovered 2 VL-A sequences from clone 5B5

VQLLESGGDRVQPGSLRLSCTSSGFNFGDYAIHWVRQ AAGGGLEWVGFISNKNESQTRYAASVQRFSISRDESRD IAYLOMNRLRVEDTARYYCVREPGGAYAFDYWGQGTL VTVSASTKGPSVFPLAPSSKSTGGTA

VQLLESGPQLVKPSQTLSLTCVISGDSVSSNAAWNW VRQSPSRGLEWGRTYYRSKWYNDYALSVKSRITIK SDISKNQFSLQLNSVTPEDMAVYYCAREYSSSSGY FQHWGQGTLVTSV6SASTKGP6VFP

(C)



Figure 8. V gene recovery and chain shuffling from clone 2G7 and 5B5. One productive VH sequence was recovered from 2G7 using the DNA method (A), while two productive VH and VL sequences were recovered from clone 5B5 using the RNA method (B). Chain shuffling transfection using 3 novel sequences reveals one novel combination (5B5a / VL2) with a corrected OD value against gp41 of 0.5 (C).

(A)			
Isotype	V Family	1515 (ML)	1356 (ML)
IgA1,2	1,3,5		X
IgA1,2	2		
IgA1,2	4	X	Χ
IgA1,2	5		
IgA1,2	6	X	
IgA1	1,3,5	X	X
IgA1	2		X
IgA1	4	X	
IgA1	5	X	
IgA1	6	X	



(C)



Figure 9. Creation of the VH –  $\alpha$  antibody library from the CMCs of ML 1356 and ML 1515. Summary of the different V families that were successfully isolated for each IgA isotype from ML1515 or ML1356 (A). Agarose gel electrophoresis of PCR amplified antibody V regions from ML1515 (B). The human PDGFR-TM region was cloned into P-dbIgG1 to create the cell surface IgG antibody expression vector P-dbIgG1-TM (C).

Co-expression of this vector with P-dbKappa/P-dbLambda permitted antibody expression on the cell surface of mammalian cells, providing a better link of the antibody's phenotype to its genotype.

Once the VH-α regions were cloned into P-dbIgG1-TM, the diversity of this library was assessed by transforming *E.coli* cells, selecting, and sequencing 24 clones, and aligning the sequences using a ClustalW alignment algorithm creating a guide tree (Figure 10). Of the 24 clones selected, 20 had productively re-arranged heavy chain sequences. 13 distinct VH re-arrangements (a minimum of 40 nucleotide differences) were identified from the 20 productive sequences which were comprised of VH families 3,4 and 5,

#### 3.2.2 Antibody Expression on the Cell Surface in Transient Transfection

P-dbIgG1-TM encoding the CMC VH-α library or B12 VH, was co-transfected with P-dbKappa/Lambda encoding the CMC VL library or B12 VL into HEK-293F cells to assess whether functional IgG was displayed on the cell surface. Transfected cells were stained with R-PE labelled anti-IgG, or FITC labelled anti-kappa antibodies and analyzed by flow cytometry (Figure 11). The R2 region in Figure 11B and C when compared with 8A, indicates that IgG is expressed on the cell surface of a significant portion of the transfected cells. Figure 11E and F, when compared with 8D, indicates that IgKappa is localized to the cell surface by its association with IgG. For both cases, expression of B12 is slightly higher than the CMC library reflecting the fact the library contained a pool of V genes.

To assess the functionality of cell surface antibodies, cells were stained with



Figure 10. The Diversity and V gene composition of the CMC-R VH -  $\alpha$  library in P-dbIgG1-TM. 24 colonies from the heavy chain library were isolated, sequenced, aligned using the ClustalW alignment algorithm and displayed as a neighbor joining guide tree. Sequences were made up of VH region domains 3, 4 and 5. 13 distinct sequences (greater than 40 nucleotide substitutions) were isolated from the 20 clones that had productively re-arranged heavy chain genes.



Figure 11. Assessment of antibody expression on the cell surface. Cells were transfected with: an irrelevant light chain expression vector (A,D); the cervical VH- $\alpha$ , VL $\lambda$ , $\kappa$  library from ML1515/1356 encoded into P-dbIgG1-TM, P-dbKappa and P-dbLambda respectively (B,E); or P-dbIgG1B12-TM/ P-dbKappa4L (C,F). Cells were stained with either R-PE anti-human IgG Fc (A-C) or FITC anti-human IGKC (D-F). The percentage of gated cells with fluorescence levels in the positive region (R2) when compared with the negative control (A, D), are listed.

either recombinant gp41 (CMC library) or recombinant gp120 (B12) and analyzed by flow cytometry. The R2 region in Figure 12D indicates that when IgG1/kappa B12 is transiently expressed on the cell surface, it retains its ability to bind to gp120. As expected, the CMC library displayed low reactivity with recombinant gp41 (Figure 12B).

## 3.2.3 FACS Analysis of Transiently Transfected CMC Library

Once it was determined that the recombinant antibody library was expressed as functional IgG on the surface of transiently transfected mammalian cells, FACS was done to collect only those clones displaying antibodies recognizing recombinant gp41. Scatter-gated fluorescence was done on the two transfectants (SVK18, or the IgG/Kappa/Lambda library), to eliminate dead cells, clumped cells, and cellular debris. Cells were gated at R1 and total of 45.2% of the total cell population expressing the antibody library were analyzed by FACS (Figure 13B). Gated cells were sorted for expression of IgG Fc and binding to biotinylated gp41, detected using R-PE labelled anti-human IgG, and dylight 488 labelled streptavidin respectively (Figure 13C,D). The obvious shift in fluorescence of the cells expressing the antibody library (Figure 13D) when compared with the SVK18 transfectants (Figure 13C) illustrates that while the majority of cells from the library transfectants expressed IgG on their surface, only a small fraction had affinity for recombinant gp41. A region comprising 0.2% of the gated antibody library transfectants was used as the sort window (Figure 13D).



Figure 12. Affinity of cell surface antibody for gp120 and gp41. Cells were transfected with either: an irrelevant light chain expression vector (A,C); the cervical IgA VH- $\alpha$ , VL $\lambda$ , $\kappa$  library from ML1515/1356 encoded into P-dbIgG1-TM, P-dbKappa and P-dbLambda (B); or P-dbIgG1B12-TM/ PdbKappaB12 (C). Cells were stained with either FITC labeled gp120 (C,D) or biotin labeled gp41 followed and detected using dylight 488 labeled streptavidin. The percentage of cells in the positive region (R2), when compared with the negative controls (A,C), are indicated for each sample. Results indicate that surface displayed B12 has affinity for gp120, and a small minority of surface displayed antibodies from the CMC library have affinity for g41.



Figure 13. Isolation of gp41 specific antibodies by FACS. Cells were transfected with the negative control (SVK18) (A,C) or the CMC plasmid library (B,D). Scatter-gated fluorescence analysis was done on both samples (A,B) to exclude dead cells, clumped cells, and cellular debris. Live cells were gated on R1 for both samples. For sorting, cells were stained for the presence of cell surface IgG, and affinity to recombinant gp41. The sort window was set to collect the top 0.2% of cells positive for both variables from the CMC library (D) for V gene recovery.

# 3.2.4 VH and VL Recovery and Sub-Cloning

A 50 cell aliquot was removed from the cell population sorted for affinity to gp41 and expression of cell surface IgG. From this 50 cell sub-library, VH and VL genes were recovered from total RNA and cloned into the expression plasmids PdblgG1, P-dbKappa or P-dbLambda. The resulting three plasmid sub-libraries were transformed into *E.coli* cells and either, 96, 84, or 12 clones were selected from the P-dbIgG1, P-dbKappa, or P-dbLambda transformations respectively and sequenced. Productive VH and VL sequences were aligned using the ClustalW alignment algorithm to generate heavy and light chain guide trees. A total of 94 of the 96 PdbIgG1 clones had productively re-arranged VH sequences, while 66 of the 84 PdbKappa and none of the 12 P-dbLambda clones that were selected had productively rearranged VL sequences. Figure 14 shows the genetic diversity of the 94 VH and 66 VL sequences. The VH guide tree is composed of 16 different VH sequences, while the VL guide tree is composed of 15 different sequences. To test which combinations produced functional antibody, 11 of the P-dbIgG1-VH and 13 of the P-dbKappa-VL clones were selected from the guide tree (Figure 14). When one of the chosen clones was composed of related sequences with individual point mutations, a single clone was selected.

Each of the 11 P-dbIgG1-VH clones were transfected with each of the PdbKappa-VL clones in a chain swapping experiment to identify which VH/VL combinations produced a functional antibody with affinity for gp41. Supernatants from each transfection were tested for reactivity against recombinant gp41 and BSA using an ELISA. Figure 15 identifies the 4 heavy chain clones that resulted in gp41



Figure 14. Analysis of recovered VH and VL genes from sorted gp41 positive cells. V genes were recovered from a 50 cell sub-library from sorted gp41 positive cells and cloned into P-dbIgG1/Kappa/Lambda vectors. 94 clones from the VH sub-library (A) and the 66 clones from the VL $\kappa$  sub-library (B) were analyzed. Sequences were aligned using the ClustalW alignment algorithm to generate heavy and light chain guide trees. The 11 productive VH sequences (A) and 13 productive VL sequences (B) recovered from the sub-library and used in subsequent chain shuffling experiments are indicated in green. Underlined sequences are those that were found to have affinity for gp41 in subsequent chain experiments.



Figure 15. Identification of productive VH/VL combinations via chain shuffling. The 11 recovered VH genes were each paired with the 13 recovered VL genes in a chain shuffling transfection in HEK 293F cells. Of the 143 different combinations, 16 produced an antibody with affinity for recombinant gp41. Shown are the 4 productive VH sequences that had affinity for gp41 when paired with the indicated light chains. B12 and G75, a chimeric PA specific mAb previously characterized by our lab, were included as positive and negative controls. The light chains selected as the best partner for each heavy chain is indicated with a circle. These four combinations were selected for further characterization.

specific antibodies when paired with the indicated light chain clones. Supernatant from B12 and G75 transfections were included as positive and negative controls respectively. Promiscuous light chain pairing was observed for each heavy chain clone. The light chain partner resulting in the highest OD value against gp41 and the lowest OD against BSA was selected for each of the heavy chain clones for furthercharacterization. The following combinations were selected: H3/L2, H4/L9, H6/L9, H9/L2.

Figure 16 shows the sequence alignment of the 4 heavy chain sequences and the 6 light chain sequences that produced a gp41 specific antibody in the chainswapping experiment. Each sequence is distinct and the diversity between sequences is most prevalent in the three CDR regions (Figure 16).

## 3.3 Characterization of gp41 Specific MAbs

## 3.3.1 Evaluation of MAb Specificity for gp41

To assess the specificity of the four gp41 mAbs, each were expressed, purified and titrated against gp41, gp120 and BSA (Figure 17). All four mAbs show similar binding curves to gp41 and demonstrate little cross-reactivity with either gp120 or BSA (Figure 17).

### **3.3.2 Determination of the Nature of the MAb Epitopes**

To determine if the 4 mAbs bind to conformational or linear epitopes, each mAb was titrated against gp41 that was either denatured using heat or reduced

(A)

# Heavy Chain Clones

	←					_		-FR	t1					-	-						÷	-	-Cl	DR1			•	←-					-
H3	SΙ	s v	QI	LL	Е	S	G	GG	ι	V	KA	G	G	S	LR	L	s	C	vv	v s	S G	F	Т	F S	SD	Y	Y	М	SV	II	R	QA	l
H9	SΙ	s v	QI	L	Е	S	G (	G G	L	V	K P	G	G	S	LR	L	s	C	v	A S	S G	F	s	F	S D	Y	A	м	SV	٧	R	QA	
H6	SI	s v	QJ	L	Е	S	G	PG	L	VI	K P	S	F	Т	Ľ٤	L	т	d	Т	vs	S G	A	s	I S	S	Y	Y	W	SV	I I	R	Q₽	í
84	SΙ	s v	QI	LL	Е	S	G	G G	L	V	ģΡ	G	G	S	L R	L	S	C	A I	AS	S G	F	P	F	R S	Y	G	М	SV	۷ I	R	QA	
										_														_		_							
	F	R2—			-		<b>*</b>			CDR	2		<b></b> ♦		_				_	-							-	-		-FR	3		
H3	ΡG	E K	LI	ΞW	v	S	Υ	I S	s	G	GN	T	1	H	Y A	D	s	V I	K (	GF	۲۶	Ъ	v	SF	۲D	N	A	ĸ	N S	SL	F	ΙQ	
H9	ΡG	KG	LI	E W	v	S	V I	IH	R	G	GT	'N	т	G	Y A	D	s	v I	к	GF	ξĒ	Ъ	I	SI	2 D	D	s	ĸ	N _	ΓL	Y	гŎ	
H6	<b>A</b> G	ΤG	LI	E W	I	G	R 1	ΙD	T	s	G-	s	т	I	YN	i P	s	L	Q :	S F	×A	A	М	S	1 D	т	s	к	N	Q 7	9	LK	l
Н4	ΡG	KG	LI	S W	v	S	A I	I	D	<b>S</b> 1	DD	R	т	ΥĽ	YA	D	s	v	К	GF	₹F	Ъ	Ι	SF	٦D	S	s	E	N .	ΓL	Y	гŎ	
												-																					
										<b>∢</b>				_	-CD	R3-							-				-F	R4			->		
H3	M N	SЬ	R F	ΑE	D	ТŻ	A 1	ΙY	Y	CI	A R	E	-	-[	GΊ	8	ď	s	-	- 1	D	Y	W	Gç	2 G	т	L	v	ΤV	I S	s		
89	м и	SL	R٦	ΓE	D	т	Λ	J Y	Y	CI	Υ K	E	R	Y	GS	G	W	H.	Y	- 1	D	Y	W	G	2 G	т	L	A	т١	V S	s		
H6	LT	s v	ΤZ	A	D	a i	۸V	V Y	Y	Cł	A R	D	8	D	GΊ	8	W	S	Y	Y	D	Y	W	G	2 G	т	L	v	Т٧	V S	s		
H4	ΜN	sг	R Z	ΑE	D	T /	A٦	γY	Y	Cł	A K	P	L	I	GS	Y	G	s	Q	- 1	D	L	W	G	2 G	т	Q	v	ТΝ	v s	s		

(B)

**Light Chain Clones** 

	←		-	-	-								F	R1																	$\rightarrow$	←		CD	R1			¢	<u>-</u> -
L1	S	I	S	Y	G	Е	L	Ι	L	Т	Q	S	Р	G	Т	L	S	L	S	Р	G	Е	R	A	Т	F	s	С	R	s	S	H	S	Ι	R	s	R	R	v
L2	S	I	s	Y	G	Е	L	I	M	т	Q	s	P	A	т	L	s	L	s	₽	G	Е	R	A	Т	L	s	С	R	Α	s	Q	s	v	G	s	-	Y	L
L6	s :	I	s	Y	G	Е	L	õ	м	т	Q	F	Р	S	F	L	s	A	s	v	G	Е	R	v	т	I	т	с	R	A	s	Q	A	I	G	s	-	Y	L
L9	S	I	s	Y	G	Е	L	Ι	L	т	Q	S	Р	s	S	L	s	A	s	v	G	D	R	v	т	I	т	С	R	A	s	Q	D	Ι	S	N	-	S	L
L10	S	I	s	Y	G	Е	L	Т	м	т	Q	s	Р	s	т	L	s	A	s	v	G	D	R	v	т	I	т	с	R	A	s	Q	s	Ι	G	D	-[	W	L
L11	s :	I	s	Y	G	Е	г	L	L	т	Q	s	Р	s	т	L	s	A	s	v	G	D	R	v	т	I	т	с	R	A	s	Q	s	I	G	D	-	W	L
																																			_				_
					F	R2			-		-				_	÷	←C	DR	2-)	•		-								FR	3						-		
L1	AV	1	Y	Q	н	к	Ρ	G	0	A	Ρ	R	L	v	I	н	G	V	S	N	R	A	S	G	Ι	s	Ъĺ	R	F	s	G	s	G	S	G	т	D	F	т
L2	AV	<b>N</b> 2	Y	0	0	к	₽	G	0	₽	₽	R	L	L	I	Y	D	А	s	к	R	A	т	G	т	P	A	R	F	s	G	s	G	s	G	т	D	F	s
L6	Αī	<b>N</b> 2	Y	õ	õ	Е	Р	G	к	А	Р	к	F	L	I	Y	A	А	s	т	L	0	s	G	v	Р	s	R	F	s	G	s	G	s	G	т	Е	F	т
L9	ΑV	<b>1</b> 1	Y	õ	Ĥ	к	Р	G	к	A	Р	ĸ	L	L	L	F	G	A	S	s	L	Ĥ	S	G	v	S	s	R	F	N	G	S	G	S	G	т	D	F	т
L10	ΑV	a 3	Y	õ	0	к	Р	G	к	А	Р	N	ь	L	I	0	W	А	s	т	L	н	s	G	v	Ρ	г	R	F	R	G	s	G	s	G	т	Е	F	т
L11	AV	N 3	Y	õ	õ	ĸ	P	G	к	A	P	к	L	L	I	Ŷ	A	A	s	s	L	0	S	G	v	P	s	R	F	s	G	s	G	s	G	т	D	F	т
				~	~																	~					-												
																. <b></b>	←				CD	R3					÷					-FR	4					÷	
L1	L '	Г	I	т	R	v	E	Ρ	Е	D	F	A	Ы	Y	Y	С	0	v	Y	G	А	S	s	_	Y	т	F	G	0	G	т	к	L	E	I	к	A	м	
L2	L	г	т	s	s	L	E	Р	Е	D	F	A	$\mathbf{v}$	Y	Y	С	0	0	Y	N	D	W	s	P	R	т	F	G	õ	G	т	к	v	Е	I	к	А	м	
L6	L	г	г	s	s	L	0	P	Е	D	F	A	T	Y	Y	c	õ	õ	T	Y	s	т		P	v	т	F	G	õ	G	т	R	L	Е	I	к	A	м	
г.9	L	г	тГ	N	G	L	õ	P	Е	D	F	A	ī	Y	Y	c	õ	õ	s	Y	s	т	L	P	F	т	F	G	P	G	Т	ĸ	v	E	I	ĸ	A	м	
T.10	т	- г	- <b>L</b>	s	s	т.	ñ	P	Б	D	s	A	ᆔ	Y	- Y	c	õ	õ	_	Y	0	т	v	8	v	- т	- ਸ	G	0	G	- т	к	T.	E	т	к	Α	м	
T.11	T. 1	- г	т.	т	s	т.	õ	P	Е	D	<u>न</u>	A	ᆔ	Y	Y	c	õ	õ	s	Y	S	т	т.	Р	Y	- т	- न	G	õ	G	- т	R	T.	E	т	ĸ	Α	м	
	_	-	-	-	-	-	×	-	_	2	-		-1	-	-	0	×	×	1	-	-	-		-	-	-	-	•	×	-	-		_	_	-				

Figure 16. Sequences of recovered gp41 specific VH (A) and VL (B) clones. Regions of similarity are indicated with grey boxes. As expected, diversity is most prevalent in the three CDR regions.



Figure 17. ELISA titrations of the gp41 mAbs against gp41, gp120, and BSA. OD values suggest all four mAbs demonstrate little reactivity with both negative controls BSA and gp120.

using DTT. The conformational b12 antibody and recombinant gp120 was included as a control. DTT treatment eliminated b12 binding to reduced gp120; however, all four gp41 mAbs maintained some level of binding to reduced gp41 (Figure 18B). H3/L2 and H4/L9 were affected the most by DTT treatment as binding was reduced by approximately 30% and 50% respectively (Figure 18B). Heat treatment alsocompletely eliminated b12's affinity for gp120; however, as was seen with DTT treatment, all four gp41 mAbs maintained affinity for heat-denatured gp41. Again, H3/L2 and H4/L9 were the most affected by heat denaturation as binding was reduced by approximately 45% and 55% respectively (Figure 18C). The differences in gp41 binding suggest that all four mAbs bind to predominantly linear epitopes. However, the reduced binding of H3/L2 and H4/L9 suggest that at least part of the epitope recognized by these antibodies has a conformational component.

# 3.3.3 Binding of MAbs to HIV-1 Envelope Spike

While each mAb was shown to bind to recombinant, monomeric gp41, the binding ability of each against gp41 in its native conformation as the transmembrane portion of the viral spike is more important in identifying any *in vivo* effector functions the mAbs may possess. Figure 19 A shows the flow cytometry results when HEK-293F cells expressing either SVK18 or HIV-1 Env from clade B or C were stained with each of the four novel gp41 mAbs and detected using R-PE labelled anti-human IgG Fc. Examining the median fluorescence value for each population indicates that three of the four mAbs bind to cell surface HIV-1 envelope clade B and clade C (Figure 19B). Only H6/L9 showed low reactivity against the Env



Figure 18. ELISA titrations of the 4 gp41 mAbs against intact gp41 (A), gp41 reduced using DTT (B), and heat denatured gp41 (C). B12/4L was included as a control due to the conformational epitope it recognizes on gp120. OD values for the gp41 mAbs against the denatured/reduced gp41 compared with those of B12, suggest each novel mAb recognizes a predominantly linear epitope on gp41.



Figure 19. Binding of gp41 mAbs to HIV-1 envelope expressed on the surface of 293F cells. (A) Flow cytometry histograms showing binding profile the four novel gp41 mAbs to HIV-1 gp160 expressed on the cell surface. Cells were transfected with an irrelevant light chain expression vector SVK18, HIV-1 clade B consensus Env expression vector, or HIV-1 clade C consensus Env expression vector. Cells were stained with 0.1 and 10  $\mu$ g/ml of the four gp41 mAbs, and the anti-PA antibody G75 as a negative control. (B) The median log fluorescence of each antibody tested is indicated for each for the different cell populations.

from clade B or C (Figure 19). H3/L2, H9/L2, and H6/L9 all showed reactivity against cell surface expressed Env, with H9/L2 showing the strongest affinity (Figure 19 A,B). All three mAbs showed similar reactivity against consensus clade B and clade C indicating that the epitopes recognized by these mAbs are conserved among both clades. To confirm these results, purified mAbs were titrated against the same transfectants, in addition to recombinant gp41 coated beads, and analyzed using the ABI cellular detection system. Examining each titration curve indicated that high concentrations of primary antibody caused a "hook-effect", as identified in Corbett et al (2007) where too much antibody saturates the system and inhibits antibody binding (Figure 20) [172]. The binding of all four gp41mAbs to surface Env was considerably less than mAb b12 (Figure 20). All four mAbs had lower endpoint titers against gp41 coated beads when compared with the Env transfectants (Figure 20). Mabs H4/L9 and H9/L2 showed the highest affinity for surface Env with an endpoint titer of 0.5 µg/ml. Confirming the flow cytometry results, clone H6/L9 was not positive against surface Env. Also consistent with results observed in Figure 19, all clones showed similar reactivity for surface Env from both clade C and clade B (Figure 20).

# 3.3.4 Molecular Characterization of gp41 Specific MAbs

Molecular analysis was carried out on the VH and VL genes comprising the novel mAbs specific for gp41 using IMGT software (Table 1). All sequences were highly mutated from their germline predecessors (Table 1). The VH genes all differed in their D(H) gene usage; however, all sequences did contain the same J



Figure 20. Titration of MAbs vs. HIV-1 envelope spike (clade B and C) using FMAT technology. HEK 293F cells transfected with clade B or clade C HIV-1 envelope expression vectors, or SVK18, were incubated with varying dilutions of the four gp41 specific mAbs, IgG1B12, or G75 and labeled using FMAT blue labeled goat anti-human IgG. Six-micron polystyrene beads coated with recombinant gp41 or BSA are included in place of transfected cells for comparison purposes. Data is expressed as the number of fluorescent events (counts) per well, with a positive result being defined as a count greater than 50 events per well. Results are the mean value of three independent experiments.

gene (J4\*02). Analysis of somatic mutations in the FR and CDR regions of the VH sequences revealed that, as is the case with affinity-matured antibodies, H4 and H9 have replacement (R) to silent (S) nucleotide change ratios (R:S) that were higher in the CDRs when compared with the FRs (Table 1). H4 and H9 also had a higher percentage of absolute mutations occur in the CDRs (42%, 38%) when compared with H3 and H6 (17%, 18%). The  $P_M$  value for H4 (0.0006) and H9 (0.009) suggests that there is an extremely low probability that the high R:S ratio observed in the CDRs of these sequences occurred by chance through random mutation of the original germline gene (Table 1). This suggests that H4 and H9 have been antigen selected through multiple rounds of affinity maturation resulting in an accumulation of R mutations in the CDRs and S mutations in the FRs.

Analysis of the VL genes indicates that all sequences have diverged considerably from their germline predecessors with L1 and L9 showing the lowest homology (86% and 88%) (Table 1). Examining the  $P_M$  value suggests that only clones L1, L10 and L11 have been antigen-selected. Both clones L2 and L9 have low numbers of mutations occurring in the CDR, with approximately 18% and 7% of the total number of mutations occurring in these regions.

					Muta	tions	F	R:S	
		Identity				CDR		CDR	
Clone	Germline gene	(%)	DH	ЈН	FR	1,2	FR	1,2	$P_M$
Н3	IGHV3-11*01	92	IGHD2-2*01	IGHJ4*02	19	4	2.8	1.0	0.729
H4	IGHV3-23*01	93	IGHD3-16*01	IGHJ4*02	12	9	0.8	8.0	0.0006
H6	IGHV4-4*07	92	IGHD6-13*01	IGHJ4*02	18	4	3.5	1.0	0.457
Н9	IGHV3-23*03	93	IGHD6-19*01	IGHJ4*02	13	8	1.6	7.0	0.009

# (A) Characteristics of Recovered gp41 Specific VH Genes

(B) Characteristics of Recovered gp41 Specific VL Genes												
				Muta	R							
		Identity			CDR		CDR					
Clone	Germline gene	(%)	JL	FR	1,2	FR	1,2	$P_M$				
L1	IGKV3-20*01	86	IGKJ2*01	25	9	2.1	8.0	0.004				
L2	IGKV3-11*01	92	IGKJ1*01	14	3	1.8	2.0	0.288				
L6	IGKV1-9*01	94	IGKJ5*01	11	2	1.8	x	0.175				
L9	IGKV1-NL1*01	88	IGKJ3*01	28	2	2.1	$\infty$	0.544				
L10	IGKV1-5*03	91	IGKJ2*01	16	5	7.0	$\infty$	0.018				
L11	IGKV1-39*01	91	IGKJ5*01	17	7	1.0	6.0	0.010				

Table 1. Gene characteristics of gp41 specific VH (A) and VL (B) sequences. Each rearranged VH or VL chain is listed with its most homologous germline gene, identity of each JH and DH segment, number of total mutations occurring in the framework (FR) regions, and in the complementary determining (CDR1,2) regions, and the replacement (R) to silent (S) nucleotide mutation ratio for each region. A higher R:S value in the CDR regions is indicative of antigenic selection. The  $P_M$  value represents the probability, calculated using a multinomial model [142] that the observed excess of R mutations in the CDRs and S mutations in the FRs occurred by chance through random mutation. A  $P_M$  value of < 0.05 suggests that the V sequence has been subject to antigen driven affinity maturation.
#### 4 Discussion

#### 4.1 Introduction

Human monoclonal antibodies against HIV-1 envelope proteins continue to play an important role in HIV-1 research. The structural characterization of gp120 and gp41, the identification of epitopes conserved among different HIV-1 subtypes, and the discovery of neutralizing epitopes on Env can all be attributed to the use of gp120 and gp41 specific monoclonal antibodies. Furthermore, bNt mAbs will likely play a major role in any topical microbicide yet to be developed, as only the passive transfer of gp120 and gp41 specific mAbs has proven capable of providing complete protection against SHIV mucosal viral challenge in macaques [48-52, 89-91, 144].

Efforts to produce novel bNt mAbs against HIV-1 have been hampered by inherent methodological limitations that have prevented the isolation of unique antibodies against gp120 or gp41. These limitations include: a considerable bias towards antibodies generated against clade B virus; a bias towards antibodies selected for biophysical properties advantageous for expression in prokaryotic systems; a lack of antibodies representing the repertoire of B cells at the site of initial contact with the virus; and a bias towards antibodies isolated from HIV positive individuals that, by definition, have been unsuccessful in preventing HIV-1 infection.

This paper describes the generation and characterization of 4 novel mAbs against the HIV-1 envelope protein gp41 using methods designed to limit the biases prevalent in previous production methods. A potential novel source of bNt mAbs

against HIV-1, the cervical B cell repertoire from HEPS sex workers in Nairobi, Kenya was used as the source for the antibody library. The characterization and gene analysis of these four IgA gp41 specific mAbs confirms that there are HEPS commercial sex workers in Nairobi that have cervical IgA specific for the HIV-1 envelope protein gp41 despite repeatedly testing seronegative.

#### 4.2 Advantages of the Cell Display System.

Initially, an antibody secretion system was developed from the IgG antibody repertoire of the PBMCs of a HIV positive donor with a high serum titer to gp120 and gp41. Using the serum IgG repertoire from this individual eliminated the question of whether the isolated B cells contained clones specific for HIV-1 Env. This strategy was based on the secretion of recombinant antibody by transiently transfected HEK 293F cells. Cells were re-suspended in semi-solid media with antibiotic selection, clones selected, expanded, and screened against gp120 or gp41.

Our hypothesis was that this system would be an efficient and effective method of isolating monoclonal antibodies of desired specificity from an immune antibody library. In practice; however, this system proved to be inefficient, time consuming, and rife with limitations. Consequently, it was concluded that this system was not the optimal choice for the far more challenging task of isolating Env specific mAbs from the CMC IgA repertoire of HIV-1 resistant individuals. The ineffectiveness of the antibody secretion system can be traced to four major problems: First, because the phenotype or binding specificity of the antibody was not physically linked to its' genotype, individual cells could not be screened for

binding. Cells had to be grown for over 2 weeks for clonal expansion to have occurred to the point where the clone was visible and could be isolated. A second problem was that because both vectors contained the same resistance gene (zeocin), clones were initially selected for the presence of either the heavy or light chain expression vector, not both. Ideally, the initial selection of clones would have included only those that contained both vectors and secreted a functional antibody.

This leads to the third major problem with this system: screening the sub clones was both time consuming and not high-throughput. Once a clone was chosen from the semi-solid media and transferred into a 96 well plate, the clone had to be cultured another 2 to 3 weeks prior to screening. The total number of clones screened was limited by practical limitations of cell culture, and when dealing with a random combinatorial library, a high number of clones must be screened to isolate an antibody with the desired specificity. The final problem with this strategy is that due to the transient nature of the transfection, the expression of antibody at the point when the clones were ready to be screened was very low. Clones were screened well over a month after the initial transfection. A month is well beyond the culture time recommended for transiently transfected cells, as cells begin to loose the plasmids even in the presence of a selective agent. Consequently, the amount of recombinant antibody present in the culture supernatant at the time of screening was not sufficient to give a positive signal on the ELISA.

To isolate mAbs from the CMCs of HIV-1 resistant donors, a new strategy was developed that would eliminate the problems posed by the antibody secretion system. This system permitted antibody expression directly on the cell surface of

transfectants, providing a physical link of the antibodies phenotype to its genotype, allowing single cells to be individually screened for antigen binding only 48 hours post transfection. To mediate surface expression, the human PDGFR-TM region was cloned onto the C terminal end of IgG1-CH3 encoded in the secretory vector PdblgG1, creating P-dblgG1-TM. Flow cytometry analysis of HEK 293F cells cotransfected with P-dbIgG1-TM and P-dbKappa indicated high expression levels of IgG1 on the cell surface. Significantly, analysis also demonstrated the presence of IGKC on the cell surface. Due to the fact that P-dbKappa does not encode for the TM region, the presence of IGKC on the cell surface indicated that a functional IgG1/kappa antibody was expressed on the cell membrane, and orientated away from the cell. As expected, surface expression of IgG Fc and IGKC was higher in the b12 transfectants than the library transfectants. As mentioned above, IGKC surface expression requires the VH region to associate with a complementary VL region to form a functional antibody. Because only a portion of the VH and VL combinations generated by the antibody library will be productive, surface expression of IGKC is lower in the library transfectants when compared with cells expressing native heavy and light chain pairs. Flow cytometry analysis of IgG1/Kappa b12 transfectants confirmed that the surface expressed antibodies maintained antigen affinity.

The ability to screen individual cells for antigen binding permitted the use of FACS to separate the positive from negative clones. This method is very highthroughput, as 10 million cells were screened in minutes for both antibody expression and antigen specificity. VH and VL genes were recovered from a 50 cell aliquot of positive clones, inserted into excretory expression vectors, and 96 clones

sequenced. Only 16 different heavy chains were identified in the pool of 96 sequences. This indicated that the sorting process substantially decreased the complexity of the library, as an experiment carried out to prior to sorting identified 13 distinct sequences in a total of 20 productive re-arrangements. Contrary to what was anticipated, the frequency of the observed heavy and light chain sequences recovered from the 50 cell aliquot did not appear to correlate with gp41 specificity. This is evident by the fact that sequence H1, making up 40% of the recovered sequences, did not show reactivity with gp41 when paired with any of the recovered light chains. It is possible that the native VL sequence for H1 was not recovered from the 50 cell pool.

Ideally, B cells recovered from the cervix of the HIV-1 resistant donors would have been sorted for affinity to gp41 prior to the construction of the library. This would have increased the chances of generating a high affinity clone by eliminating irrelevant VH and VL sequences, and increasing the likelihood of observing the natural heavy and light chain pairs. This method has been used to generate antibodies either form single sorted B cells [117-119], or from a surface displayed antibody library generated from pre-sorted B cells [112,116]. Due to technical and geographical limitations, isolated CMCs could not be maintained in the live state necessary for sorting; consequently, the library was generated from the total CMC population. The fact that 4 novel gp41 specific mAbs could be isolated from nonsorted B cells speaks to both the feasibility of the methods used, and possibly, to the prevalence of IgA antibodies specific for gp41 in the cervix of these HEPS women.

Due to the nature of HIV infection, it was important that the antibody library from the HEPS donors be generated from the repertoire of the current antibody forming cell response (AFC) against HIV-1 [146]. Consequently, the antibody library was generated from the mRNA of B cells found at the site of initial contact with the virus. The fact that activated plasma cells have 1000 times more mRNA than resting memory B cells [146] suggests that it is likely this antibody library is representative of the AFC in the cervical mucosa within these Kenyan HEPS individuals.

#### 4.3 Nature of the gp41 Epitope Recognized by the Four Mabs

Each of the four gp41 mAbs isolated from the CMCs of two Kenyan HEPS sex workers maintained affinity for heat-denatured gp41, as well as DDT reduced gp41. This suggests that each antibody binds to a predominantly linear epitope present on gp41. Two of the clones, H3/L2, and H4/L9, had their relative antigen affinity affinity via ELISA drop by approximately 50% against denatured/reduced gp41, suggesting that these two clones bind to epitopes with a partial conformational determinant. These results are in contradiction to several studies suggesting that the majority of antibodies raised against gp41 recognize conformational determinants. In a comprehensive study of serum IgG from six HIV infected individuals, Scheid et al (2009) found that the majority of gp41 antibodies isolated bound to conformational epitopes [132]. Furthermore, of the 25 gp41 specific Fabs isolated from HIV-1 infected donor by Binley et al (1996), not one bound to a linear epitope on gp41 [134]. The predominance of conformational antibodies in the serum of HIV-1 infected individuals was confirmed by serological studies, as titers

against reduced/denatured gp41 were on average 1 log lower than titers against the intact protein [134].

While conformational antibodies make up the majority of the antibody response against gp41, the only bNt mAbs isolated against gp41 bind to linear epitopes. Clones 2F5, 4E10, and Z13 are broadly neutralizing antibodies that bind to linear epitopes present on the MPER of gp41 [41,61,138]. The nature of the gp41 epitope optimal for HIV-1 neutralization has not been definitively established. In the case of gp120, the rigidification of the trimer prevents antibodies to linear epitopes from neutralizing the virus due the lack of a "correct" fine specificity for the functional spike [149]. This lowers the ability of antibodies raised against monomeric forms of gp120 common in viral debris, from reacting with the trimer to neutralize the virus [149]. Indeed, the majority of the anti-HIV serum antibody response is elicited by monomeric gp120 and unprocessed gp160 found in viral debris [27,151]. However; In the case of gp41, the occurrence of bNt mAbs that recognize linear epitopes on gp41 suggests that linear antibodies may play an important role in the response against HIV-1.

The ability of antibodies to neutralize HIV-1 has been correlated with their ability to bind with the native envelope spike [72, 152, 153, 155]. As a result antibody binding to the native, functional HIV envelope spike appears to be a prerequisite for antibody mediated virus neutralization [150, 154, 156]. In this regard, clones H3/L2, H4/L9, and H9/L2 were shown to bind to two clades of Env on the surface of HEK 293F cells using flow cytometry and confirmed by using the ABI cellular detection system.

Simply binding to an Env species on the virus; however, does not in itself accurately predict the neutralization potential of a given antibody. This is because several non-neutralizing Env specific antibodies have been shown to capture infectious HIV-1 virions in a highly specific manner [25, 26, 157-160]. Moore et al (2006) showed that these non-neutralizing antibodies were binding with alternative non-functional forms of Env present on the infectious virion [156]. This study showed that both gp41 timers from which gp120 has been shed, and gp120/gp41 monomers decorate the surface of viral particles [156]. The presence of uncapped gp41 trimers on the surface of HIV-1 virions has been observed previously [23, 27]. The gp120/gp41 monomer is likely formed during Env synthesis prior to oligomerization, while gp120 is most likely shed from the gp41 stump following virion maturation [156]. Furthermore, uncleaved gp160 precursor molecules have been shown to make up a substantial portion of the surface Env expressed by HIV infected cells [161]. This non-functional Env spike can be recognized by both neutralizing and non-neutralizing antibodies specific for gp120 and gp41 [161].

Given the considerable heterogeneity in the surface expressed Env of HIV-1, it is not possible to conclusively say that mAbs H3/L2, H4/L9, and H9/L2 bind with the functional Env spike from the data presented. Future work includes conducting neutralization studies to determine if these gp41 mAbs bind with functional Env and mediate virus neutralization. Without the neutralization data; however, we can say that these 3 mAbs will likely have the ability to bind directly to infectious virus by binding to functional or non-functional Env species on the viral membrane. Even if the mAbs bind to non-functional Env species, antibody coating of non-functional Env

can have consequences distinct from antibody mediated neutralization [156]. Nonneutralizing antibodies have been shown to induce Fc receptor-mediated phacocytosis and inhibit HIV-1 infection of macrophages [162]. In addition, antibody coating in the absence of neutralization may explain the high concentrations of immune complexes in HIV positive individuals with low serum neutralization titers [163,164]. Virus coating with non-neutralizing antibodies may encourage trapping of these complexes on the surface of follicular dendritic cells by prolonging their persistence [165, 166]. Furthermore, because these mAbs were generated from the IgA repertoire of the cervical mucosa, the ability to bind directly to the virus may have an increased correlation with protection attributable to the ability of mucosal IgA to trap virus in the mucosa by virion cross-linking and immune exclusion [78, 80].

### 4.4 Molecular Analysis of the gp41 Specific MAbs

Antigen specificity is largely encoded by the 6 CDRs, 3 on the VH and 3 on the VL. X-Ray diffraction studies have shown that the CDR3 of the VH region is integral to antigen binding in most mAbs due to its position at the center of the antigen binding site [147]. The role played by the light chain in the antigen specificity of an antibody is variable. Some antibodies require a specific VL region to maintain both functionality and antigen specificity, while antibodies in which the VH region is largely responsible for antigen binding can incorporate many different light chains [125,134]. While examining the HIV-1 mAbs isolated from phage display libraries, both Burton et al (1991) and Binley et al (1996) observed that the same heavy chain

could be paired with different light chains and maintain antigen specificity [125,134]. This light chain promiscuity suggests that the heavy chain plays a dominant role in antigen binding to gp120 and gp41. Accordingly, chain swapping experiments conducted with the heavy and light chains isolated from the present antibody library also displayed light chain promiscuity, as all four gp41 specific heavy chains could be paired with at least 3 and as many as 5 different light chains and maintain affinity to gp41. This finding is also in agreement with the previously described role of the heavy and light chain for the gp41 clones 69 and 77 described by Tudor et al (2009), and for the bNt mAb 2F5 [104,148].

Extensive antigen-driven maturation is essential for the development of neutralizing antibodies against HIV-1. The majority of HIV-1 bNt mAbs have a high frequency of somatic mutation associated with a marked increase in replacement mutations indicating antigen-driven affinity maturation has occurred [168]. Accordingly, the VH and VL of all four mAbs generated herein were shown to have extensive somatic mutations when compared with the most homologous germline sequence. Consistent with antigen-driven affinity maturation, clones, H4/L9 and H9/L2 have high replacement to silent nucleotide exchange ratios in the CDRs compared with the FRs. Furthermore, when the multinomial model developed by Losos et al (2000) to predict the probability of antigenic selection was applied to the these clones, both had P<sub>m</sub> values that suggested there was an extremely low probability that the observed excess of R mutations in the CDRs and S mutations in the FRs occurred by random mutation [142]. While this model does not take into account intrinsic biases caused by mutation hotspots in the VH and VL, it does

provide a good approximation of the degree of antigen-driven affinity maturation experienced by a given antibody gene [142, 169]. Alternatively, these individuals may be polymorphic in the VH locus and their V genes may be inherently different at the DNA level in the germline [105]. To eliminate this possiblility, VH sequences need to be examined from chromosomal DNA.

It has been suggested that HIV has evolved a protective mechanism that shields its most vulnerable epitopes, including the CD4 binding site and the MPER, by exploiting "holes" in the human germline antibody repertoire [167]. This was proposed by Xia et al (2009) who showed that germline predecessors of bNt mAbs b12, 2G12 and 2F5 lacked measurable binding to Env, while the predecessors of weakly neutralizing mAbs showed high affinity to Env [167]. Consequently, the predecessors of bNt mAbs may be unable to bind strongly enough to Env during the initial stages of infection to initiate their development via antigen-driven affinity maturation [167]. This suggests that the continued B cell exposure to viral antigens that occurs during a chronic HIV-1 infection is an important factor in the development of a broadly neutralizing antibody response [167]. Discouragingly, by the time the antibody response has undergone sufficient antigen-driven affinity maturation to target these vulnerable epitopes, the infection is far too advanced for the antibodies to have any significant effect. However, if HIV seronegative individuals are subject to repeated exposed to HIV antigens at mucosal sites, it could result in the generation of affinity-matured HIV-1 specific mucosal IgA antibodies at the site of initial contact with the virus [104]. The presence of affinity-matured IgA at mucosal sites may provide the best chance at preventing initial infection, as once

the virus has travelled through the mucosal epithelium, the chances of eliminating the infection are remote.

The study population where these antibodies have the best chance at being found are the HEPS individuals. These individuals are repeatedly exposed to viral antigens at mucosal sites through unprotected intercourse with HIV positive individuals (estimated at 64 viral encounter per year in the Nairobi cohort [73]). Accordingly, Tudor et al (2009) isolated several IgA antibodies from the cervix of HEPS sex workers from Cambodia, specific for gp41, that show extensive somatic mutations and appear to have undergone antigen-driven affinity maturation [104]. These mAbs were also able to block epithelial cell transcytosis and neutralize CD4+ T cell infection, suggesting that they are specific for susceptible epitopes on gp41 [104]. As mentioned above, the gp41 mAbs generated, herein, from the cervical IgA repertoire of HEPS women in Nairobi appear to have been somatically mutated, with two of the clones showing strong evidence of being subject to extensive antigen-driven affinity maturation. Whether or not these mAbs bind with susceptible epitopes on gp41 and play a role in HIV-1 resistance is yet to be determined.

#### 4.5 Path Forward

Attempts to produce all four gp41 mAbs on a large scale have so far been unsuccessful. This is most likely due to inefficiencies in the vector expression system, as it was not designed for large scale transfections, and problems with antibody purification. However, efforts to modify the expression/purification

system to allow for large scale expression of the 4 mAbs continues. Testing all 4 mAbs in standard HIV-1 CD4+ T cell neutralization assays is of the highest priority. Furthermore, because the mAbs were isolated form the IgA repertoire of cervical B cells, transcytosis assays also need to be performed. Once any *in vivo* effector functions have been identified for the 4 mAbs, epitope mapping should be carried out. Due to the linear nature of the epitopes recognized by the four mAbs, an ELISA against a set of gp41 peptides should elucidate the gp41 epitope recognized by each. Furthermore, competition ELISAs with known gp41 mAbs such as 2F5, and 4E10 are underway. Identifying the epitopes important in eliciting protective IgA affinitymatured antibodies is critical to the development of a mucosal vaccine capable of eliciting a local antibody response that prevents HIV-1 infection at the point of entry.

Attempts to generate microbicides with the goal of preventing HIV-1 transmission have so far been unsuccessful. Efforts to generate the next generation of microbicides based on antiretroviral agents is underway. This approach, however, is likely to become harmful to women over the long term due to antiretroviral drug induced changes in vaginal and virological parameters that could lead to increased rates of viral infection [104, 170]. Natural agents, such as affinitymatured HIV-1 neutralizing IgA antibodies, may be among the safest and most effective components of an effective microbicide. These antibodies, provided they possess potent effector functions, would prevent free virus from invading the mucosa, and prevent viral release from infected cells [104]. Indeed, vaginal administration of the bNt mAb b12 has prevented vaginal transmission of SHIV in

macaques [171]. The characterization of existing affinity matured Env specific IgA mAbs, and the discovery of additional clones is necessary for the continued progress of antibody based HIV-1 microbicides.

#### **4.6 Conclusion**

A mammalian cell display system has been developed for the generation of fully human recombinant monoclonal antibodies. Using this system, four novel gp41 monoclonal antibodies were isolated from the IgA repertoire from cervical B cells obtained from two HEPS sex workers from Nairobi, Kenya. All four mAbs bind to predominantly linear epitopes on gp41, and three of the clones demonstrated affinity for HIV-1 Env clade B and clade C expressed on the surface of transfected cells. The two clones showing the highest affinity for Env also appear to have undergone extensive antigen-driven affinity maturation. The presence of these gp41 specific antibodies in the cervix of HEPS sex workers from Nairobi confirms there are HEPS commercial sex workers in Nairobi that have cervical IgA specific for the HIV-1 envelope protein gp41 despite repeatedly testing seronegative. Further characterization of the effector functions of these mAbs is required to determine their role in HIV-1 resistance.

## **5** References

- 1 UNAIDS. 2008. Report on the global AIDS epidemic. Executive Summary. www. unaids.org
- 2 Gottlieb MS, Schroff R, Schanker HM, Weisman JD, Fan PT, Wolf RA, and A Saxon. *Pneumocystis carinii* pneumonia and mucosal candidiasis in previously healthy homosexual men: evidence of a new acquired cellular immunodeficiency. *N Engl J Med.* 1981. 305: 1425-1431.
- Barre-Sinoussi F, Chermann JC, Rey F, Nugeyre MT, Chamaret S, Gruest J,
  Dauguet C, Axler-Blin C, Vezinet-Brun F, Rouzioux C, Rozenbaum W, and L
  Montagnier. Isolation of a T-lymphocyte retrocirus from a patient at risk for
  acquired immune deficiency syndrome (AIDS). *Science.* 1984. 224: 868-871.
- 4 Popovic M, Sarngadharan MG, Read E, and RC Gallo. Detection, isolation, and continuous production of cytopathic retroviruses (HTLV-III) from patients with AIDS and pre-AIDS. *Science*. 1984. 224: 497-500.
- 5 Wang WE, Chen MY, Chuang CY, Jeang KT, and LM Huang. Molecular biology of human immunodeficiency virus type 1. *J Microbiol Immunol Infect*. 2000. 33: 131-140.
- 6 Rambaut A, Posada D, Crandall KA, and EC Holmes. The causes and consequences of HIV evolution. *Nat Rev Gen*. 2004. 52: 52-61.
- 7 Rees PH. HIV/AIDS: the first 25 years a view from Nairobi. *East Afr Med J*. 2008. 85: 292-300.
- 8 McKnight A, and Marlen MI Aasa-Chapman. Clade specific neutralising vaccines for HIV: An appropriate target? *Curr HIV Res.* 2007. 5: 554-560.
- 9 Klatzmann D, Barre Sinoussi F, Nugeyre MT, Danquet C, Vilmer E, Griscelli C, Brun-Veziret F, Rouzioux C, Gluckman, JC, and JC Chermann. Selective tropism of lymphadenopathy associated virus (LAV) for helper-induced T lymphocytes. *Science*. 1984. 225: 59-63.
- 10 Klatzmann D, Champagne E, Chamaret S, Gruest J, Guetard D, Hercend T, Gluckman JC, and L Montagnier. T-lymphocyte T4 molecule behaves as the receptor for human retrovirus LAV. *Nature*. 1984. 312: 767-768.
- 11 Maddon PJ, Dalgeish AG, McDougal JS, Clapham PR, Weiss RA, and R Axel. The T4 gene encodes the AIDS virus receptor and is expressed in the immune system and the brain. *Cell.* 1986. 47: 333-348.

- 12 Fent, Y, Broder CC, Kennedy PE, and EA Berger. HIV-1 entry cofactor: functional cDNA cloning of a seven-transmembrane, G protein-coupled receptor. *Science*. 1996. 272: 872-877.
- 13 Alkhatib G, Combadiere C, Broder CC, Feng Y, Kennedy PE, Murphy PM, and EA Berger. CC CKR5: a RANTES, MIP-1 alpha, MIP-1 beta receptor as a fusion cofactor for macrophage-tropic HIV-1. *Science*. 1996. 272: 1955-1958.
- 14 Knight SC, and SE Macatonia. Dendritic cells and viruses. *Immunol Lett.* 1988. 19: 177-181.
- 15 Weiss RA. Special Anniversary Review: Twenty-five years of human immunodeficiency virus research: successes and challenges. *Clin Exp Immunol.* 2008. 152: 201-210.
- 16 Patterson S, Rae A, Hockey N, Gilmour J, and F Gotch. Plasmacytoid dendritic cells are highly susceptible to human immunodeficiency virus type 1 infection and release infectious virus. *J Virol*. 2001. 75: 6710-6713.
- 17 Stebbing J, Gazzard B, and DC Douek. Where does HIV live? *N Engl J Med*. 2004. 350: 1872-1880.
- 18 Mellors JW, Munoz A, Giorgi JV, Giorgi JV, Margolick JB, Tassoni CJ, Gupta P, Kingsley LA, Todd JA, Saah AJ, Detels R, Phair JP, and CR Rinaldo Jr. Plasma viral load and CD4+ lymphocytes as prognostic markers of HIV-1 infection. *Ann Intern Med*. 1997. 126: 946-954.
- 19 Aasa-Chapmann MM, Holouigue S, Aubin K, Wong ML, Jones NA, Cornforth D, Pellegrino P, Newton P, Williams I, Borrow P, and A McKnight. Detection of antibody-dependent complement-mediated inactivation of both autologous and heterologous virus in primary human immunodeficiency virus type 1 infection. *J Virol*. 2005. 79: 2823-2830.
- 20 Huber M, Fischer M, Misselwitz B, Manrique A, Kuster H, Niederost B, Weber R, von Wyl V, Huldrych F, Gunthard, and A Trkola. Complement lysis activity in autologous plasma is associated with lower viral loads during the acute phase of HIV-1 infection. *PLoS Med.* 2006. 3: e441.
- 21 Wu H, Zhu H, Miao H, and AS Perelson. Parameter identifiability and estimation of HIV/AIDS dynamic models. *Bull Math Biol*. 2008. 70: 785-799.
- 22 Hladik F, and JM McElrath. Setting the stage: host invasion by HIV. *Nat Rev Immunol.* 2008. 8: 447-458.
- 23 Pantophlet R, and DR Burton. Gp120: Target for neutralizing HIV-1 Antibodies. *Annu Rev Immunol.* 2006. 24: 739-769.

- 24 Roux KH, and KA Taylor. Aids virus envelope spike structure. *Curr Opin Struct Biol*. 2007. 17: 244-252.
- 25 Kuznetsov YG, Victoria JG, Robinson WE Jr, and A McPherson. Atomic force microscopy investigation of uman immunodeficiency virus (HIV) and HIVinfected lymphocytes. *J Virol*. 2003. 11896-11909.
- 26 Herrera C, Klasse PJ, Michael E, Kake S, Barnes K, Kibler CW, Campbell-Gardener L, Sodroski J, Moore JP, and S Beddows. The impact of envelope glycoprotein cleavage on the antigenicity, infectivity, and neutralization sensitivity of Env-pseudotyped human immunodeficiency virus type 1 particles. *Virology*. 2003. 338: 154-172.
- 27 Parren PWHI, Burton DR, and QJ Sattentau. HIV-1 antibody-debris or virion? *Nat Med.* 1997. 3: 366-367.
- 28 McKnight A, and Marlen MI Aasa-Chapman. Clade specific neutralising vaccines for HIV: An appropriate target? *Curr HIV Res.* 2007. 5: 554-560.
- 29 Spearman P. Current progress in the development of HIV vaccines. *Curr Pharm Des.* 2006. 12: 1147-1167.
- 30 Montero M, van Houten NE, Wang X, and JK Scott. The membrane-proximal external region of the human immunodeficiency virus type 1 envelope: dominant site of antibody neutralization and target for vaccine design. *Microbiol Mol Biol Rev.* 2008. 72: 54-84.
- 31 Kuiken C, Foley B, Hahn P, Marx F, McCutchan F, Mellor J, Wolinsky S, and B Kobert. HIV Sequence Compendium. Los Alamos national Laboratory, Los Alamos, NM.
- 32 Bosch ML, Earl PL, Fargonoli K, Picciafuoco S, Giombini F, Wong-Staal F, and G Franchini. Identification of the fusion peptide of primary immunodeficiency viruses. *Science*. 1989. 244: 694-697.
- 33 Freed EO, Delwart EL, Buchschacher Jr. GL, and AT Panganiban. A mutation in the human immunodeficiency virus type 1 transmembrane glycoprotein gp41 dominantly interferes with fusion and infectivity. *Proc Natl Acad Sci USA*.1992. 89: 70-74.
- 34 Schaal H, Klein M, Gehrmann P, Adams O, and A Scheid. Requirement of Nterminal amino acid residues of gp41 for human immunodeficiency virus type 1-mediated cell fusion. *J Virol.* 1995. 69: 3308-3314.

- 35 Brasseur R, Vandenbranden M, Cornet B, Burny A, and JM Ruysschaert. Orientation into the lipid bilayer of an asymmetric amphipathic helical peptide located at the N-terminus of viral fusion proteins. *Biochem Biophys Acta*. 1990. 1029: 267-273.
- 36 Melikyan GB, Markosyan RM, Hemmati H, Delmedico MK, Lambert DM, and FS Cohen. Evidence that the transition of HIV-1 gp41 into a six-helix bundle, not the bundle configuration, induces membrane fusion. *J Cell Biol*. 2000. 151: 413-423.
- 37 Gallaher WR, Ball JM, Garry RF, Griffin MC, and RC Montelaro. A general model for the transmembrane proteins of HIV and other retroviruses. *AIDS Res Hum Retrovir*. 1989. 5: 431-440.
- 38 Ofek, G, Tang M, Sambor A, Katinger H, Mascola JR, Wyatt R, and PD Kwong.Structure and mechanistic analysis of the anti-human immunodeficiency virus type 1 antibody 2F5 in complex with its gp41 epitope. *J Virol*. 2004. 78: 10724-10737.
- 39 Pai EF, Klein MH, Chong P and A Pedyczak. Fab'-epitope complex from the HIV-1 cross-neutralizing monoclonal antibody 2F5. 2000. U.S patent WO-00/61618.
- 40 Barbato G, Bianchi E, Ingallinella P, Hurni WH, Miller MD, Ciliberto G, Cortese R, Bazzo R, Shiver JW, and A Pessi. Structural analysis of the epitope of the anti-HIV antibody 2F5 sheds light into its mechanism of neutralization and HIV fusion. *J Mol Biol.* 2003. 330: 1101-1115.
- 41 Muster, T, Steindl F, purtscher M, Trkola A, Klima A Himmler G, Ruker F, and H Katinger. A conserved neutralizing epitope on gp41 of human immunodeficiency virus type 1. *J Virol*. 1993. 67: 6642-6647.
- 42. Bellamy-McIntyre AK, Lay CS, Bar S, Maerz AL, Talbo GH, Drummer HE, and P Poumbourios. Functional links between the fusion peptide-proximal polar segment and membrane-proximal region of human immunodeficiency virus type 1 gp41 in distinct phases of membrane fusion. *J Biol Chem*. 2007. 282: 23104-23116.
- 43 Nyholm TK, Ozdirekcan S, and JA Killian. How protein transmembrane segments sense the lipid environment. *Biochemistry*. 2007. 46: 1457-1465.
- 44 Matoba N, Magerus A, Geyer BC, Zhang Y, Muralidharan M, Alfsen A, Arntzen CJ, Bomsel M, and TS Mor. A mucosally targeted subunit vaccine candidate eliciting HIV-1 transcytosis-blocking abs. *Proc Natl Acad Sci USA.* 2004. 101: 13584-13589.

- 45 Qiao ZS, Kim M, Reinhold B, Montefiori D, Wang JH, and EL Reinherz. Design, expression, and immunogenicity of a soluble HIV trimeric envelope fragment adopting a prefusion gp41 configuration. *J Biol Chem*. 2005. 280: 23138-23146.
- 46 Gilbert PB, Peterson ML, Folllmann D, Hudgens MG, Francis DP, Gurwith M, Heyward WL, Jobes DV, Popovic V, Self SG, Sinangil F, Burke D and PW Berman.
- 47 Anonymous. HIV vaccine failure prompts Merck to halt trial. *Nature*. 2007. 449:390.
- 48 Shibata R, Igarashi T, Haigwood N, Buckler-White A, Ogert R, Ross W, Willey R, Cho MW, and MA Malcom. Neutralizing antibody directed against the HIV-1 envelope glycoprotein can completely block HIV-1/SIV chimeric virus infections of macaque monkeys. *Nat Med.* 1999. 5: 204-210.
- Mascola JR, Lewis GM, Stiegler G, Harris D, Vancott TC, Hayes D, Louder MK, Brown CR, Sapan CV, Frankel SS, Lu Y, Robb ML, Katinger H, and DL Birx.
   Protection of macaques against pathogenic simian/human immunodeficiency virus 89.6PD by passive transfer of neutralizing antibodies. *J Virol*. 1999. 73: 4009-4018.
- 50 Mascola JR, Stiegler G, Vancott TC, Katinger H, Carpenter CB, Hanson CE, Beary H, Hayes D, Frankel SS, Birx DL, and MG Lewis. Protection of macaques against vaginal transmission of a pathogenic HIV-1/SIV chimeric virus by passive infusion of neutralizing antibodies. *Nat Med*. 2000. 6: 207-210.
- 51 Parren WHI, Marx AP, Hessell AJ, Luckay A, Harouse J, Cheng-Mayer C, Moore JP, and DR Burton. Antibody protects macaques against vaginal challenge with a pathogenic R5 simian/human immunodeficiency virus at serum levels giving complete neutralization *in vitro*. *J Virol*. 2001. 75: 8340-8347.
- 52 Ferrantelli F, Rasmussen RA, Buckley KA, Li PL, Wang T, Montefiori DC, Katinger H, Stiegler G, Anderson DC, McClure HM, Ruprecht RM. Complete protection of neonatal macaques against oral challenge with pathogenic simian-human immunodeficiency virus by human anti-HIV monoclonal antibodies. *J Infect Dis.* 2004. 189: 2167-2173.
- 53 Ezzell C. Trials of vaccine against AIDS to begin in humans. *Nature*. 1987. 328: 747.
- 54 Redfield RR, Birx DL, Ketter N, Tramont E, Polonis V, Davis C, Brundage JF, Smith G, Johnson S, Fowler A, et al. A phase I evaluation of the safety and immunogenicity of vaccination with recombinant gp160 in patients with

early human immunodeficiency virus infection. Military Medical Consortium for Applied Retroviral Research. *N Engl J Med.* 1991. 324: 1677-1684.

- 55 Dolin R, Graham BS, Greenberg SB, Tacket CO, Belshe RB, Midthun K, Clements ML, Gorse GJ, Horgan Bw, Atmar RL, et al. The safety and immunogenicity of a human immunodeficiency virus type 1 (HIV-1) recombinant gp160 candidate vaccine in humans. NIAID AIDS Vaccine Clinical Trials Network. *Ann Intern Med.* 1991. 114: 119-127.
- 56 Belshe RB, Clements ML, Dolin R, Graham BS, McElrath J, Gorse GJ, Schwartz D, Keefer MC, Wright P, Corey L, Bolognesi DP, Mathews TJ, Stablein DM, O'Brien FS, Eible M, Dorner F, and W Koff. Safety and immunogenicity of a fully glycosylated recombinant gp160 human immunodeficiency virus type 1 vaccine in subjects at low risk of infection. *J Infect Dis.* 1993. 1387-1395.
- 57 Belshe RB, Graham BS, Keefer MC, Gorse GJ, Wright P, Dolin R, Matthews T, Weinhold K, Bolognesi DP, Sposto R, Stablein DM, Twaddell T, Berman PW, Gregory T, Izu AE, Walker MC, and P Fast. NIAID AIDS Vaccine Clinical Trials Network. *JAMA*. 1994. 272: 475-480.
- 58 Kahn JO, Sinangil F, Baenziger J, Murcar N, Wynne D, Coleman RI, Steimer KS, Dekker CL, and D Chernoff. Clinical and immunologic responses to human immunodeficiency virus (HIV) type 1SF2 gp120 subunit vaccine combined with MF59 adjuvant with or without muramyl tripeptide dipalmitoyl phosphatidylethanolamine in non-HIV-infected human volunteers. *J Infect Dis.* 1994. 170: 1288-1291.
- 59 Schwarz DH, gorse G, Clement ML, Belshe R, Izu A, Duliege AM, Berman P, Twaddell T, Stablein D, Sposto R, Siliciano R, and T Matthews. Induction of HIV-1 neutralising and syncytium-inhibiting antibodies to uninfected recipients of HIV-1IIIB rgp120 subunit vaccine. *Lancent*. 1993. 342: 69-73.
- 60 Trkola A, Pomales AB, Yuan H, Korber B, Maddon PJ Allaway GP, Katinger, H, Barbas 3<sup>rd</sup> CF, Burton DR, and DD Ho .Cross-clade neutralization of primary isolates of human immunodeficiency virus type1 by human monoclonal antibodies and tetrameric CD4-IgG. *J Virol.* 1995. 69: 6609-6617.
- 61 Zwick, MB, Labrijn AF, Wang M, Spenlehauer C, Saphire EO, Binley JM, Moore JP, Stiegler G, Katinger H, Burton DR, and PWHI Parren. Broadly neutralizing antibodies targeted to the membrane-proximal external region of human immunodeficiency virus type 1 glycoprotein gp41. *J Virol*. 2001. 75: 10892-10905.
- 62 Alam SM, Morelli M, Dennison SM, Liao HX, Zhang R, Xia SM, Rits-Volloch S, Sun L, Harrison SC, Haynes BC, and B Chen. Role of HIV membrane in

neutralization by two broadly neuralizing antibodies. *Proc Natl Acad Sci USA*. 2009. 106: 20234-20239.

- 63 Barbas 3<sup>rd</sup> CF, Bjorling E, Chiodi F, Dunlop N, Cababa D, Jones TM, Zebedee SL, Persson MA, Nara PL, and E Norrby . Recombinant human Fab fragments neutralize human type 1 immunodeficiency virus *in vitro*. *Proc Natl Acad Sci*. 1992. 89: 9339-9343.
- 64 Fouts T, Godfrey K, Bobb K, Montefiori D, Hanson CV, Kalyanaraman VS, Debico A, and R Pal. Crosslinked HIV-1 envelope-CD4 receptor complexes elicit broadly cross-reactive neutralizing antibodies in rhesus macaques. *Proc Natl Acad Sci.* 2002. 99: 11842-11847.
- 65 Fouts TR, Tuskan R, Godfrey K, Reitz M, Hone D, Lewis GK, and AL DeVico. Expression and characterization of a single-chain polypeptide analogue of the human immunodeficiency virus type 1 gp120-CD4 receptor complex. *J Virol.* 2000. 74: 11427-11436.
- 66 LaCasse Ra, Follis KE, Trahey M, Scarborough JD, Littman DR, and NH Nunberg. Fusion-complement vaccines: broad neutralization of primary isolates of HIV. *Science*. 1999. 283: 357-362.
- 67 Pantophlet R, Wilson IA, and DR Burton. Improved design of an antigen with enhanced specificity for the broadly HIV-neutralizing antibody b12. *Protein Eng Des Sel*. 2004. 17: 749-758.
- 68 Frey, G, Peng H, Rits-Volloch S, Morelli M, Cheng Y, and B Chen. A fusionintermediate state of HIV-1 gp41 targeted by broadly neutralizing antibodies. *Proc Natl Acad Sci.* 2008. 105: 3739-3744.
- Sanders RW, Vesanen M, Schuelke N, Master A, Schiffner L, Kalyanaraman R, Paluch M, Berkhout B, Maddon PJ, Olson WC, Lu M, and JP Moore.
   Stabilization of the soluble, cleaved, trimeric form of the envelope glycoprotein complex of human immunodeficiency virus type 1. *J Virol*. 2002. 76: 8875-8889.
- 70 Yang X, Lee J, Mahoney EM, Kwong PD, Wyatt R, and J Sodroski. Highly stable trimers formed by human immunodeficiency virus type1 envelope glycoproteins fused with the trimeric motif of T4 bacteriophage fibritin. *J Virol*. 2002. 76: 4634-4642.
- 71 McGaughey GB, Barbato G, Bianchi E, Freidinger RM, Garsky VM, Hurni WM, Joyce JG, Liang X, Miller MD, Pessi A, Shiver JW, and MJ Bogusky. Progress towards the development of a HIV-1 gp41 directed vaccine. *Curr HIV Res.* 2004. 2: 193-204.

- 72 Fouts TR, Binley JM, Trkola A, Robinson JE, and JP Moore. Neutralization of the human immunodeficiency virus type 1 primary isolate JR-FL by human monoclonal antibodies correlates with antibody binding to the oligomeric form of the envelope glycoprotein complex. *J Virol*. 1997. 71: 2779-2785.
- Fowke KR, Nagelkerke NJD, Kimani J, Simonsen JN, Anzala AO, Bwayo JJ,
  MacDonald KS, Ngugi EN, and FA Plummer. Resistance to HIV-1 infection
  among persistently seronegative prostitutes in Nairobi, Kenya. *Lancet.* 1996.
  348: 1347-1351.
- 74 Plummer FA, Ball TB, Kimani J, and KR Fowke. Resistance to HIV-1 infection among highly exposed sex workers in Nairobi: what mediates protection and why does it develop? *Immunol Lett.* 1999. 66: 27-34.
- 75 Kulkarni PS, Butera ST, and AC Duerr. Resistance to HIV-1 infection: Lessons learned from studies of highly exposed persistently seronegative (HEPS) Individuals. *Aids Rev.* 2003. 5: 87-103.
- 76 Rowland Jones S, Dong T, Fowke KR, Kimani J, Krausa P, Newell H, Blanchard T, Ariyoshi K, Oyugi J, Ngugi E, Bwayo J, MacDonald KS, McMichael AJ, and FA Plummer. Cytotoxic T cell responses to multiple conserved HIV epitopes in HIV-resistant prostitutes in Nairobi. *J Clin Invest*. 1998. 102: 1758-1765.
- 77 Fowke KR, Kaul R, Rosenthal KL, Oyugi J, Kimani J, Rutherford WJ, Nagelkerke NJ, Ball TB, Bwayo JJ, Simonsen JN, Shearer GM, and FA Plummer. HIV-1specific cellular immune responses among HIV-1-resistant sex workers. *Immunol Cell Biol*. 2000. 78: 586-595.
- 78 Kozlowski PA, and MR Neutra. The role of mucosal immunity in prevention of HIV transmission. *Curr Mol Med*. 2003. 3: 217-228.
- 79 Kozlowski PA, Williams SB, Lynch RL, Flanigan TP, Patterson RR, Cu-Uvin S, and MR Neutra. Differential induction of mucosal and systemic antibody responses in women after nasal, rectal, or vaginal immunization: influence of the menstral cycle. *J Immunol.* 2002. 169: 566-574.
- 80 Burns JW, Siadat-Pajouh M, Krishnaney AA, and HB Greenberg. Protective effect of rotavirus VP6-specific IgA monoclonal antibodies that lack neutralizing activity. *Science*. 272: 104-107.
- 81 Alexander R, and J Mestecky. Neutralizing Antibodies in Mucosal Secretions: IgG or IgA? *Curr HIV Res.* 2007. 5: 588-593.
- 82 Alfsen A, Iniguez P, Bouguyon E, and M Bomsel. Secretory IgA specific for a coserved epitope on gp41 envelope glycoprotein inhibits epithelial transcytosis of HIV-1. *J Immunol.* 2001. 166: 6257-6265.

- 83 Zolla-Pazner S. Identifying epitopes of HIV-1 that induce protective antibodies. *Nat Rev Immunol.* 2004. 4: 199-210.
- 84 Huang YT, Wright A, Gao X, Kulick L, Huimin Y, and ME Lamm. Intraepithelial cell neutralization of HIV-1 replication by IgA. *J Immunol*. 2005. 174: 4828-4835.
- 85 Connick E, Marr DG, Zhang XQ, Clark SJ, Saag MS, Schooley RT, and TJ Curiel. HIV-specific cellular and humoral immune responses in primary HIV infection. *Aids Res Hum Retroviruses*. 1996. 1129-1140.
- 86 Baum LL, Cassutt KJ, Knigge K, Khattri R, Margolick J, Rinaldo C, Kleeberger CA, Nishanian P, Henrard DR, and J Phair. HIV-1 gp120-specific antibodydpendent cell-mediated cytotoxicity correlates with rate of diseas progression. *J Immunol*. 1996. 2168-2173.
- 87 Duval M, Posner MR, and LA Cavacini. A bispecific antibody composed of a nonneutralizing antibody to the gp41 immunodominant region and an anti-CD89 antibody directs broad human immunodeficiency virus destruction by neutrophils. *J Virol*. 2008. 82: 4671-4674.
- 88 Trkola A, Purtscher M, Muster T, Ballaun C, Buchacher A, Sullivan N, Srinivasan K, Sodroski J, Moore JP, and H Katinger. Human monoclonal antibody 2G12 defines a distinctive neutralization epitope on the gp120 glycoprotein of human immunodeficiency virus type 1. *J Virol*. 1996. 70: 1100-1108.
- 89 Hessell AJ, Rakasz EG, Poignard P, Hangartner L, Landucci G, Forthal DN, Koff WJ, Watkins DI, and DR Burton. Broadly neutralizing human anti-HIV antibody 2G12 is effective in protection against mucosal SHIV challenge even at low serum neutralizing titers. *PLoS Pathog.* 2009. 5: e1000433.
- 90 Hessell AJ, Poignard P, Hunter M, Hangartner L, Tehrani DM, Bleeker WK, Parren PWHI, Marx PA, and DR Burton. Effective, low-titer antibody protection against low-dose repeated mucosal SHIV challenge in macaques. *Nat Med.* 2009. 15: 951-956.
- 91 Svehia LY, Louder MK, Wycuff D, Phogat S, Tang M, Migueles SA, Wu X, Phogat A, Shaw GM, Connors M, Hoxie J, Mascola JR, and R Wyatt. Analysis of neutralization specificities in polyclonal sera derived from human immunodeficiency virus type1-infected individuals. *J Virol.* 2009. 83: 1045-1059.

- Haigwood NL and VM Hirsch. Blocking and tackling HIV. *Nat Med.* 2009. 15: 841-842.
- 93 Kaul R, Trabattoni D, Bwayo JJ, Arienti D, Zagliani A, Mwangi FM, Kariuki C, Ngugi EN, MacDonald KS, Ball TB, Clerici M, and FA Plummer. HIV-1 specific mucosal IgA in a cohort of HIV-1 resistant Kenyan sex workes. *AIDS*. 1999. 13: 23-29.
- 94 Beyrer C, Artenstein AW, Rugpao S, Stephens H, VanCott TC, Robb ML, Rinkaew M, Birx DL, Khamboonruang C, Zimmerman PA, Nelson KE, and C Natpratan. Epidemiologic and biologic characterization of a cohort of HIV-1 highly exposed persistently seronegative commercial sex workers in Northern Thailand. *J Infect Dis.* 1999. 179: 59-67.
- 95 Nguyen M, Pean P, Lopalco L, Nouhin J, Phoung V, Ly N, Vermisse P, Henin Y, Barre-Sinoussi F, Burastero SE, Reynes JM, Carcelain G, and G Pancino. HIVspecific antibodies but not T-cell responses are associated with protection in seronegative partners of HIV-1-infected individuals in Cambodia. *J Acquir Immune Defic Syndr*. 2006. 4: 412-419.
- 96 Buchacz K, Parekh BS, Padian NS, van der Straten A, Phillips S, Jonte J, and SD Homberg. HIV-specfic IgG in cervicovaginal secretions of exposed HIVuninfected female sexual partners of HIV-infected men. *Aids Res Hum Retrovirusesi.* 2001. 17: 1689-1693.
- 97 Devito C, Hinkula J, Kaul R, Kimani J, Kiama P, Lopalco L, Barass C, Piconi S, Trabottoni D, Bwayo JJ, Plummer F, Clerici M, and K Broliden. Cross-clade HIV-1-speific neutralizing IgA in mucosal and systemic compartments of HIV-1-exposed persistently seronegative subjects. *J Acquir Immune Syndr*. 2002. 30: 413-420.
- 98 Devito C, Broliden K, Kaul R, Svensson L, Johansen K, Kiama P, Kimani J, Lopalco L, Piconi S, Bwayo JJ, Plummer F, Clerici M, and J Hinkula. Mucosal and plasma IgA from HIV-1-exposed uninfected individuals inhibit HIV-1 transcytosis across human epithelial cells. *J Immunol*. 2000. 165: 5170-5176.
- 99 Clerici M, Barassi C, Devito C, Pastori C, Piconi S, Trabottini D, Longhi R, Hinkula J, Broliden K, and L Lopalco. Serum IgA of HIV-exposed uninfected individuals inhibit HIV through recognition of a region within the α-helix of gp41. *AIDS*. 2002. 16: 1731-1741.
- 100 Lopalco L, Barassi C, Paolucci C, Breda D, Brunelli D, Nguyen M, Jouhin J, Luong TT, Truong LX, Clerici M, Calori G, Lazzarin A, Pancino G, and SE Burastero. Predictive value of ati-cell and anti-human immunodeficiency virus (HIV) humoral responses in HIV-1-exposed seronegative cohorts of European and Asian origin. J Gen Virol. 2005. 86: 339-348.

- Ghys PD, Belec L, Diallo MO, Ettiegne-Traore V, Becquart P, Maurice C, Nkengasong JN, Coulibaly IM, Greenberg AE, Laga M, and SZ Wikitor. Cervicoaginal anti-HIV antibodies in HIV-seronegative female sex workers in Abidjan Cote d'Ivoire. *AIDS*. 2000: 14: 2603-2608.
- 102 Dorrell L, Hessell AJ, Wang M, Whittle H, Sabally S, Rowland-Jones S, Burton DR, and PWHI Parren. Absence of specific mucosal antibody responses in HIV-exposed uninfected sex workers from the Gambia. *AIDS*. 2000: 14: 1117-1122.
- 103 Horton RE, Ball TB, Wachichi C, Jaoko W, Rutherford JW, McKinnon L, Kaul R, Rebbapragada A, Kimani J, and FA Plummer. Cervical HIV-specific IgA in a population of commercial sex workers correlates with repeated exposure but not resistance to HIV. *AIDS Res Hum Retro*. 2009. 25: 83-92.
- 104 Tudor D, Derrien M, Diomede L, Drillet AS, Houimel M, Moog C, Reynes JM, Lopalco L, and M Bomsel. HIV-1 gp41- specific monoclonal mucosal IgAs derived from highly exposed but IgG-seronegative individuals block HIV-1 epithelial transcytosis and neutralize CD4+ cell infection: an IgA gene and functional analysis. *Muc Immunol*. 2009: 1-15.
- 105 Berry JD, Rutherford J, Silverman GJ, Kaul R, Elia M, Gobuty S, Fuller R, Plummer FA, and CF Barbas III. Development of functional human monoclonal single-chain variable fragment antibody against HIV-1 from human cervical B cells. *Hybrid Hybridomics*. 2003. 22: 97-108.
- 106 Viau M, and M Zouali. Molecular determinants of the human antibody response to HIV-1: implications for disease control. *J Clin Immunol*. 2001. 21: 410-419.
- 107 Pandley J. Genetics of immunoglobulins. *Immunol Ser.* 1993. 58: 103-116.
- 108 Zouali M. Unraveling antibody genes. *Nat Genet*. 1994. 7: 118-120.
- 109 Azzazy HM, and WE Highsmith Jr. Phage display technology: clinical applications and recent innovations. *Clin Biochem*. 2002. 35: 425-445.
- 110 Michnick SW, and SS Sidhu. Submitting antibodies to binding arbitration. *Nat Chem Biol.* 2008. 4: 326-329.
- 111 Hoogenboom HR. Selecting and screening recombinant antibody libraries. *Nat Biotechnol.* 2005. 23: 1105-1116.

- 112 Beerli RR, Bauer M, Buser RB, Gwerder M, Muntwiler S, Maurer P, Saudan P, and MF Bachmann. Isolation of human monoclonal antibodies by mammalian cell display. *Proc Natl Acad Sci*. 2008. 105: 14336-14341.
- 113 Barbas 3<sup>rd</sup> CF, Kang AS, Lerner RA, and SJ Benkovic. Assembly of combinatorial antibody libraries on phage surfaces: the gene III site. *Proc Natl Acad Sci*. 1991. 88: 7978-7982.
- 114 Hanes J, and A Pluckthun. In vitro selection and evolution of functional proteins by using ribosome display. *Proc Natl Acad Sci*. 1997. 94: 4937-4942.
- 115 Boder ET, and KD Wittrup. Yeast surface display for screening combinatorial polypeptide libraries. *Nat Biotechnol.* 15: 553-557.
- 116 Ho M, Nagata S, and I Pastan. Isolation of anti-CD22 Fv with high affinity by Fv display on human cells. *Proc Natl Acad Sci*. 2006. 103: 9637-9642.
- 117 Smith K, Garman L, Wrammert J, Zheng NY, Capra JD, Ahmed R, and PC Wilson. Rapid generation of fully human monoclonal antibodies specific to a vaccinating antigen. *Nat Protoc*. 2009. 4: 372-384.
- 118 Tiller T, Meffre E, Yurasov S, Tsuiji M, Nussenzweig MC, and H Wardemann. Efficient generation of monoclonal antibodies from single human B cells by single cell RT-PCR and expression vector cloning. *J Immunol Met*. 2008. 329: 112-124.
- 119 Wrammert J, Smith K, Miller J, Langley WA, Kokko K, Larsen C, Zheng NY, Mays I, Garman L, Helms C, James J, Air GM, Capra JD, Ahmed R, and PC Wilson. Rapid cloning of high-affinity human monoclonal antibodies against influenza virus. *Nature*. 2008. 453: 667-672.
- 120 Huber M, Olson WC, and A Trkola. Antibodies for HIV treatment and prevention: window of opportunity? Current Topics in Microbiology and Immunology. *Human Antibody Therapeutics for Viral Disease*. Springer Berlin Heidelberg. 2008. 317: 39-66.
- 121 Sawyer LA. Antibodies for the prevention and treatment of viral diseases. *Antiviral Res.* 2000. 47: 57.
- 122 Brekke OH, Sandlie I. therapeutic antibodies for human diseases at the dawn of the twenty-first century. *Nat Rev Drug Discov*. 2003 2: 52-62.
- 123 Reichert JM, Rosensweig CJ, Faden LB, and MC Dewitz. Monoclonal antibody success in the clinic. *Nat Biotechnol.* 2005. 23: 1073-1078.

- 124 Haigwood NL, Montefiori DC, Sutton WF, McClure J, Watson AJ, Voss G, Hirsch VM, Richardson BA, Letvin NL, Hu SL, and PR Johnson. Passive immunotherapy in simian immunodeficiency virus-infected macaques accelerates the development of neutralizing antibodies. *J Virol*. 2004. 78: 5983-5995.
- 125 Burton DR, Barbas 3<sup>rd</sup> CF, Persson MAA, Koenig S Chanock RM, and RA Lerner. A large array of human monoclonal antibodies to type 1 human immunodeficiency virus from combinatorial libraries of asymptomatic seropositive individuals. *Proc Natl Acad Sci*. 1991. 88: 10134-10137.
- Burton DR, Pyati J, Koduri R, Sharp SJ, Thornton GB, Parren PWHI, Sayer LSW, Hendry RM, Dunlop N, Nara PL, Lamacchia M, Garratty E, Stiehm ER, Bryston YJ, Cao Y, Moore JP, Ho DD, and Barbas 3<sup>rd</sup> CF. Efficient neutralization of primary isolates of HIV-1 by a recombinant human monoclonal antibody. *Science*. 1994. 226: 1024-1027.
- 127 Zhou T, Xu L, Dey B, Hessell AJ, Van Ryk D, Xiang SH, Yang X, Zhang MY, Zwick MB, Arthos J, Burton DR, Dimitrov DS, Sodroski J, Wyatt R, Nabel GJ, and PD Kwong. Structural definition of a conserved neutralization epitope on HIV-1 gp120. *Nature*. 2007. 445: 732-737.
- 128 Koefoed K, Farnaes L, Wang M, Svejgaard A, Burton DR, and HJ Ditzel. Molecular characterization of the circulating anti-HIV-1 gp120-specific B cell repertoire using antibody phage display libraries generated from preselected HIV-1 gp120 binding PBLs. *J Immunol Meth.* 2005. 297: 187-201.
- 129 Zhang MY, Shu Y, Sidorov, and DS Dimitrov. Identification of a novel CD4i human monoclonal antibody Fab that neutralizes HIV-1 primary isolates from different clades. *Antiviral Res.* 2004. 61: 161-164.
- 130 de Haard JJW, Kazemier B, Oudshoorn P, Boender P, van Gemen B, Koolen MJM, van der Groen G, Hoogenboom HR, and JW Arends. Selection of human anti-human immunodeficiency virus type 1 envelope single-chain antibodies from a peripheral blood cell-based phage repertoire. *J Gen Virol*. 1998. 79: 2883-2894.
- 131 Bjorling E, von Garrelts E, Morner A, Ehnlund M, and MAA Persson. Human neutralizing human immunodeficiency virus type 2-specific Fab molecules generated by phage display. *J Gen Virol*. 1999. 80: 1987-1993.
- 132 Scheid JF, Mouquet H, Feldhahn N, Seaman MS, Velinzon K, Pietzsch J, Ott RG, Anthony RM, Zebroski H, Hurley A, Phogat A, Chakrabarti B, Yuxing L, Connors M, Pereyra F, Walker BD, Wardemann H, Ho D, Wyatt RT, Mascola JR, Ravetch JV, and MC Nussenzweig. Broad diversity of neutralizing

antibodies isolated from memory B cells in HIV-infected individuals. *Nature*. 2009. 458: 636-640.

- 133 Xu JY, Gorny MK, Palker T, Karwoska S, and S Zolla-Pazner. Epitope mapping of two immunodominant domains of gp41, the transmembrane protein of human immunodeficiency virus type 1, using ten human monoclonal antibodies. *J Virol*. 1991. 65: 4832-4838.
- 134 Binley JM, Ditzel HJ, Barbas 3<sup>rd</sup> CF, Sullivan N, Sodroski J, Parren PWHI, and DR Burton. Human antibody responses to HIV type 1 glycoprotin 41 cloned in phage display libraries suggest three major epitopes are recognized and give evidence for conserved antibody motifs in antigen binding. *AIDS Res Hum Retro.* 1996. 12: 911-924.
- 135 Gustchina E, Louis JM, Lam SN, Bewley CA, and GM Clore. A monoclonal Fab derived from a human nonimmune phage library reveals a new epitope on gp41 and neutralizes diverse human immunodeficiency virus type 1 strains. *J Virol.* 2007. 81: 12946-12953.
- 136 Mascola JR, D'Souza P, Gilbert P, Hahn BH, Haigwood NL, Morri L, Petropoulos CJ, Polonis VR, Sarzotti M, and DC Montefiori. Recommendations for the design and use of standard virus panels to assess neutralizing antibody responses elicited by candidate human immunodeficiency virus type 1 vaccines. *J Virol*. 2005. 79: 10103-10107.
- 137 Choudry V, Zhang MY, Sidrov IA, Louis JM, Harris L, Dimitrov AS, Bouma P, Cham F, Choudhary A, Rybak SM, Fouts T, Montefiori DC, Broder CC, Quinnan Jr. GV, and DS Dimitrov. Cross-reactive HIV-1 neutralizing monoclonal antibodies selected by screening of an immune human phage library against an envelope glycoprotein (gp140) isolated from a patient (R2) with broadly HIV-1 neutralizing antibodies. *Virology*. 2007. 363: 79-90.
- 138 Song L, Sun ZYJ, Coleman KE, Zwick MB, Gach JS, Wang JH, Reinherz EL, Wagner G, and M Kim. Broadly neutralizing anti-HIV-1 antibodies disrupt a hinge-related function of gp41 at the membrane interface. *Proc Natl Acad Sci.* 2009. 106: 9057-9062.
- 139 Steiner LA, and HN Eisen. The relative affinity of antibodies synthesized in the secondary response. *J Exp Med*. 1967. 126: 1161-1183.
- 140 Bose B, and S Sinha. Problems in using statistical analysis of replacement and silent mutations in antibody genes for determining antigen-driven affinity selection. *Immunology*. 2005. 116: 172-183.

- 141 Chang B, and P Casali. The CDR1 sequences of a major proportion of human germline Ig VH genes are inherently susceptible to amino acid replacement. *Immunol Today.* 1994. 8: 367-373.
- 142 Lossos IS, Tibshirani R, Balasubramanian N, and R Levy. The inference of antigen selection on Ig Genes. *J Immunol*. 2000. 165: 5122-5126.
- 143 Brochet X, Lefranc MP, and V Giudicelli. IMGT/V-QUEST: the highly customized and integrated system for IG and TR standardized V-J and V-D-J sequence analysis. *Nucleic Acids Res.* 2008. 36: W503-8.
- Hessell AJ, Rakasz EG, Tehrani DM, Huber, M, Weisgrau KL, Landucci G, Forthal DN, Koff WC, Poignard P, Watkinds DI, and DR Burton. Broadly neutralizing monoclonal antibodies 2F5 and 4E10, directed against the human immunodeficiency virus type 1 (HIV-1) gp41 membrane proximal external region (MPER), protect against SHIV<sub>Ba-L</sub> mucosal challenge. *J Virol*. 2009. 0: JVI.01272-09v1
- 145 Walker LM, Phogat SK, Chan-Hui PY, Wagner D, Phung P, Goss JL, Wrin T, Simek MD, Fling S, Mitcham JL, Lehrman JK, Priddy FH, Olssen OA, Frey SM, Hammond PW, Kaminsky S, Zamb T, Moyle M, Koff WC, Poignard P, and DR Burton. Broad and potent neutralizing antibodies from an African donor reveal a new HIV-1 vaccine target. *Science*. 2009. 326: 285-289.
- 146 Koefoed K, Farnaes L, Wang M, Svejgaard A, Burton DR, and HJ Ditzel. Molecular characterization of the circulating anti-HIV-1 gp120-specific B cell repertoire using antibody phage display libraries generate from pre-selected HIV-1 gp120 binding PBLs. *J Immunol Methods*. 2005. 297: 187-201.
- 147 Chothia C, Lesk AM, Gherardi E, Tomlinson IM, Wlater G, Markds JD, Llewelyn MB, and G Winter. Structural repertoire of the human VH segments. *J Mol Biol*. 1992. 227: 799-817.
- 148 Bryson, S, Julien JP, Isenman DE, Kunert R, Katinger H, and EF Pai. Crystal structure of the complex between the F(ab)' fragment of the cross-neutralizing anti-HIV-1 antibody 2F5 and the F(ab) fragment of its anti-idiotypic antibody 3H6. *J Mol Biol.* 2008. 382: 910-919.
- 149 Pantophlet R, Wang, M, Aguilar-Sino RO, and DR Burton. The human immunodeficiency virus type 1 envelope spike of primary viruses can suppress antibody access to variable regions. *J Virol.* 2009. 83: 1649-1659.
- 150 Gorny MK, Stamatatos L, Volsky B, Revesz K, Williams C, Wang XH, Cohen S, Staudinger R, and S Zolla-Pazner. Identification of a new quaternary neutralizing epitope on human immunodeficiency virus type 1 virus particles. *J Virol.* 2005. 79: 5232-5237.

- 151 Parren PW, Gauduin MC, Koup RA, Poignard P, Fisicaro P, Burton DR, and QJ Sattentau. Relevance of the antibody response against human immunodeficiency virus type 1 envelope to vaccine design. *Immunol Lett*. 1997. 57:125-132.
- 152 Berberian L, Goodglick, L, Kipps, TJ, and J Braun. Immunoglobulin VH3 gene products: natural ligands for HIV gp120. *Science*. 1993. 261: 1588-1591.
- 153 Roben P, Moore JP, Thali M, Sodroski J, Barbas III CF, and DR Burton. Recognition properties of a panel of human recombinant Fab fragments to the CD4 binding site of gp120 that show differing abilities to neutralize human immunodeficiency virus type 1. *J Virol*. 1994. 68: 4821-4928.
- 154 Stamatatos L, Morris L, Burton DR, and JR Mascola. Neutralizing antibodies generated during natural HIV-1 infection: good news for an HIV-1 vaccine? *Nat Med.* 2009. 1: 866-870
- 155 Stamatatos L, Zolla-Pazner S, Gorny MK, and C Chen-Mayer. Binding of antibodies to virion-associated gp120 molecules of primary-like human immunodeficiency virus type 1 (HIV-1) isolates: effect on HIV-1 infection of macrophages and peripheral blood mononuclear cells. *Virology.* 1997. 229: 360-369.
- 156 Moore PL, Crooks ET, Porter L, Zhu P, Cayanan CS, Grise H, Corcoran P, Zwick MB, Franti M, Morris L, Roux KH, Burton DR, and JM Binley. Nature of nonfunctional envelope proteins on the surface of human immunodeficiency virus type 1. *J Virol.* 2006. 80: 2515-2528.
- 157 Burrer R, Haessig-Einius S, Aubertin AM, and C Moog. Neutralizing as well as non-neutralizing polyclonal immunoglobulin (Ig)G from infected patients capture HIV-1 *via* antibodies directed against the principal immunodominant domain of gp41. *Virology*. 2005. 33: 102-113.
- 158 Cavacini L, and M Posner.native HIV type 1 virion surface structure: relationships between antibody binding and neutralization or lesions from the viral capture assay. *Aids Res Hum Retroviruses*. 2004. 20: 435-441.
- 159 Nyambi PN, Mbah HA, Burda S, Williams C, Gorny MK, Nadas A, and S Zolla-Pazner. Conserved and exposed epitopes on intact, native, primary human immunodeficiency virus type 1 virions of group M. *J Virol*. 74: 7069-7107.
- 160 Poignard P, Moulard M, Golez E, Vivona V, Franti M, Venturini S, Wang P, Parren PW, and DR Burton. Heterogeneity of envelope molecules expressed on primary human immunodeficiency virus type 1 particles as probed by the binding of neutralizing and nonneutralizing antibodies. *J Virol.* 77: 353-365.

- 161 Pancera M, and R Wyatt. Selective recognition of oligomeric HIV-1 primary isolate envelope glycoproteins by potently neutralizing ligands requires efficient precursor cleavage. *Virology*. 2005. 5: 145-156.
- 162 Holl V, Hemmerter S, Burer R, Schmidt S, Bohbot A, Aubertin AM, and C Moog. Involvement of Fc gamma RI (CD64) in the mechanism of HIV-1 inhibition of polclonal IgG purified from infected patients in cultured monocyte-derived macrophages. *J Immunol*. 2004. 173: 6274-6283.
- Fiscus SA, Wallmark EB, folds JD, Fryer J, and CM van der Horst. Detection of infectious immune complexes in human immunodeficiency virus type 1 (HIV-1) infections: correlation with plasma viremia and CD4 cell counts. *J Infect Dis.* 1991. 164: 765-769.
- 164 Morrow WJ, Wharton M, Stricker RB, and JA Levy. Circulating immune complexes in patients with acquired immune deficiency syndrome contain the AIDS- associated retroviruses. *Clin Immunol Immunopathol*. 1986. 40: 515-524.
- 165 Heath SL, tew JG, Szakal AK, and GF Burton. Follicular dendritic cells and human immunodeficiency virus infectivity. *Nature*. 1995. 377: 740-744.
- 166 Smith-Franklin BA, Keele BF, Tew JG, Gartner S, Szakal AK, Estes JD, Thacker TC, and GF Burton. Follicular dendritic cells and the persistence of HIV infectivity: the roe of antibodies and FC gamma receptors. *J Immunol.* 2002. 168: 2408-2414.
- 167 Xiao, X, chen W, Feng Y, Zhu Z, Prabakaran P, Wang Y, Zhang MY, Longo NS, and DS Dimitrov. Germline-like predecessors of broadly neutralizing antibodies lack measurable binding to HIV-1 envelope glycoproteins: implications for evasion of immune responses and design of vaccine immunogens. *Biochem Biophys Res Commun.* 2009. 18: 404-409.
- 168 Huang CC, Venturi M, Majeed S, Moore MJ, Phogat S, Zhang MY, Dimitrov DS, Hendrickson WA, Robinson J, Sodroski J, Wyatt R, Choe H, Farzan M, and PD Kwong. Structural basis of tyrosine sulfation and VH-gene usage in antibodies that recognize the HIV type 1 coreceptor-binding site on gp120. *Proc Natl Acad Sci USA*. 2004. 2706-2711.
- 169 Betz AG, Neuberger MS, and C Milstein. Discriminating intrinsic and antigenselected mutational hotspots in immunoglobulin V genes. *Immunol Today*. 14: 405-411.

- 170 Wilson DP, Coplan PM, Wainberg MA, and SM blower. The paradoxical effects of using antiretroviral-based microbicides to control HIV epidemics. *Proc Natl Acad Sci USA*. 2008. 105: 9835-9840.
- 171 Veazey RS, Shattock RJ, Pope M, Kirijan JC, Jones J, Hu Q, Ketas T, Marx PA, Klassse PJ, Burton DR, and JP Moore. Prevention of virus transmission to macaque monkeys by a vaginally applied monoclonal antibody to HIV-1 gp120. *Nat Med*. 2003. 9: 343-346.
- 172 Corbett CR, Elias MD, Simpson LL, Yuan XY, Cassan RR, Ballegeer E, Kabani A, Plummer FA, and JD Berry. High-throughput homogeneous immunoassay readily identified monoclonal antibody to serovariant clostridial neurotoxins. *J Immunol Methods*. 2007. 328: 128-138.
- 173 Kohler G, and C Milstein. Continuous cultures of fused cells secreting antibody of predefined specificity. *Nature*. 1975. 256: 495-497.
- 174 Bowley DR, Labrijn AF, Zwick MB, and DR Burton. Antigen selection from an HIV-1 immune antibody library displayed on yeast yields many novel antibodies compared with selection from the same library displayed on phage. *Protein Eng Des Sel*. 2007. 20: 81-90.
- 175 Tonegawa S. Somatic generation of antibody diversity. *Nature*. 1983. 302: 575-581
- 176 Manser T. Evolution of antibody structure during the immune response. The differentiative potential of a single B lymphocyte. *J Exp Med.* 1989. 170: 1211-1230.

# PRIMER

Fr-1 VH1,3	CAGGTGCAGCTGCAGTCTGG
Fr-1 VH2	CAGGTCACCTGGCTCGAGTCTGG
Fr-1 VH4	CAGGTGCAGCTGCAGTCGGGG
Fr-1 VH4b	CAGGTGCAGCTACTCGAGTGGGG
Fr-1 VH5	AAAAAGCCCGGGGAGTCTCTGARGA
Fr-1 VH6	CAGGGTACAGCTGCTCGAGTCAGG
IgA1,2 Reverse	CTAGTGACCTTGGGGCTGGTCGGGGATGC
IgA1 Reverse	AGTTGAACTAGTTGGGCAGGGCACAGTCAC
IgG1 Reverse	CATGTACTAGTTTTGTCACAAGATTTGGG
IgG2a Reverse	TCGACACTAGTTTTGCGCTCAACTGTCTT
IgG3 Reverse	TGTGTGACTAGTGTCACCAAGTGGGGTTTT
IgM Reverse	CTCACACTAGTAGGCAGCTCAGCAATCAC
IgK-V1	TACGGGGAACTCCAGATGACCCAGTCTCC
IgK-V2,4	TACGGGGAACTCGTGATGACYCAGTCTCC
IgK-V3	TACGGGGAACTCGTGWTGACRCAGTCTCC
IgK-V5	TACGGGGAACTCACACTCACGCAGTCTCC
IgLam-V1a	TACGGGGAACTCGTGBTGACGCAGCCGCCCTC
IgLam-V1b	TACGGGGAACTCGTGCTGACTCAGCCACCCTC
IgLam-V2	TACGGGGAACTCGCCCTGACTCAGCCTCCCTCCGT
IgLam-V3	TACGGGGAAGCTCGAGCTGACTCAGCCACCCTCAGTGTC
IgLam-V4	TACGGGGAACTCGTGCTGACTCAATCGCCCTC
IgLam-V6	TACGGGGAACTCATGCTGACTCAGCCCCACTC
IgLam-V7,8	TACGGGGAACTCGTGGTGACYCAGGAGCCMTC
IgLam-V9	TACGGGGAACTCGTGCTGACTCAGCCACCTTC
lgLam-V10	TACGGGGAACTCGGGCAGACTCAGCAGCTCTC
VjKRev1	TTTGATTTCCACCTTGGTCCC
VjKRev2,4	TTTGATCTCCAGCTTGGTCCC
VjKRev3	
VJKRev5	
VjLamRev1236	
VjLamRev4	
VJLainkev57	
VH12340EAI	ULAUIIULAUUAAIIU <u>UAIAIU</u> UUIULAUUIULAUUIULAUUUUU CCACTTCTCACCAATTCCATATCCCACCTCCACCTACTCCACTCCA
VH4UEAI VHEEVT	
VIJEAI JaChnEVT	
Iguupea i Iga12bpEVT	
IgA120pEA1 KannaFoyt	
LambdaFovt1	
LambdaFoxt2	
VikRov1.Y	ΔΩΓΓΑΓΑΛΟΥΑΓΙΓΟ <u>ΟΜΙΑΓΙ</u> ΟΟΙΟΟΙΟΛΟΙΟΛΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟΟ
ViKRev2 4-X	
ViKRev2.Y	ΑGCCACAGTACCCATGCCTTTGATATCCACCTTTGCTCCC
ViKRev5-X	AGCCACAGTACCCATGGCTTTTAATCTCCAGTCGTCTCCC
VIL1236RevX	AGCCACAGTACCCATGGGCCTAGGACGGTCASCTTGGTSCC
.,	

VJL4RevX	AGCCACAGTAC <u>CCATGG</u> GCCTAAAATGATCAGCTGGGTTCC
VJL57RevX	AGCCACAGTAC <u>CCATGG</u> GCCGAGGACGGTCAGCTSGGTSCC
VJL1236-link1	ACCAAGSTGACCGTCCTAGGCCAGCCCAAGGCTGCCCCCTCGGTC
VJL4-link1	ACCCAGCTGATCATTTTAGGCCAGCCCAAGGCTGCCCCCTCGGTC
VJL57-link1	ACCCAGCTGATCATTTTAGGCCAGCCCAAGGCTGCCCCCTCGGTC
VJL1236-link2	GGGGGCAGCCTTGGGCTGGCCTAGGACGGTCASCTTGGTSCC
VJL4-link2	GGGGGCAGCCTTGGGCTGGCCTAAAATGATCAGCTGGGTTCC
VJL5,7-link2	GGGGGCAGCCTTGGGCTGGCCGAGGACGGTCAGCTSGGTSCC
IgGconF	CCTCCACCAA <u>GGGCCC</u> ATCGGTCTTCCCCCTGG
Link PCR3-g	GTCCTGGCCCACAGCATTCTCTCTCCCCTGTTGAAGCTCTTTGTGAC
PDGFR-rev	AGATTACTAT <u>GCTAGC</u> CTAACGTGGCTTCTTCTGCCAAGC
Link PCR4-g	AGCCTCTCCCTGTCTCCGGGTAATGCTGTGGGCCAGGACACGC
IL2ssVrecovF	GCACTAAGTCTTGCACTTGTCACG
KcVrecovR	CGGGAAGATGAAGACAGATGGTGC
LcVrecovR	CGGGAACAGAGTGACCGAGGGGGC
GcVrecovR	GGGAAGACCGATGGGCCCTTGGTGGA
PdbLiRevSEQ	GTATCTTATCATGTCTGGCCAGCTAGC
LamConF	CAGCCCAAGGCTGCCCCTCGGTC
LamConR	TGAACACTTCGTAGGGGCAACTGTTTCCTCCAC
4LFext	GCACTTGTCACGAATTCGATATCGGAGCTCACGCAGTCTCCAGGC
4LRext	TTATCATGTCTGGCCAGCTAGCCTAACACTCTCCCCTGTTGAAGC
B12VHFext	GCACTTGTCACGAATTCGATATCCCTCGAGCAGTC
pIgGmut1F	GGGGGGTTGGCGCTGACTAGTCAAAA
pIgGmut1R	CTAGTCAGGGCCGCAACCCCCCAA
pIgGmut2F	GGGGGGTTGGGGCGCTGACTAGTCAAA
pIgGmut2R	TTTTGACTAGGTCAGCGCCCCAACCCCCC
pIgGmut3F	GGGGGGTTGGGGCCGTGACTAGTCAAACA
pIgGmut3R	TGTTTTGACTAGTCACGGCCCCAACCCCCC
LamConFext	ATGCGATTCATA <u>CCATGG</u> CAGCCCAAGGCTGCCCCCTCGGTC
LamConRext	ATGCGATTCATA <u>GCTAGC</u> TGAACACTTCGTAGGGGCAACTGTTTCCTCCAC
SeqF1	ATGTACAGGATGCAACTCCTGTCTTGC

\* Underlined portions of sequence indicate the location of the restriction endonuclease site indicated in the text

# Appendix II

Plasmid	Backbone	Isotype	Restriction Enzymes	Mammalian Resistance	Alterations
P-dbIgG1	pFuse	IgG CH1-3	EcoRV, Apal	Zeocin	2 <sup>nd</sup> Apal site mut
P-dbKappa	pFuse	IGKC	EcoRV, Ncol	Zeocin	N/A
P-dbLambda	pFuse	IGLC	EcoRV, Ncol	Zeocin	N/A
P-dbIgG1-TM	pFuse	IgG CH1-3	EcoRV, Apal	Zeocin	PDGFR-TM at 3' CH3